

Bangor University

DOCTOR OF PHILOSOPHY

Development of thin film flexible and transparent electronics

Kumar, Dinesh

Award date: 2019

Awarding institution: Bangor University

Link to publication

General rights Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal ?

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.



Bangor University

DOCTOR OF PHILOSOPHY

Development of thin film flexible and transparent electronics

Kumar, Dinesh

Award date: 2019

Awarding institution: Bangor University

Link to publication

General rights Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal ?

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

DEVELOPMENT OF THIN FILM FLEXIBLE AND

TRANSPARENT ELECTRONICS



Dinesh Kumar

School of Computer Science and Electronic Engineering

Bangor University

A thesis submitted in partial fulfilment for the degree of

Doctor of Philosophy

October 2019

"Life is and will ever remain an equation incapable of solution, but it contains certain known factors."

- Nikola Tesla

ABSTRACT

This thesis focuses on the development of transparent and flexible components such as silver nanowire (AgNW) based transparent conducting electrodes (TCEs) and metal oxide thin film transistors (TFTs) with the aim of improving the stability and performance under real-world conditions. The electrical and environmental stability of these components is, therefore, investigated. For AgNWs a 3-step post processing method was used to enhance the electrical, surface properties of AgNWs TCEs. Thermal embossing process parameters were optimised using a Taguchi orthogonal array, where pressure, time and temperature of the embossing step were all varied. An in-depth study of the effect of thermal embossing, effect of N₂ plasma treatment and photonic sintering on the AgNW TCE properties was undertaken. X-ray photoelectron spectroscopy, atomic force microscopy and scanning electron microscopy analysis of AgNW TCEs revealed several mechanisms responsible for the reduction of sheet resistance and surface roughness. However, transmittance and haze remained unchanged after post processing of TCEs. Electrical and environmental stability of AgNW TCEs is investigated through accelerated lifetime testing (ALT) and method to enhance stability of AgNW TCEs are developed.

A Plackett-Burman design was applied to optimise the production of anodised (Al₂O₃) dielectric films for ZnO thin film transistor (TFTs). A comprehensive study of the effect of the processing parameters on the performance of anodized Al₂O₃ gate dielectric films for ZnO TFTs is conducted. The impact of the process factors on the overall performance of the ZnO TFTs is also investigated. The development of low voltage metal oxide TFTs based upon ZnO and flexible IGZO is reported using Al₂O₃ as the gate dielectric. The performance of transparent IGZO TFTs base on AgNW and ITO gate electrode is also studied.

Understanding the origin of electrical instability in inorganic metal oxide TFTs over long periods is also essential to realize high performance circuits. In this thesis, the effects of positive gate bias stress (PGBS) on IGZO TFTs are also investigated. It was found that the threshold voltage, V_T , always shifted in the direction of the applied gate voltage. It was also observed that the threshold voltage shift, ΔV_T , is reduced with CYTOP passivation and white light illumination. A density of states (DoS) analysis is carried out during PGBS to confirm the origin of trap states due to bias stressing.

ACKNOWLEDGEMENTS

First of all, I would like to thank my supervisor, Dr. Jeff Kettle, for giving me this opportunity and for his support and guidance over the last four years. Without his continuous enthusiasm, encouragement, and help this research work would hardly have been completed.

I would like to thank Professor Dr. Lucas Fugikawa Santos of UNESP Rio Claro, SP, Brazil and my colleague Tiago for their support in anodisation work. I would also like to thank Prof. Graham C Smith of the University of Chester for his support analysis of X-ray photoelectron spectroscopy (XPS) of ZnO films, and SEM of AgNW films at Chester University. I would like to thank Dr. Emmanuel Brousseau from Cardiff University for obtaining high resolution SEM images of transparent electrodes.

I also owe thanks to all my colleagues at Bangor University for sharing this journey with me: Noel, Vasil, Tudur, Luke, Eifion, Colin, Nor, Ali, Helder, Ricardo, Paul Sayers, Iwan, Wendy, David, Julie, Siân and Gaynor.

I am in debt to my parents, family; my wife Sunita Devi, my twin kids Rishab Bhatia and Rishit Bhatia, brother Er Sunil Kumar, sisters Anita Kumari and Anjna Kumari, for being there and for believing in me for all these years. I owe my deepest gratitude to Govt. of India for their financial support.

Finally, I would like to thank "God" whose Divine light and warmth provided me the inspiration, faith and strength to carry out the research work. This thesis is dedicated to all of them.

PUBLICATIONS

- **1. Dinesh Kumar** and J. Kettle. "IGZO TFT performance investigation using Taguchi Orthogonal arrays." (Under review in Micro & Nano Letters)
- **2. Dinesh Kumar**, T. C. Gomes, N. Alves, L. F. Santos, G. C. Smith and Jeff Kettle "UV phototransistors based upon spray coated and sputter deposited ZnO TFTs" (Under review in IEEE Sensors Journal)
- **3. Dinesh Kumar**, V. Stoichkov, E. Brousseau, G. C. Smith and J. Kettle. "High performing silver nanowires transparent conducting electrodes with a sheet resistance of 2.5 Ω Sq⁻¹ based upon a roll-to-roll compatible post-processing technique." *RSC Nanoscale* 11(2019) 5760.
- **4.** T. C. Gomes, **Dinesh Kumar**, L. F. Santos, N. Alves, J. Kettle. "Optimisation of the anodisation processing for aluminium oxide gate dielectrics in ZnO thin film transistors by multivariate analysis." *ACS Comb. Sci.*, 21 (2019) 370–379.
- V. Stoichkov, Dinesh Kumar, P. Tyagi, J. Kettle. "Multi stress testing of OPV modules for accurate predictive ageing and reliability predictions. *IEEE Journal of Photovoltaics* 8 (2018) 1058–1065.
- J. Kettle, V. Stoichkov, Dinesh Kumar, M. Corazza, S. Gevorgyan, F. K. "Using ISOS consensus test protocols for development of quantitative life test models in ageing of organic solar cells." *Sol. Energy Mater. Sol. Cells* 167 (2017) 53–59.
- Dinesh Kumar, V. Stoichkov, S. Ghosh, G. Smith, J. K. "Mixed-dimension silver nanowires for solution-processed, flexible, transparent and conducting electrodes with improved optical and physical properties. *Flex. Print. Electron.* 2 (2017) 15005.

PATENT APPLICATION

Dinesh Kumar, V. Stoichkov and J. Kettle, "Method for producing transparent conductor" UK patent application, No. 1706606.9 (GB)

BOOK CHAPTER PUBLISHED

Dinesh Kumar and Jeff Kettle. Chapter: High performing metal-oxide semiconductor thin film transistors (**DOI**: <u>10.1049/PBCS073F ch12</u>). Book Name: VLSI and Post-CMOS Devices, Circuits and Modelling. **Publisher: Institution of Engineering and Technology**, Publication City/Country Stevenage, United Kingdom.

CONFERENCES

- Dinesh Kumar, Aly Abdou, Jeff Kettle. Half-volt IGZO flexible thin-film transistors with Ebeam deposited Al₂O₃ gate dielectric. First edition of the IEEE International Conference on Flexible and Printable Sensors and Systems - FLEPS 2019. (Oral presentation- paper published in IEEE explore digital library)
- Jeff Kettle and Dinesh Kumar. Enhancing the stability of AgNW transparent conducting electrodes with 2.5Ω-Sq sheet resistance by the use of post-processing. First edition of the IEEE International Conference on Flexible and Printable Sensors and Systems FLEPS 2019 (Oral presentation- paper published in IEEE explore digital library)
- **3. Dinesh Kumar** and Jeff Kettle. High performing AgNW transparent conducting electrodes with a sheet resistance of 2.5 Ω Sq-1 based upon a roll-to-roll compatible post-processing technique. **Fifth Innovation in Large-Area Electronics Conference (innoLAE 2019)** at Genome Campus Conference Centre in Cambridge, United Kingdom. (Oral presentation)
- 4. Dinesh Kumar and Jeff Kettle. Control over voltage threshold shift in IGZO TFTs with enhanced electrical characteristics using novel passivation. Fifth Innovation in Large-Area Electronics Conference (innoLAE 2019) at Genome Campus Conference Centre in Cambridge, United Kingdom. (Poster presentation)
- Dinesh Kumar, Tiago Gomes, Neri Alves, Jeff Kettle. Understanding UV Sensor Performance in ZnO TFTs through the Application of Multivariate Analysis. International Conference: IEEE SENSORS 2018 held at Delhi, India during 28-31 October 2018. (paper published in IEEE explore digital library)
- 6. D. Kumar, V. Stoichkov, S. Ghosh, G.C. Smith, J. Kettle. Mixed dimensions silver nanowires thin, flexible, transparent and conducting electrodes with improved optical and physical properties. International Conference: PVSAT-13 held at Bangor University, UK during 5-7 April 2017. (poster presentation)
- Jeff Kettle, Vasil Stoichkov, Dinesh Kumar. Accelerated testing for predictive ageing in organic solar cells for outdoor applications. Third Innovation in Large-Area Electronics Conference (innoLAE 2017) at Genome Campus Conference Centre in Cambridge, United Kingdom. (Oral presentation)
- D. Kumar, V. Stoichkov, S. Ghosh, G.C. Smith, J. Kettle. Spray coated silver nanowire electrodes as a transparent electrode in OPVs. Innovation in Large-Area Electronics Conference (innoLAE 2016) at Genome Campus Conference Centre in Cambridge, United Kingdom. (poster presentation)

CONTENTS

Declaration and Consent	i
Abstract	iv
Acknowledgements	v
Publications	vi
Conferences	vii
Contents	viii
Chapter 1. Introduction	1
1.1 Highlights and Structure of this Thesis	3
Chapter 2. Literature Review of Transparent Ele	ectronics5
2.1 Review of Transparent and Flexible Electronic	cs5
2.1.1 Review of State-of-the-Art in Transpare	nt Electrodes7
2.2 Review of Properties of Transparent Conducti	ive Electrodes8
2.2.1 Sheet Resistance (R _{sh})	8
2.2.2 Transmittance (T)	
2.2.3 Figure of Merit (FoM)	
2.2.4 Mechanical Flexibility	
2.3 Post Processing Treatment	
2.4 Physical and Chemical Deposition Techniques	
2.5 Review of Thin Film Transistors and Circuits	
2.6 Materials for Transparent and Flexible TFTs	
2.6.1 Transparent and Flexible Substrates	
2.6.2 Dielectric Layers	
2.6.3 Metal oxide semiconductors	

2.7	Device Structures of TFTs	22
2.8	Operation Principles of TFTs	23
2.9	Extraction of Performance Parameters of TFTs	27
2.1	0 State-of-the-Art Performance in Metal Oxide TFTs	28
2.1	1 Process Optimization Techniques	29
	2.11.1 Taguchi Orthogonal Arrays	29
	2.11.2 Properties of Orthogonal Arrays	31
	2.11.3 Assumptions of the Taguchi Method	32
	2.11.4 Plackett-Burman Design	33
2.12	2 Accelerated Lifetime Testing	34
	2.12.1 Quantitative Accelerated Life Tests (QALT)	34
2.1	3 Electrical and Environmental effects in AgNW Films	35
2.14	4 Summary	35
Chapte	er 3. Experimental Methods Used for Fabrication and Characterization	37
3.1	AgNWs solution Preparation	37
	3.1.1 Materials and AgNW Solution Preparation	37
	3.1.2 Fabrication Steps of AgNW Electrodes	39
	3.1.3 Experimental Setup for Spray Coating AgNWs	39
3.2	AgNWs Film Deposition and Post Processing Approach	40
	3.2.1 Embossing Experimental Setups for AgNW TCEs	40
	3.2.2 Post Processing using Nitrogen Plasma and Photonic Curing	42
3.3	TFT Fabrication	44
	3.3.1 Materials used for TFT Preparation	44
	3.3.2 ZnO Precursor Solution Preparation and Deposition	44
	3.3.2 ZnO Precursor Solution Preparation and Deposition3.3.3 Fabrication Steps for Oxide Semiconductor with Solution Process and	44
	3.3.2 ZnO Precursor Solution Preparation and Deposition3.3.3 Fabrication Steps for Oxide Semiconductor with Solution Process and Radio Frequency (RF) Sputtering	44 45

3.3.5 Lift-Off Process	48
3.4 Characterization of TCEs and TFTs	48
3.4.1 TFT CV/Cf and MIM Capacitance Characterization	48
3.4.2 UV Saturation Experiment Setup	49
3.4.3 TFT Current Voltage (IV) Characterization	49
3.4.4 IV Characterization and Accelerated Lifetime Testing (ALT) of AgNW T	CEs
	50
3.4.5 Setup for Sheet Resistance Measurement of AgNW TCEs	52
3.4.6 Atomic Force Microscopy	52
3.4.7 Scanning Electron Microscopy	53
3.4.8 Optical Characterization	53
3.4.9 White light interferometry (WLI) and haze measurements	53
3.4.10 Bias Stress Characterization	54
3.5 Extraction of the Density of Trap State Using the Grünewald Model	55
Chapter 4. Application of Thermal Embossing to Enhance the Properties of	
AgNW TCEs	57
4.1 Initial AgNW TCE Fabrication and Optimisation	57
4.2 Effect of Thermal Embossing on AgNW TCEs Properties	58
4.3 Taguchi Orthogonal Array Optimisation before and after Thermal Embossing	ŗ
Process	60
4.4 Effect of Photonic Curing and N_2 Plasma on AgNW TCEs	62
4.5 Effect of Post Processing on Electrical and Optical Performance of AgNW TCE	s 63
4.6 Surface Topographic Properties	64
4.6.1 Surface Topographic Properties Before and After Post Processing	64
4.6.2 Application of Zinc Oxide Nanoparticles to Further Improve Surface	
Properties	66
4.6.3 Demonstration of OPV Device	68

4.7 Mixed AgNW TCEs	70
4.7.1 Optical and Electrical Properties	72
4.7.2 Surface Roughness	74
4.8 Effect of Post Processing on Stability of AgNW Films	76
4.9 QALT Testing of the Reliability of AgNW Films	77
4.10 Adhesion and Mechanical Testing	
4.11 XPS Analysis of the AgNW Films	
4.12 Summary	
Chapter 5. Development of ZnO TFTs Using Design of Experiments	(DOE)
Techniques	
5.1 Development of High-k Dielectric (Al ₂ O ₃)	
5.2 Optimisation of Anodic Al ₂ O ₃ using Plackett-Burman Design	
5.3 ZnO TFT using anodic Al_2O_3 as the gate dielectric	
5.4 Influence of the Dielectric Layer Fabrication Factors on the ZnO Th	FTs
Characteristics	91
5.5 ZnO TFT Overall Performance	98
5.6 Optimisation of Sputter Coated ZnO TFTs using Taguchi OA	100
5.6.1 Pareto Chart and Term Effects for Mobility	101
5.7 Comparison of Sprayed and Sputtered ZnO TFTS	103
5.7.1 Optical Spectroscopy of ZnO Films	103
5.7.2 Atomic Force Microscopy of ZnO Films	104
5.7.3 X-Ray Diffraction of ZnO Films	105
5.7.4 XPS Analysis of the ZnO Films	107
5.7.5 Surface Chemistry	109
5.8 Electrical Characterization of ZnO Thin-Film Transistors Fabricate	ed using Spray
and Sputter Deposition	112
5.9 UV Saturation Experiment	

5.10 Sum	ımary	119
Chapter 6.	Effect of Bias Stressing on Indium Gallium Zinc Oxide TFT	s and
Flexible 7	ſFTs	121
6.1 Expe	rimental details	122
6.2 Effec	t of CYTOP Passivation on Performance of IGZO TFTs	123
6.3 Bias	Stressing of IGZO TFTs	123
6.3.1	Comparison of Unpassivated Positive Gate Bias Stress	124
6.4 Reco	very Procedure	127
6.5 Effec	t of CYTOP Passivation, White light (WL) and Positive Gate Bias	stress129
6.6 CV M	easurements Before and After Bias Stressing	130
6.7 Dens	ity of States (DOS) Analysis	131
6.8 Low	Voltage Flexible TFTs	134
6.9 Tran	sparent IGZO TFTs on AgNW and ITO Electrodes	139
6.10 Sum	ımary	141
Chapter 7.	Conclusions and Further Work	142
7.1 Conc	lusions	142
7.2 Futu	re Work	144
References.		149
Appendix A	Surface Roughness Parameters	I
Roughne	2SS	I
Appendix B	Fabrication recipe of AgNWs TCEs	III
Appendix C	Aluminum Anodization Process and Transfer Curves	VI

Chapter 1. Introduction

There has been significant research to realise the high performance of transparent and flexible electronics. Market demand for high-performance electronics has also increased because of new emerging applications. It is now estimated that the market for flexible electronics will reach \$300 billion by 2028, with growth from \$29.28 billion in 2017 to over \$63 billion in 2023 [1]. Flexible electronics could transform the way we make and use electronic devices. Many applications that require bendability to conform to the curved surface are driving the progress in the field, which in turn opens the door to futuristic applications such as foldaway smartphone displays, solar cells on plastic substrates, advanced medical devices, mobile health, and wearable sensors [2]. The anticipation is that these new form-factors will expand the scope of microelectronics and lead to further innovation. However, materials, packaging and manufacturing must evolve together with these devices to make these concepts economical and environmentally sustainable. Added value features such as flexibility and transparency have the potential to radically transform the electronics industry and spur a new era of growth. Figure 1.1 shows the current market size of the printed, organic and flexible electronics industry for year 2019 [3].



2019 Market Snapshot

Figure 1.1: Current market size of the printed, organic and flexible electronics industry [3].

2

Transparent electronics use transparent conductive oxides (TCOs) [4], wide bandgap semiconductors [5], 1-D inorganic nanostructures [6] and nanomaterials like metallic silver nanowires (AgNWs). Over the past two decades the use of transparent conductive materials has rapidly increased as a result of the increasing demand for display technology [7] and the development of thin film based solar cells. TCEs are also a crucial component of next generation electronic devices such as portable organic solar cells, organic light emitting diodes (OLEDs), bendable displays, touch screens, modern personal electronic devices, and transparent stretchable heater [8][9]. AgNWs with well-defined dimensions is a promising material for electrical and optical devices, particularly for flexible transparent conductive electrodes (TCEs). They are considered a strong competitor as a substitute to Indium Tin Oxide (ITO) in many of the applications. One of the primary reasons for this is that the incumbent technology (ITO) is not flexible; upon repeated bending, the films are known to crack and lose conductivity [10], which is particularly prohibitive for future product emergence of flexible electronic devices. When deposited from solution, AgNWs combine several advantages such as high optical transparency, low sheet resistance (R_{sh}) and mechanical flexibility. Despite the significant progress in TCEs represented by the above reports, understanding the stability issues and challenges such as long-term environmental stability, high surface roughness, electrical shorting problems must be overcome to fully integrate these new electrodes into high performance commercial devices and circuits.

High mobility thin-film transistors are also crucial for future high resolution and fast response flexible displays [11]. The issue of low mobility (~1 cm² V⁻¹s⁻¹) of electrons in amorphous silicon (a-Si) based TFT backplanes has led to the rise of new technologies that will overcome this problem. Metal oxides are one of such replacement technologies that have been researched and gained huge industry acceptance over the last few years. Today the display industry is extensively using metal-oxide TFT backplane technology in displays based on liquid crystal and light emitting diode (LEDs). Zinc oxide (ZnO) and especially indium gallium zinc oxide (IGZO) has been extensively studied over the last few decades, with much current academic and industrial research and development. Other favourable aspects of ZnO include its broad chemistry leading to many opportunities for wet chemical etching, low power threshold for optical pumping, radiation hardness, and biocompatibility. Optical pumping is a process in which light is used to raise (or "pump") electrons from a lower energy level in an atom to a higher energy level. ZnO and related compounds (IGZO) offer carrier mobility of more than 10cm²V⁻¹s⁻¹ approaching 100cm²V⁻

¹s⁻¹ [12][13]. High mobility oxide semiconductors, along with high-k gate dielectrics, are projected to be key ingredients for future transparent and emerging flexible and stretchable electronics [14]. Shown in Figure 1.2 are the applications of flexible electronics such as sensing, computing, data storage, and energy.



Figure 1.2: Applications of flexible electronics enabled through underpinning research in areas such as sensing, computing, data storage, and energy [1].

This thesis investigates high performance transparent components including AgNW TCEs for OPV, metal oxide TFTs and their electrical and environmental stability as outlined in the next section.

1.1 Highlights and Structure of this Thesis

This thesis focuses on the development of transparent and flexible components such as silver nanowire transparent conducting electrodes (AgNW TCEs), metal oxide TFTs with the aim of improving their stability and performance under real world conditions.

The highlights of this research are:

- The world leading Figure of Merit (FOM = 933) of AgNW TCE which is 2 × better as compared with current literature.
- Detailed stability analysis of AgNW TCE performance as a function of temperature and high current conditions.

- Demonstration of improved electrical and surface roughness performance of flexible AgNW TCEs using post R2R processing techniques (thermal embossing, N₂ plasma treatment and photonic sintering).
- Demonstration of mixed dimension AgNWs for solution-processed, flexible, transparent and conducting electrodes with improved optical and physical properties.
- Demonstration of improved performance of metal oxide TFTs and demonstration of process optimisation techniques such as Taguchi orthogonal array and Plackett-Burman Design for screening active factors.
- Demonstration of flexible, low voltage and transparent IGZO TFTs with high performance.

The structure of this thesis is as follows:

Chapter 2 provides a background of the science behind this research. In this chapter, a state-of-the-art review of transparent electronics, AgNW TCEs, and their post processing, metal oxide TFT development is discussed. Then, the TFT operation and the electrical characteristics, as well as the density of states, are explained. Finally, the electrical and environmental stability of AgNW TCEs is discussed.

Chapter 3 reports the experimental methods, setups used for manufacturing and characterisation of flexible TCEs and Metal oxide TFTs.

Chapter 4 is the first results chapter demonstrating the outcomes of flexible TCEs, TCEs post processing and performance optimisation using Taguchi Orthogonal Arrays. Also presented are the results from solution processed mixed dimension AgNW flexible TCEs with improved the optical and physical properties. Finally, the effect of post processing on high current and temperature stability of AgNW TCEs has been investigated.

Chapter 5 reports the ZnO based TFTs development, as well as optimisation of high-k anodic dielectric (Al₂O₃) for TFTs using Plackett-Burman design.

Chapter 6 presents the effect of positive gate bias stressing on IGZO TFTs, flexible IGZO TFTs performance. Moreover, transparent AgNW film gated IGZO TFTs electrical, optical performance are reported and compared with TFTs on incumbent ITO transparent electrodes.

Chapter 7 summarises the main results from the thesis and discusses possible directions for further work.

Chapter 2. Literature Review of Transparent Electronics

This chapter details the current state and progress in the following areas investigated during this Ph.D.: silver nanowire based (AgNW) transparent conductive electrodes (TCEs) and metal oxide thin film transistors (TFTs). Basic fundamental background theory of TCEs and metal oxide TFTs are presented. Electrical, optical and physical properties, as well as deposition methods and post processing of AgNW TCEs, are reviewed. Then, the materials for TCEs, TFTs and n-channel TFT operation principles are explained. Furthermore, process optimization techniques such as Taguchi Orthogonal Arrays (OA) and Plackett-Burman Design (PBD) are detailed. Finally, the environmental and electrical stability of TCEs and TFTs are discussed.

2.1 Review of Transparent and Flexible Electronics

This section reviews the state of the art concerning transparent and flexible electronics, with the focus on AgNW based TCEs as well as ZnO and IGZO based TFTs. Flexible and transparent electronics presents a new age of electronic technologies. Recently, various applications in wearable electronics, flexible transparent OLED displays [15]–[20], radio-frequency identifications (RFIDs) [21] and sensing systems [22]–[25] have been demonstrated. Also, the technical progress in metal oxide TFTs is a significant factor for promoting the development of transparent and flexible displays [26]–[31].



Figure 2.1: (a) SEM image of a AgNW transparent electrode, (b) Optical transmittance as a function of sheet resistance for silver, graphene and single walled carbon nanotubes (SWNTs) [32].

Shown in Figure 2.1(a) are SEM images of a AgNW transparent electrode, (b) Transmittance as a function of sheet resistance for silver, graphene, single walled carbon nanotubes (SWNTs). It is evident that AgNWs are one of the most promising in terms of electrical and optical performance, outperforming graphene and SWNTs. This is because even low density of AgNWs on substrate produces high transmittance and low sheet resistance.

Shown in Table 2.1 is a general guide to the different transparent conductive material electrode fabrication.

Fabrication method ^a	TCOs	Graphene	CNTs	AgNWs
Chemical vapour deposition	+++	+++	+	-
Sputtering	+++	-	-	-
Spin coating	-	+	++	++
Spray deposition	++	+	+++	+++
Screen printing		+	++	++
Cost	Low-high	High	High	Medium
Processing temperature (°C)	> 200 ^b -1000	RT-1000 ^c	RT-700 ^c	RT-700
Uniformity	+++	+	++	++
Typical thickness (nm)	100-300	<5	<10	25>600 ^d
Typical Rsh Ω sq ⁻¹	5-100	30-5000	60-300	1-50
Transmission (at 550 nm)	80-97 %	80-96 %	80-91 %	80-96 %

Table 2.1: General guide to transparent conductive material electrode fabrication [33]. + symbols indicate positive results – symbols represent poor results.

^a Fabrication method refers to the production of electrodes directly, not to the production of the integral components which are used to fabricate the network.

^b Deposition temperatures of transparent conductive oxides (TCOs) frequently need several hundred degrees Celsius or the use of vacuum processes such as sputtering.

^c Networks can be fabricated at room temperature (RT) but if device fabrication requires high temperature processes vacuum or encapsulation is required to stabilize the films.

^d Thickness is diameter and density dependent though with high mechanical pressure it can be reduced to the diameter of a single nanowire.

The search for materials which can be used to make thin films that simultaneously show high transparency, low sheet resistance [34] and flexibility [35] has mainly focused on nanostructures such as conducting polymer [36], graphene [37], carbon nanotubes (CNT) [38], copper nanowires [39] and AgNWs [40]–[43]. Nanowires are one dimensional nanostructures possessing diameter of the order of a nanometer (10⁻⁹ meters). CNT, graphene, and transparent conducting polymer films show low conductivity as compared to metal nanostructures and hence metallic nanostructures are

an appealing alternative. Metallic nanostructured materials are normally solution processed and deposited as disordered percolation networks. A solution process technique, such as spray coating can be used to readily prepare AgNW TCEs [44], [45]. Spray pyrolysis is a process in which a thin film is deposited by spraying a solution on a heated surface. This process can be carried out at low temperature (<80 °C) and does not require any vacuum equipment.

The focus of this thesis is on AgNWs films so CNT, graphene, and TCO are not discussed. There are three specific advantages of AgNW networks as transparent conductive materials. First is the low sheet resistances attainable with AgNW percolative networks, which can be approximately 10 Ω sq⁻¹. Secondly, they can demonstrate flexibility on plastic substrates and thirdly they possess high transparency in the infrared region of the electromagnetic spectrum [46]. However, creating TCEs with high conductivity and transparency is not a trivial matter, since these properties are often mutually exclusive. Nevertheless, AgNW based TCEs are relatively less expensive and are compatible with a roll-to-roll manufacturing process [47]–[52].

2.1.1 Review of State-of-the-Art in Transparent Electrodes

Over the last couple of decades, extensive work has been performed for the development of new TCEs. Shown in Table 2.2, is a summary of the best performing AgNW TCEs compared with other TCEs and their fabrication processes.

Reference	Material	FoM	Rsh (Ω sq ⁻¹)	T(%)	Remark/Process	
L. Li <i>et al.</i> [53]	metal grid	198.3	9.8	83.1	Photolithography	
J. Y. Lee <i>et al.</i> [54]	AgNWs	150	16	86	Solution process	
Y. Jia <i>et al.</i> [55]	AgNWs	207.3	35	95	Dynamic Heating Method	
H. Du <i>et al.</i> [56]	AgNWs	212.8	7.1	79.2	Spin coating	
C. Preston <i>et al.</i> [57]	AgNWs	300	13	91	Lamination process	
L. J. Andres <i>et al.</i> [58]	AgNWs	338	20.2	94.7	Spray coating	
A. Kim <i>et al.</i> [59]	ZnO/AgNWs/ZnO	487.9	8	91	DC magnetron sputtering	
S. H. Jo <i>et al.</i> [60]	CNT/PEDOT:PSS	6	280	80	spray coating	
N. Kim <i>et al.</i> [61]	PEDOT:PSS	76	46.1	90	Solution process	
S. Bae Jo <i>et al.</i> [62]	Graphene	116.1	30	90	CVD	

Table 2.2: A summary of the best performing non TCO electrodes compared with other TCEs.

In early 2008 J. Y. Lee *et al.* demonstrated one of the first solution processed AgNW network TCEs [54]. Since then, a major focus has been on the inorganic synthesis methods to synthesize significant quantities of metallic nanowires, such as copper or AgNWs, with well controlled physical dimensions, formulated into stable dispersions [46], [63]–[68]. A. Kim *et al.* reported the best performing composite electrode based on AgNWs and ZnO films showing sheet resistance 8 Ω sq⁻¹ and transmittance 91% at 550 nm [59], although this process relied on a vacuum deposition step.

2.2 Review of Properties of Transparent Conductive Electrodes

TCEs are characterised by various electrical and optical properties such as sheet resistance, transmittance and haze [69], [70]. As TCEs play a critical role in electronic devices including liquid crystal displays, OLEDs and photovoltaics, the preferred TCE properties generally depend on their application. For instance, large optical haze (percentage of the transmitted light which is scattered) is disadvantageous to applications of TCEs in most electronic displays, however, large haze can enhance efficiency in photovoltaics because of the extended optical path length which increases the probability of photon absorption by the photosensitive material [71][72]. Irrespective of device type, the two most significant performance parameters are the TCE sheet resistance and optical transmittance. Thermal stability, processability and chemical compatibility with other materials, and work function are other key parameters.

Shown in Table 2.3 are the desired properties of transparent conductive TCEs for main applications in photovoltaics, OLEDs, film heaters and touch screens.

Table 2.3: Desired properties of transparent conductive electrodes for main applications [33]. The number of plus symbols indicates the importance of the property to the application.

Application	Transparency	Conductivity	Flexibility	Haze
Photovoltaics	+++	+++	+	+++
OLED	++	++	+	+
Film Heaters	+	+++	++	+
Touch Screens	+++	++	++	+

2.2.1 Sheet Resistance (R_{sh})

In terms of electrical properties, the low sheet resistance is usually desired. The sheet resistance of a TCE is a critical factor that requires precise control during the electrode fabrication process depending on the desired application. For example, in modern displays and touchscreen applications, in order to allow for fast screen response, the sheet resistance of the TCE has to be as low as possible. In the case of AgNWs, several physical characteristics of the nanowires themselves and network morphology affect the electrical

properties of nanowire networks and the sheet resistance [73]. In general, there is an increase in resistance of the network with a decrease in the length of the nanowires used and this trend is continued if the nanowire diameter is decreased [40], [46], [74]. For very low values of nanowire diameter, surface scattering effects lead to an increase in sheet resistance. Shown in Figure 2.2 (a) are the sheet resistance of AgNW films as a function of increased silver content and (b) AgNW weight percent (wt. %). As more AgNW material is deposited percolation network becomes bigger and more AgNWs junctions are created, leading to a decrease in the sheet resistance.



Figure 2.2: (a) The sheet resistance of AgNW films as a function of silver content and (b) the sheet resistance differences between single-layer AgNW films and AgNW/PEDOT:PSS films and the transmittance at 550 nm as a function of the concentration of AgNWs [75].

The nanowire density, which is deposited onto a substrate, plays a major role in the conduction properties of the network. In many studies, it has been shown that increasing the number of nanowires decreases the resistance; however, this also results in a decrease in the optical transmission of the network. Literature shows that the sheet resistance decreased with an increase in AgNW contents on the substrate [76],[77], as shown in figure 2.2 (a). Maintaining an equilibrium between the electrical and optical properties has long been a challenge for transparent conductive materials and fabrication processes. Electrical conductivity requirements differ according to the application of TCEs, for example, solar cells require a TCE sheet resistance on the scale of 10Ω sq⁻¹ [78][79]. Sheet resistance (R_{sh}) is normally measured using a four point probe, from which sheet resistance can be calculated from equation 2.1 [80], where, $\rho(t)$ is the film

sheet resistivity, t is film thickness, V is the applied voltage and I is the current passing through TCE. This equation is used in this thesis for sheet resistance calculations.

$$R_{Sh} = \frac{\rho(t)}{t}$$
 Eq. 2.1

$$\rho(t) = \frac{t * V(4.532)}{I}$$
Eq. 2.2

2.2.2 Transmittance (T)

Another important property of transparent electrodes is transmittance. Percentage transmittance is defined as the percentage of light passing through the AgNW network exclusive of the absorption in the reference substrate and is given by the following equation 2.3:

$$\%T = \frac{S(\lambda) - D(\lambda)}{R(\lambda) - D(\lambda)} \times 100\%$$
 Eq. 2.3

Where, S is the intensity of light at wavelength λ , D is the dark intensity at wavelength λ , and R is the reference intensity at wavelength λ .



Figure 2.3: (a) Plot of AgNW film transmittance ($\lambda = 550$ nm) versus sheet resistance data for films deposited on PET (b) Transmittance spectra for various AgNW films [81].

Shown in Figure 2.3 (a) is the plot of AgNW film transmittance (λ = 550 nm) versus sheet resistance data for films deposited on polyethylene terephthalate (PET) and Figure 2.3 (b) is the transmittance spectra for various AgNW films. It has also been shown

in the literature that longer AgNWs form a better network due to longer conducting paths, lower deposition density and an increased number of junctions between AgNWs, leading to lower junction resistance and high transmittance [82]. These factors indicate that the optimum of AgNWs should be longer, as this can improve the conductivity of TCEs, and thinner AgNWs as this decreases the deposition area and light scattering, improving the optical transparency and haze.

2.2.3 Figure of Merit (FoM)

To evaluate the trade-off between Transmittance (T) and sheet resistance (R_{sh}) values, a combined 'Figure of Merit' can be used to evaluate the quality of a transparent electrode. The Figure of merit (FoM) is often reported in the literature [58][83] and is given by equation 2.4. This equation is used in this thesis for FOM calculations.

$$FOM = \frac{188.5}{R_{sh}(\sqrt{\frac{1}{T}} - 1)}$$
 [58] Eq. 2.4

The best reported FoM [59] of composite AgNW/ZnO TCE to date is 487.9. In this work AgNWs form a random percolating network embedded between the ZnO layers prepared by direct current (DC) magnetron sputtering at room temperature.

2.2.4 Mechanical Flexibility

In contrast to ITO, AgNW TCEs on plastic are mechanically flexible. The literature confirms that the mechanical flexibility of the AgNW electrodes is particularly good. For example, J. Kettle *et al.* demonstrated that AgNWs TCE remains robust even after extortionate bending tests (100 cycles with a bending radius of 20 mm) [58]. After testing, no modification of the optical properties was observed and a drop lower than 5% in sheet resistance after 100 bending cycles were recorded. Conversely, ITO electrodes showed a drop of 77% in sheet resistance after the first 10 bending cycles and catastrophic failure after 20 cycles during the same test.

Recently, Y. Wang *et al.* reported that the relative change in sheet resistance of the film over 300 bending cycles of outward and inward bending was less than 1.5% and the films exhibited good conductivity and heating performance [84]. Figure 2.4 shows the relative change in the sheet resistance of the AgNW TCE versus the number of bending

cycles, where R and R₀ represent the sheet resistance of films before and after the bending test, respectively.



Figure 2.4: Example of a transparent conductive electrode showing the relative change in the sheet resistance of the AgNW TCE versus the number of bending cycles [79].

2.3 Post Processing Treatment

After the AgNW deposition process, some nanowires protrude from the TCE surface and some of them lie on top of each other, which results in increased surface roughness to a few hundred nanometers. However, much lower surface roughness is needed for various TCE applications. For instance, the active films in organic solar cells are thin (typically between 100 to 300nm) [85]. Optical, electrical and physical properties of AgNW based transparent electrodes depends on a number of factors including type of AgNWs [58],[86], deposition process [87] and also post treatment of electrodes such as thermal annealing [88][89], mechanical pressing [90]–[92], fast sintering [93], etc.

To enhance the quality and performance of TCEs, post-processing treatments on electrodes are needed. For example, H. H. Khaligh *et al.* reported the hot rolling of AgNW TCEs for surface roughness minimization, achieving the root-mean-square surface roughness (R_q) of 7nm and the maximum peak-to-valley (P_V) value 30 nm, making the electrodes suitable for typical organic devices [91]. In addition, the adhesion of the AgNWs to the substrate significantly increased with this process.

Shown in Table 2.4 are the surface roughness data of the AgNW electrodes post processed using a hot rolling process.

Table 2.4: Surface roughness data of the annealed AgNW TCEs and post processed using hot rolling [94].

Post treatment	RMS Roughness, Rq (nm)	Max peak to valley, Pv (nm)	
Annealed	14	>90	
Rolled	7	<30	

Shown in Figure 2.5 are SEM images of tilted (45°) AgNW film on PET after (a) annealing, and (b) hot rolling. (c) shows a SEM image of a tilted (85°), a hot-rolled electrode which demonstrates that the nanowires are embedded in the substrate surface.



Figure 2.5: SEM images of tilted (45°) AgNW film on PET after (a) annealing, (b) hot rolling. (c) SEM image of a tilted (85°) hot-rolled electrode [91].

The post treatment on TCEs is shown to mostly affect the resistance and surface roughness of the nanowire network. Even though AgNW TCEs have better electrical and optical properties, some drawbacks need to be addressed. For example, one major issue is too high surface roughness of AgNW TCEs which causes shorting of device electrodes [95]. This imposes difficulties to integrate AgNW TCEs in many electronic devices such as solar cells and TFTs.

A simple post processing technique used is plasma treatment of the surface of AgNW TCEs for the reduction of the sheet resistance cited in the literature. For example, Jun Li *et al.* demonstrated that argon (Ar) plasma treatment removes the residual surfactant, polyvinylpyrrolidone (PVP), and enhances the contact between adjacent AgNWs. They reported AgNWs TCEs with a sheet resistance of 7.2 Ω sq⁻¹ and a transmittance of 78% at 550 nm [96].

Another new technology for post processing AgNW TCEs is photonic curing that allows the sintering of AgNWs on low temperature, flexible substrates such as paper and plastic in less than 2 ms [97][98]. D. J. Lee *et al.* reported improvements in sheet resistance of AgNW TCE using the intense pulsed light (IPL) method, which welds the interconnections among AgNWs in a short time without heat or pressure treatment [99], [100]. They achieved the sheet resistance of 12.6 Ω sq⁻¹ with a transmittance of 85.7% at 550 nm.

2.4 Physical and Chemical Deposition Techniques

Physical methods such as sputtering, evaporation, laser deposition and chemical methods like chemical vapour deposition, sol-gel, chemical bath deposition, electroplating have been used for preparing TCEs. The best performances are generally obtained using physical methods such as chemical vapor deposition (CVD) or plasma-enhanced chemical vapor deposition (PECVD). However, these are not generally scalable to roll to roll manufacturing processes. Historically some of the earliest TCEs were deposited by solution processing using spray pyrolysis.

Based on the direct fabrication of devices, there are many deposition techniques for semiconductors and electronic materials. To obtain thin films with good quality, two common deposition techniques: physical and chemical depositions are used. Shown in Table 2.5 is a summary of the above mentioned deposition techniques.

Physical deposition	Chemical deposition
Vacuum thermal evaporation	Sol-gel technique
Electron beam evaporation	Chemical bath deposition
Laser beam evaporation	Spray pyrolysis technique
Molecular beam epitaxy	Electroplating technique
Ion plating evaporation	Low pressure (LPCVD)
DC sputtering	Plasma enhanced (PECVD)
RF sputtering	Atomic layer deposition (ALD)

Table 2.5: A summary of deposition techniques [101].

In this thesis, the spray pyrolysis technique, electron beam evaporation and radio frequency (RF) sputtering for deposition of thin films were used.

2.5 Review of Thin Film Transistors and Circuits

TFTs are widely used as a backplane to drive display components, such as LCD, OLED, and even μ -LCD. Response time and resolution of display highly dependent upon the mobility

of charge carriers and hence high mobility TFTs are crucial for future high resolution and fast response flexible displays [102].

A TFT is a type of field-effect transistor (FET) made by depositing thin films of a semiconductor layer as well as the dielectric layer and metallic contacts over a supporting substrate. The main differentiator from other FET technologies is that common substrate is glass or polyimide, unlike conventional FETs, where the semiconductor is used as substrate. Glass and certain plastic films are good insulators and this limits the leakage current to a minimum. In 1962, P. K. Weimer [103] fabricated the first *n*-channel TFT employing a top gate, staggered structure with micro-crystalline cadmium sulfide (CdS) deposited by evaporation in vacuum as the channel layer and silicon monoxide as the dielectric layer. These TFTs possessed field-effect mobility values of 1.1 cm²V⁻¹s⁻¹and an on-off current ratio of 10.

Since these early studies, a number of materials have been utilised in TFTs. Metal oxides, such as ZnO and IGZO, show better air stability and higher electron mobility. However, oxide-based TFTs so far provide only high-performance *n*-channel TFTs, which prevents the realization of CMOS technology. Therefore, organic-inorganic hybrid complementary structures comprised of *n*-type oxide semiconductor and *p*-type organic semiconductor can be potential candidates for complementary circuit design.

Metal oxide semiconductors have been demonstrated the TFTs with a number of materials specifically ZnO [104]–[106], zinc tin oxide (ZTO) [107]–[111], indium zinc oxide (IZO)[112]–[114], indium gallium oxide (IGO)[115], and indium gallium zinc oxide (IGZO) [103], [116]–[122].

Amongst metal oxide semiconductors, IGZO, ZnO, and IZO are more attractive because they can be processed at room temperature whilst maintaining good performance, making them compatible with flexible substrates and consequently reducing manufacture costs. IGZO and IZO films, which are normally deposited by physical vapor deposition or RF-magnetron sputtering at room temperatures, show amorphous nature and still achieve high electron mobility. The amorphous material structure of the IGZO films is stable up to 500 °C in air. Due to this, IGZO and IZO are predominantly attractive for microelectronic circuits on flexible substrates and mobility as high as 100 cm²V⁻¹s⁻¹, and on/off current ratio of 10⁷ have been demonstrated. For example, C. M. Hsuet *et al.* [123] fabricated IGZO films on glass substrates using cosputtering, demonstrating high carrier mobility around about 163.4 cm²V⁻¹s⁻¹. The rapid progress in the field of *n*-channel metal oxide TFTs has motivated research on *p*-channel metal oxide TFTs to realise metal oxide-based CMOS circuits that enable low power consumption, large-area electronics. For example, C. W. Shih, reported high performance *p*-tin oxide (SnO) thin-film transistor (*p*-TFT) by a simple process of reactive sputtering from a tin (Sn) target under oxygen environment, demonstrating high field effect mobility 7.6 cm²V⁻¹s⁻¹, subthreshold slope 140 mV/dec, and on-off current ratio 3 × 10⁴ [124]. Nickel oxide (NiO) is an attractive *p*-type semiconductor because its properties are very beneficial for many photocatalytic, battery, electrochromic and chemical sensing applications [125]. Fukai Shan *et al.* reported a full solution processed NiOx/AlOx TFTs with an operating voltage of 3.5 V that exhibits high hole mobility of around 25 cm²V⁻¹s⁻¹ [126]. E. Fortunato *et al.* reported copper oxide (Cu₂O) based TFTs deposited by reactive RF magnetron sputtering at room temperature and the TFTs exhibited field-effect mobility of 3.9 cm²V⁻¹s⁻¹and an on/off ratio of 2×10².

Oxide TFTs based electronics are not limited to lab scale research but are now expanding to the commercialization in display applications. W. J. Nam *et al.* successfully fabricated and demonstrated 55-inch OLED TV using IGZO TFTs [127]. Additional applications have also been demonstrated such as ring oscillators by Jiawei Zhang *et al.* [128], a complementary circuit based on *n*-type IGZO and *p*-type tin monoxide (SnO) with an oscillation frequency of 2.63kHz and the peak-to-peak oscillation amplitude 36.1 V at a supply voltage of 40 V.

The ongoing research and continuous increase in TFT performance could establish new markets. A complementary electronic device requires both *p*-channel and *n*-channel TFTs. Complementary inverters are preferred compared to only *n* or *p*-channel inverters because they allow low power consumption, higher gains, and low noise margins. Reported complementary inverters on glass or flexible substrates are mostly composed of organic TFTs for both *p*-channel and *n*-channel. Devices in organic electronics, *p*-type material development occurred much quicker. Among organic semiconductors, *p*-type pentacene has been the most widely studied material, with hole mobility values of 0.1 ~ $1 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$. However, recent development has seen hole mobilities >10 cm² V⁻¹ s⁻¹ [129] reported for solution processed polymers. Therefore, hybrid organic-inorganic complementary structures composed of the *n*-type oxide semiconductor and *p*-type organic semiconductor are potential candidates for circuit design based on CMOS technology. For example, Takeya *et al.* reported solution-processed organic(C₁₀-DNBDT-NW)-inorganic(IZO) hybrid CMOS inverter exhibiting a high gain reaching 890 [130].

2.6 Materials for Transparent and Flexible TFTs

In this section properties of various materials used for manufacturing transparent and flexible TFTs are described. They include transparent and flexible substrates, dielectric layer materials of different kinds.

2.6.1 Transparent and Flexible Substrates

TCEs and TFTs are usually made on optically transparent substrates. The properties of plastic substrates affect quality and carrier transportation behaviour and limit maximum fabrication temperature. Polymer optical films can be classified according to the glass transition temperature (Tg) into three types [131]: (1) conventional polymers (Tg < 100 $^{\circ}$ C), (2) common high temperature polymers (100 \leq Tg < 200 $^{\circ}$ C), and (3) high temperature polymers (Tg \geq 200 $^{\circ}$ C). Glass transition temperature is the temperature, below which the physical properties of plastics change to those of a glassy or crystalline state. Table 2.6 lists the basic properties of three kinds of polymer substrates, which are presently widely used, namely polyimide (PI), polyethylene-napthalate (PEN) and polyethylene terephthalate (PET).

Material	T _g (⁰ C)	λ_{c} (nm)	CTE (ppm ⁰ C)	Chemical resistance	Surface
					roughness
PI	300	500	12	Good	Good
PEN	140	380	20	Good	Moderate
PET	80	300	33	Good	Moderate
PDMS	-120	200	301	Good	Poor

Table 2.6: Basic properties of commonly used flexible transparent substrates [132].

Among all of the flexible substrates, PI shows the highest T_g , the smallest coefficient of thermal expansion (CTE) and surface roughness. Additionally, PI has also good chemical stability in acid, alkali and organic solvents. Though the cut-off wavelength (λ_c) is in the visible range, the PI substrates have deep colour and poor optical transmittance. Colourless and optically transparent polyimide (CPI) films have only recently been developed. It appears that the CPI substrates will be one of the best choices for next generation flexible transparent electronics [133]. The optical transparency of PET substrates is good (> 85%) [134] and they are currently widely used as the protection layer in various liquid crystal displays devices (LCDs), such as televisions, computers and cell phones. However, the main drawback of PET is its low T_g (\approx 80 °C), which may bring challenges in device fabrication, integration and operation. Furthermore, PET possesses

a high-water vapour transmission rate. PEN substrate has a higher T_g (\approx 140 °C) than PET, which makes it a suitable substrate for manufacturing devices [135]. Recently, polydimethylsiloxane (PDMS) substrates are used in flexible electronics applications. PDMS is a type of elastomer and upon flexing and stretching, it induces no plastic deformation [136]. Also, it shows high optical transmittance, short λ_c , which makes PDMS the perfect substrate for the emerging stretchable electronics [137]. It is worth noting that ultra-thin (25-100 µm) flexible glass [138], [139], for example, Willow by Corning company [140], is also regarded as a promising substrate for flexible transparent electronics. It shows a higher processing temperature (~ 500 °C), good resistance to scratching and higher optical transmittance [141]. However, up to now, a few pieces of research on flexible glass were reported in the literature.

To decrease the surface roughness of substrates and gas permeability, as well as increase the chemical resistance and adhesion of the films, a barrier or encapsulation layer is often used onto the substrates. The commonly used encapsulation materials are Al₂O₃, Si₃N₄, SiO and SiO₂, which are electrically insulating and easily grown by chemical vapour deposition [142]–[145].

2.6.2 Dielectric Layers

The quality of the dielectric and the semiconductor/dielectrics interface is of vital importance in many transparent electronic applications. Currently, there are three types of dielectric materials that are commonly used in flexible metal oxide TFTs.

Inorganic dielectrics: Silicon dioxide (SiO₂) and silicon nitride (Si₃N₄) are two kinds of inorganic dielectric materials adopted from a-Si:H and poly-Si TFTs. Nevertheless, the high deposition temperature (above 100 °C) for high-quality films by plasma enhanced chemical vapour deposition (PECVD) hinders their application on flexible substrates. Instead of SiO₂ [146], high-k dielectrics (Al₂O₃, HfO₂) are more widely used in flexible transparent metal oxide TFTs, because they can be synthesized at low temperature by atomic layer deposition (ALD) or solution processes like anodization of aluminum [147]. Inorganic dielectrics, such as Al₂O₃, exhibit low-temperature fabrication convenience and result in excellent device performance. However, mechanical failure may occur when the films are under tensile or compressive strains.

Organic dielectrics: Organic dielectric materials [148], such as poly(methyl methacrylate) (PMMA) and polystyrene (PS), can sustain larger strain [149] because the

molecules in them are linked through the van der Waals forces and/or hydrogen bonds which are weak interactions. In addition, simple and low-cost processes, such as spin coating and printing, can be used to deposit polymer dielectrics. More recently, low-k fluorinated polymers such as CYTOP have been used as dielectric materials [150]–[152]. Also, CYTOP has been used as a passivation layer in TFTs due to its hydrophobic nature [153], [154].

Organic/inorganic hybrid dielectrics: Even though polymeric dielectrics withstand greater strain than their inorganic counterparts and can relax the stress in the channel layer, they have some drawbacks. Firstly, polymers are usually soft, and consequently, the deposition of the channel layer may induce damage inside the polymer layer or at the polymer/channel interface, which will significantly influence the transport behaviour of charge carriers. Secondly, the dielectric constants (k) of most polymers are relatively low (2.5-2.6 for PVP). Thirdly, the polymeric dielectrics are more hydrophobic than inorganic materials, which is undesirable for the direct growth of channel semiconductors [155]. By utilizing a stacked organic/inorganic hybrid gate dielectric, [156] these problems can be solved. Also, by introducing inorganic nanoparticles into polymer matrices to form polymer nanocomposites. This was achieved, for example, Lai et al. fabricated a nanocomposite dielectric by incorporating high-k Al₂O₃ nanoparticles into polymer PVP films [157]. Majewski and co-workers reported solution processed high-k nanocomposite/low-k polymer bilayer (Barium strontium titanate and barium zirconate nanoparticles were dispersed in a poly (vinylidene fluoride-co-hexafluoropropylene) P(VDF-HFP) polymer matrix) gate dielectric used for the fabrication of organic field-effect transistors (OFETs) that operate effectively at 1V [158].

Various approaches can be used to manufacture flexible electronic devices. Each of them has its distinct requirements regarding the process parameters and utilized materials:

- (1) A silicon wafer thinned down to thicknesses below 50 μm becomes flexible, however, investigation on bending limits before a mechanical fracture or failure occurs needs to be carried out [159].
- (2) The thinned die can then be attached to a flexible substrate [160].
- (3) Devices can be manufactured on polymers such as PET or PEN, which are attached to a rigid, low-cost substrate.

- (4) Transfer printing methods can be used to fabricate electronic devices on a large scale, and then transferred to a plastic substrate.
- (5) Direct fabrication of devices on unconventional flexible substrates can combine the advantages of standard semiconductor technologies and large area processing, but suffers from the limited thermal stability of flexible substrates [161]. In this thesis, the latter approach was adopted for device fabrication.

2.6.3 Metal oxide semiconductors

The oxides of post-transition metals such as ZnO and IGZO have been known for a decade and these materials have large bandgap and wide controllability of carrier concentration, which can make them useful for transparent thin film transistor (TFTs) applications. The conduction mechanism in metal oxide semiconductors is very different from that in conventional Si-based semiconductors. Amorphous silicon has poor carrier transport properties (~1 cm²V⁻¹s⁻¹) relative to single crystalline silicon materials (~500 cm²V⁻¹s⁻¹). This is associated to the structure and conduction mechanism between amorphous and single crystal materials. Si has a strong covalent bond, and the conduction band minimum (CBM) and valence band maximum (VBM) are made of anti-bonding (sp³ σ *) and bonding $(sp^3 \sigma)$ states of Si hybridized orbitals, and its bandgap is formed by the energy splitting of the σ - σ * level shown by Figure 2.6 (a). The mobility of amorphous silicon is small, as carrier transport is controlled by hopping between localized tail states and band conduction is not achieved. Therefore, this hopping mechanism is strongly related to the local range order or structure, such as the strained and disordered chemical bonds from rather deep and high density of localized states below conduction band minimum (CBM) and above valence band maximum (VCM), causing the carrier trapping process. On the other hand, the oxide semiconductors have strong ionicity and charge transfer occurs from heavy metal ion to oxygen atoms shown by Figure 2.6 (b). When heavy metal ions and oxygen ions come close, charge transfer occurs due to largely different electron affinity and ionization potential (Figure 2.6 (c)). In addition, the Madelung potential which comes from the difference between cations and anions by virtue of their opposing charges stabilizes the ionized states during the ionic bonding process. As a result, in Figure 2.6 (c), the Madelung potential induces the energy splitting and the conduction band minimum (CBM) is occupied the 2p orbital of oxygen and the valence band maximum un-occupied by ns orbital of cations.



Figure 2.6: Schematic electronic structures of silicon and ionic oxide semiconductors. (a-c) [163].Bandgap formation mechanisms in (a) covalent and (b, c) ionic semiconductors. Closed and open circles denote occupied and unoccupied states, respectively. (d, e) Schematic orbital drawings for the carrier transport paths in crystalline and amorphous semiconductors [164]. (d) Covalent semiconductors with sp₃ orbitals (e) metal oxide semiconductors with the s-orbital overlap of metal cation.

The CBMs with large spatial sized s-orbital of heavy cations overlaps with the neighbouring metal s orbital, and makes the conduction path for carriers which are not influenced largely by disordered local structure. This is the reason why metal oxide semiconductors have high mobility even though there is an amorphous structure. Figure 2.6 shows the schematic drawing of the structure Si-based and metal oxide semiconductors structure. Figure 2.6 (a) is the Si semiconductor layer which has the hopping mechanism to transfer the electron carriers. In single crystal structure, there is a transfer path with hopping mechanism because of the long-range order in crystal structure using sp³ hybridization orbitals, and have a large amount of electron field mobility. However, the amorphous structure does not have any long-range order in this structure, and electron carriers cannot move fast in this structure because of the barrier with random structure. In a metal oxide semiconductor, large sized s-orbital of metal ion has an overlapping between an adjacent atom, and there provides the current path of electron carrier. Even though there has an amorphous structure in the metal oxide layer, s-orbital is easy to overlap as shown in Figure 2.6 (b).

Amorphous oxide semiconductors have a potential fluctuation above the mobility edge and the distribution of potential barriers exists in the conduction band. These different paths cause different conductance for electron transmission over the potential barrier. In other words, the carrier density creates the degenerated conduction for amorphous IGZO and crystalline IGZO.

The low defect density in the a-IGZO layer is the reason why IGZO has a good subthreshold slope in the transfer curve. This is therefore possible to apply the low voltage driven TFTs for having low power consumption.

2.7 Device Structures of TFTs

Shown in Figure 2.7 are the typical TFTs device structures [165], determined by the position of the gate, source, and drain contacts relative to the active semiconductor film [166]. The basic structures are either coplanar or staggered. In a staggered structure, also called top-contact structure, the gate contact is on the opposite side of the semiconductor film from the source and drain contacts, as shown in Figure 2.7 (a) and (c). In such an arrangement, the source-drain contacts are in direct contact with the induced channel. In a coplanar structure, also called bottom-contact structure, the gate, source, and drain contacts are all located on the same side of the semiconductor film, as shown in Figure 2.6 (b) and (d). TFTs are also categorized as bottom gate (Figure 2.7 (a) and (b)) or top gate ((Figure 2.7 (c) and (d)) based on the position of gate. In this thesis, we have used bottom-gate top-contact structure.



Figure 2.7: Cross-sections of simplified TFT device configurations: (a) Bottom-gate staggered structure (top-contact); (b) Bottom-gate coplanar structure (bottom-contact); (c) Top-gate staggered structure; (d) Top-gate coplanar structure [167].

2.8 Operation Principles of TFTs

The TFT is a three-terminal device, in which a voltage (positive for *n*-type and negative for *p*-type material) applied to gate electrode controls current flow between source and drain electrodes under an applied voltage bias [168]. In electron transporting TFTs, modulation of a negative channel current flowing between the metallic source and drain terminals by the applied positive gate voltage (V_G) forms the basis of the device operation. The gate electrode is always separated from the semiconductor by the gate dielectric. The source terminal is always grounded (Vs=0V). Electrode materials with appropriate work functions (i.e. matching energies) are selected to optimize electron injection and reduce parasitic effects such as contact resistance. Application of V_G modulates the conductivity of the semiconductor region in proximity of the gate dielectric. When a source-drain voltage (V_{DS}) is applied to the device, injected charge carriers (holes or electrons for *p*channel and *n*-channel devices, respectively) accumulate at the semiconductor/insulator interface giving rise to a channel current that flows between the source and drain terminals. When the electric field strength corresponding to V_{DS} is significantly smaller than that of V_G, the TFT operates in the so-called 'linear' regime: the distribution of electrons across the channel is uniform and the channel drain current (I_D) varies linearly with the applied V_{DS}. If the field strengths corresponding to V_G and V_D are comparable, the device enters the 'saturation' regime, where I_D saturates and becomes independent of V_{DS}. This is the point at which the depletion of electrons occurs at the region closest to the drain electrode, and thus the channel is described as pinched-off.


Figure 2.8 shows the four regions of *n*-channel TFT operation (a) off region (b) linear region (c) pinch off points (d) saturation region in a bottom-gate top-contact metal oxide TFT output characteristics and (e) transfer characteristics of n- channel TFT.

An oxide TFT device can be considered as two capacitor plates separated by an insulator between source/drain contact and gate contact as shown in figure 2.9 (a). To better understand the working of metal oxide TFTs, the energy band diagrams of a metal insulator semiconductor (MIS) structure are shown in Figure 2.9. Different from metal oxide semiconductor field-effect transistor (MOSFET) operations using an inversion-mode, organic or oxide semiconductor TFTs are operated in the accumulation-mode [169][170]. Figure 2.9 shows several energy band diagrams as viewed through the gate of an *n*-channel TFT operating in such mode. For simplicity, hereafter the operation of TFTs will be reviewed by assuming an oxide semiconductor channel.



Figure 2.8: (a) Schematic cross-section representation of a MIS capacitor. Energy band diagrams of MIS capacitor as viewed through the gate for different biasing conditions: (b) equilibrium, (c) $V_{GS} < 0$ V, and (d) $V_{GS} > 0$ V [166].

For an ideal MIS device, the work functions of the metal (Φ_m) and semiconductor (Φ_n) are equal which leads to the alignment of Fermi levels at equilibrium and no band bending or charge accumulation or depletion at the semiconductor interface. Figure 2.9 (b) shows an energy band diagram of such a device in equilibrium, with 0V applied to the gate. Figure 2.9 (c) shows an energy band diagram with the gate biased negatively. The applied negative bias repels mobile electrons from the semiconductor, leaving an electron depletion region near the insulator/semiconductor interface. When compared with the no bias condition, this biasing condition has a reduced conductance due to the reduced number of mobile electrons in the semiconductor. Figure 2.9 (d) shows an energy band diagram with the gate biased positively. The applied positive bias attracts mobile

electrons, forming an accumulation region near the insulator/semiconductor interface and these excess mobile electrons lead to an increase in the conductance.

The current-voltage characteristics of a TFT were derived based on the gradual channel approximation[171][172] with the x direction is defined as perpendicular to the channel and y parallel to the channel as shown in Figure 2.10. The edge of the source is at y = 0 and the edge of the drain at y = L. In this approximation, the charge carrier density per unit area in the channel varies as function of position in the y direction, thus the carrier concentration in channel is dependent on the lateral potential, V(y), which is largely determined by the drain potential V_D as well as the material properties of the channel[173]. The external parameters controlling the drain current I_{DS} are gate-to-source voltage (V_{GS}) and the drain-to-source voltage (V_{DS}). When gate-to-source voltage is set to be a value larger than the threshold voltage, V_T, of the TFT, carriers in the semiconducting channel layer between the source and drain are generated.



Figure 2.9: Cross-sectional view of the channel region of TFT used to derive the gradual channel approximation with x direction is perpendicular to the channel and y is parallel to the channel.

It is assumed that the carrier mobility along the channel is constant. However, in practical reality this is not always fulfilled, the threshold voltage changes along the channel since the channel voltage is not constant. Also, assume that the electric field component E_y along the y-coordinate is dominant compared to the electric field component E_x along the x-coordinate so that the current flows in the channel to the y-dimension only. This is known as gradual channel approximation. The boundary conditions for the channel voltage V(y) are as follows:

$$V(y=0) = V_S = 0$$
 and $V(y=L) = V_{DS}$ Eq. 2.5

Also, it is presumed that the entire channel region is in accumulation mode, that is:

$$V_{GS} \ge V_T, V_{GD} = V_{GS} - V_{DS} \ge V_T$$
 Eq. 2.6

Assuming C_i is the capacitance per unit area of the gate insulator and V_C channel potential equal to zero, the charge density Q in the channel, when gate potential is greater than threshold voltage, is related to gate voltage as follows:

$$Q = -C_i \left(V_{GS} - V_T \right)$$
 Eq. 2.7

Let $Q_I(y)$ be the total mobile electron charge in the accumulation layer. The charge can be expressed as a function of the gate-to-source voltage V_{GS} and the channel voltage $V_C(y)$ by equation 2.8:

$$Q = -C_i [V_{GS} - V_T - V_C(y)]$$
 Eq. 2.8

Let dR be the incremental resistance of the differential channel segment dy and dQ the corresponding differential charge. If all mobile electrons in the accumulation layer have uniform mobility, the incremental resistance can be expressed as follows:

$$dR = -\frac{dy}{W\mu Q_I(y)} \, dy$$
 Eq. 2.9

Where *W* is the channel width, μ is the field-effect mobility. Note that the minus sign is due to the negative polarity of the accumulation layer charge *Q*_{*l*}. We will assume that the channel current density is uniform across the segment. According to the one-dimensional model, the channel current, I_{DS}, flows between the source and drain in the y-coordinate direction. Using Ohm's law for this segment, the voltage drop along the incremental segment dy in the y-direction is given by equation 2.10:

$$dV_C = dR I_{DS} = -\frac{I_{DS}}{W\mu Q_I(y)} dy$$
 Eq. 2.10

Integrating the above equation along the channel from y=0 to y=L, using the boundary conditions stated earlier yields,

$$\int_{0}^{L} I_{DS} \, dy = -W\mu \, \int_{0}^{V_{DS}} Q_{I}(y) \, dV_{C}$$
 Eq. 2.11

The left-hand side of this equation is simply equal to I_{DSL} . The integral on the right-hand side is evaluated by replacing $Q_I(y)$ with equation 2.9. Thus,

$$I_{DS}L = W\mu C_i \int_0^{V_{DS}} (V_{GS} - V_T - V_C) dV_C$$
 Eq. 2.12

Assuming that the channel voltage V_C is the only variable in equation 2.11 that depends on the position y, the drain current is as follows:

$$I_{DS} = \mu C_i \frac{W}{L} ((V_{GS} - V_T) V_{DS} - \frac{1}{2} V_{DS}^2)$$
 Eq. 2.13

In the linear region ($V_{DS} \ll V_{GS}$) the drain current can be written as:

$$I_{DS} = \mu C_i \frac{W}{L} (V_{GS} - V_T) V_{DS}$$
 Eq. 2.14

When the TFT is biased within the pinch-off region and $V_{DS} = V_{GS} - V_T$, the drain current saturates. For the saturation region ($V_{DS} > V_{GS} - V_T$), equation 2.14 is no longer valid. The saturation drain current can be obtained by substituting $V_{DS} = V_{GS} - V_T$ into equation 2.13, yielding:

$$I_{DS} = \mu C_i \frac{W}{2L} (V_{GS} - V_T)^2$$
 Eq. 2.15

The gradual channel approximation assumes that the mobility is independent of the gate voltage and that the S/D contacts have negligible resistance.

2.9 Extraction of Performance Parameters of TFTs

TFTs are characterized by measuring the output and transfer characteristics. By measuring the output and transfer characteristics, the important TFT performance parameters such as saturation mobility (μ_s) in the saturation regime, the threshold voltage (V_T), Turn ON voltage (V_{ON}), on-off current ratio (I_{on}/I_{off}), and subthreshold slope (SS) were obtained.



Figure 2.10: The transfer characteristic of a *n*-channel TFT and square root of I_{DS} , showing various performance parameters such as drain ON current (I_{on}), drain off current (I_{off}), the TFT threshold voltage (V_T), TFT turn ON voltage (V_{ON}), subthreshold slope (SS).

Shown in Figure 2.11 the transfer characteristic of a *n*-channel TFT illustrating various performance parameters like subthreshold slope (SS), the threshold voltage (V_T) and drain ON current (I_{on}). The mobility in the saturation regime can also be extracted from the transfer characteristics. The saturation mobility can be extracted from the slope (equation 2.16) of the curve, which plots the square root of the saturation current as a function of the gate to source voltage (V_{GS}). The threshold voltage V_T can be determined by extrapolating this linear region to the horizontal x-intercept, as shown in Figure 2.11.

$$slope = \frac{\partial \sqrt{I_{DS}}}{\partial V_{GS}} = \sqrt{\frac{WC_i \mu_S}{2L}}$$
Eq. 2.16

$$\mu_{S} = \left(\frac{\partial\sqrt{I_{DS}}}{\partial V_{GS}}\right)^{2} \frac{2L}{WC_{i}}$$
 Eq. 2.17

Drain current on-off ratio is also extracted from the transfer characteristics. This ratio was determined by taking the ratio between the maximum on-current measured (I_{on}) and the minimum off-current measured when the TFT is off (I_{off}). Subthreshold swing is also the inverse of the subthreshold slope. The inverse subthreshold slope (SS) is estimated as the minima on a ($\partial(logI_{DS})/\partial V_{GS}$)⁻¹ vs. V_{GS} graph.

$$SS = \frac{d|V_{GS}|}{d \log|I_{DS}|}$$
 Eq. 2.18

where, SS is the subthreshold slope in V/decade.

2.10 State-of-the-Art Performance in Metal Oxide TFTs

The main progress of metal oxide TFTs is summarized in Table 2.7. Adamopoulos *et al.* reported high-mobility (85 cm²V⁻¹s⁻¹) Li-doped ZnO TFTs based on ZrO₂ high-k dielectric grown by spray pyrolysis in ambient air [174]. This is one of the highest reported mobility values for transistors based on a combination of a solution-processed oxide semiconductor and an oxide high-k dielectric to date. Hu *et al.* [120] reported a high mobility (142 cm²V⁻¹s⁻¹) Cu-contacted IGZO TFT by depositing an Al₂O₃ passivation layer on the back channel using RF sputtering. The underlying mechanism for the high mobility

was attributed to the generation of an In-rich layer at the IGZO and Al₂O₃ interface that increases the carrier concentration.

Table 2.7: Some of the state of the art flexible transparent TFTs based on ZnO and related materials from 2011-2019. "--" means not mentioned or not clear in the literature.

Material	Technique	T _{dep} .	Substrate	Dielectric	μ (cm ² V ⁻	On/off	Reference	Year
		/T _{post} .(°C)			$^{1}S^{-1}$)			
Zn0	Spray	300/500	Si++	SiO ₂	25	106	Adamopoulos	2011
	Pyrolysis						et al. [175]	
Li-doped	Spray	400	ITO	ZrO ₂	85	106	Adamopoulos	2011
Zn0	Pyrolysis						et al. [174]	
InSnO	Co-sputtering	150	Si	SiO ₂	28.7	109	Seo et al. [176]	2016
		(Vacuum)						
ZT0	CVD	350			14.6	107	Park et al. [177]	2017
IGZO	Spin	/500	Thin glass	Ta ₂ O ₅	23.8	106	Dai <i>et al</i> .[155]	2013
IGZO	Sputtering	RT/300	glass	Al ₂ O ₃	142	108	Hu <i>et al</i> .[120]	2017
1070	a		1	41.0	4.4		N . 154501	2010
IGZO	Sputtering	R1/400	glass	Al ₂ O ₃	14		Nam <i>et al</i> .[178]	2018
IGZO	Sputtering	100	glass	Al ₂ O ₃	26.03	107	Liu <i>et al.</i> [179]	2019

2.11 Process Optimization Techniques

In this section, process optimization techniques, Taguchi methods and Plackett-Burman design are discussed. Process optimisation techniques are an important tool for reducing the production time but have not been widely applied in metal oxide TFT development. The positive outcomes of process optimization techniques are minimized cost, maximized production with better accuracy.

2.11.1 Taguchi Orthogonal Arrays

The method of laying out the conditions of experiments linking multiple factors was first proposed by R.A. Fisher [180]. The method is commonly known as the factorial design of experiments. Most of the research processes usually involve a significant number of factors and a full factorial design results in a large number of experiments. To decrease the number of experiments to a practical level, only a small set from all the possibilities is selected. The method of selecting a limited number of experiments that produce the most information is known as a partial fraction experiment [181]. A well planned set of experiments, in which all parameters of interest are varied over a specified range, is a much better approach to obtain systematic data. Dr. Taguchi of Nippon Telephones and Telegraph Company, Japan has developed a method based on "Orthogonal Array"

settings " of control parameters. Taguchi constructed a special set of general design guidelines for factorial experiments that cover many applications [182]. This method uses a special set of arrays called orthogonal arrays (OA). These standard arrays lay down the way of conducting the minimal number of experiments, which could give the full information of all the factors that affect the performance parameter. The root of the OA method lies in choosing the level combinations of the input design variables for each experiment run.



Figure 2.11: Taguchi method design procedure for process optimisation.

There are many standard OAs available, each of the arrays are designated for a specific number of independent design variables and levels. For example, in order to perform an experiment to understand the influence of 4 different independent variables (as used in this work) with each variable having 3 set values (level values), then an L₉ OA might be the right choice.

Shown in Figure 2.12 is the Taguchi method design procedure for process optimisation. This procedure starts with determining suitable factors and levels and selecting proper OA. Then, all experiments are run according to selected OA. Finally, after measuring the response values analyse the data using design of experiment (DOE) software tool.

The L₉OA is selected for understanding the effect of 4 independent factors each having 3 factor level values [183][184]. This array assumes that there is no interaction between any two factors. While in many cases, no interaction model assumption is valid, there are some cases where there is clear evidence of an interaction.

L ₉ (3 ⁴) Orthogonal array										
Run	Indepe	Performance								
number		parameter								
					value					
	Variable 1	Variable 2	Variable 3	Variable 4						
1	1	1	1	1	p1					
2	1	2	2	2	p2					
3	1	3	3	3	р3					
4	2	1	2	3	p4					
5	2	2	3	1	р5					
6	2	3	1	2	р6					
7	3	1	3	2	p7					
8	3	2	1	3	p8					
9	3	3	2	1	n9					

Table 2.8: General Layout of L9 orthogonal array (OA).

Shown in Table 2.8 is the general layout of L₉ OA. There are entirely 9 experiments to be conducted and each experiment is based on the combination of level values as shown in Table 2.8. For example, the third experiment is conducted by keeping the independent design variable 1 at level 1, variable 2 at level 3, variable 3 at level 3, and variable 4 at level 3.

2.11.2 Properties of Orthogonal Arrays

The orthogonal arrays have the following distinct properties that reduce the number of experiments to be conducted.

The vertical column under each independent variables of Table 2.8 has a special combination of level settings. All the level settings appear an equal number of times. For

L₉ array under variable 4, level 1, level 2 and level 3 appears thrice. This is called the balancing property of orthogonal arrays.

All the level values of independent variables (input factors) are used for conducting the experiments. The sequence of level values for conducting the experiments shall not be changed. This means one cannot perform experiment 1 with variable 1, level 2 setup and experiment 4 with variable 1, level 1 setup. The reason for this is that the array of each factor column is mutually orthogonal to any other column of level values. The inner product of vectors corresponding to weights is zero. If the above 3 levels are normalized between -1 and 1, then the weighing factors for level 1, level 2, level 3 are -1, 0, 1 respectively. Hence the inner product of weighing factors of independent variable 1 and independent variable 3 would be

$$(-1*-1+-1*0+-1*1) + (0*0+0*1+0*-1) + (1*0+1*1+1*-1) = 0$$
 Eq.2.5

The design of experiments using the OA is, in most cases, efficient when compared to many other statistical designs. The minimum number of experiments that are required to conduct the Taguchi method can be calculated based on the degrees of freedom approach.

$$N_{Taguchi} = 1 + \sum_{i=1}^{NV} (L_i - 1)$$
 Eq.2.6

where, NV is the number of independent variables and L_i is the number of levels.

For example, in the case of 8 independent variables study having 1 independent variable with 2 levels and the remaining 7 independent variables with 3 levels (L18 orthogonal array), the minimum number of experiments required based on the above equation is 16. Because of the balancing property of the orthogonal arrays, the total number of experiments shall be multiple of 2 and 3. Hence, the number of experiments for the above case is 18.

2.11.3 Assumptions of the Taguchi Method

The additive assumption implies that the individual or main effects of the independent variables on performance parameter are separable. Under this assumption, the effect of each factor can be linear, quadratic or of higher order, but the model assumes that there exists no cross-product effects (interactions) among the individual factors. That means the effect of independent variable 1 on performance parameter does not depend on the different level settings of any other independent variables and vice versa. If at any time, this assumption is violated, then the additivity of the main effects does not hold, and the variables interact violated, then the additivity of the main effects does not hold, and the variables interact.

2.11.4 Plackett-Burman Design

Plackett–Burman's designs are experimental designs presented in 1946 by Robin L. Plackett and J. P. Burman. An established and economical approach that gives information only on the effects of single factors, but not on interactions, is the Plackett–Burman Design (PBD) method. At the early stage of an experimental search, it is important to identify the subspace of active factors among many, in other words, many factors are considered and the objective is to identify those factors that have large effects. This is known as factor screening (to identify a set of active factors from a larger set of factors). The number of runs n in a PB design is equal to a multiple of four[185]. Plackett and Burman only included the designs with $n \le 100$, and they also omitted the design where n = 92. For PB designs where the number of runs is equal to a power of two, the designs coincide with the regular ones, and the rest of the PB designs are non-regular.

Shown in Table 2.9 the 12-run PBD matrix. In this thesis a 12 run PBD with eight factors and 2-levels fractional factorial design was applied to study N-1 variables using N experimental runs, where N is a multiple of four. In Table 2.9, the letter L and H refers to the "low value" and "high value" for each process factor, respectively.

Run	Α	В	C	D	Е	F	G	Н
1	L	L	L	L	L	L	L	L
2	Н	L	Н	Н	L	Н	Н	Н
3	L	L	L	Н	L	Н	Н	Н
4	Н	L	L	L	Н	L	Н	Н
5	Н	Н	L	Н	Н	Н	L	L
6	L	Н	L	Н	Н	Н	L	L
7	Н	Н	Н	L	L	L	Н	L
8	L	L	Н	L	Н	Н	Н	Н
9	Н	L	Н	Н	Н	L	L	L
10	Н	Н	L	L	L	Н	L	Н
11	L	Н	Н	L	Н	Н	Н	L
12	L	Н	Н	Н	L	L	L	Н

Table 2.9: Layout of Plackett-Burman matrix for 12 runs screening 8 factors (A-H).

2.12 Accelerated Lifetime Testing

A challenging issue, which has not received much attention, is the stability of AgNWs over long periods. For example, silver is known to be prone to electromigration [186], where the flow of current causes the slow movements of ions in a conductor. It is also observed that AgNWs are unstable at temperatures as low as 200 °C [187] and are affected by light exposure and environmental conditions.

Long-term stability of manufactured devices is very important and the best method to check or test their stability is to subject devices to different accelerated environmental conditions such as high temperature/humidity or high voltage/current. Accelerated Lifetime Testing (ALT) is needed at various stages of a product development cycle, for example, when identifying optimal material sets, providing relative comparisons of device stability [188] and also information on the product's failure mechanisms [189]. As devices are applied in more commercial applications, there is a greater need to predict the expected life in the outdoor operational environment. For the longer term, companies trying to commercialize this technology would also need predictive ageing to estimate warranty provisions. An alternative technique is to use life test models based upon quantitative ALT to address lifetime and try to quantify the degradation through the application of a mathematical model.

2.12.1 Quantitative Accelerated Life Tests (QALT)

For all life tests, some time-to-failure information (or time-to-an-event) for the product is required since the failure of the product is the event that needs to be understood [185]. Accelerated life testing is the process of testing a product or device by subjecting it to conditions (stress, strain, temperatures, voltage, vibration rate, pressure, etc.) above its normal conditions to uncover faults and potential modes of failure in a short amount of time. Performed correctly, accelerated testing can significantly reduce test times, resulting in reduced time to market, lower product development costs.

There are several types of accelerated testing approaches and the accelerated testing strategy must be carefully designed to fit the product under consideration. Quantitative accelerated life tests (QALT), on the other hand, are designed to quantify the life of the product and to produce the data required for accelerated life data analysis. This type of test involves the controlled application of accelerated stress conditions to stimulate product failure and provide life data more quickly. The life data obtained from these tests can be used to estimate a probability density function for the product under

normal use conditions and to calculate reliability, probability of failure, mean life, failure rate, and other important reliability metrics for the product.

2.13 Electrical and Environmental effects in AgNW Films

Electrical current flow in conductors can cause their temperatures to rise if resistance is high. If the temperature approaches the melting point, the conductor will degrade. This is a particular problem in AgNWs as the diameters are very low and current densities are high, therefore understanding the electrical stability of AgNW films is of importance. Most of the previously reported works on the electrical failure of AgNWs focused on the observation of failure under current flow, without a solution offered for how to mitigate the phenomenon. For example, Khaligh *et al.* demonstrated that when AgNW electrodes conduct current at levels encountered in organic solar cells, the electrodes can fail within two days [190]. The reason for electrode failure is the Joule heating effect which causes the nanowires to breakdown and thus produces an electrical discontinuity in the nanowire network [191].

More heat is created at the localised junction of AgNWs, and consequently, failure occurs sooner in more resistive electrodes and at higher current densities [192]. Heat in conductors is transferred to other components and insulation materials, primarily by conduction and convection, so thermal damage can be caused to these. High currents can also cause component parameter values, such as resistance, to drift over time. This effect is also accelerated by high operating temperatures.

Another important issue for the failure of silver nanomaterials is that they are susceptible to oxidation or sulfidation under normal environmental conditions. For example, Jiu *et al.* reported that AgNW electrodes can be easily damaged in the air due to sulfidation or oxidation based on humidity and harmful gases [193]. Hence, for wide applications of AgNW films over long durations, one still needs to design suitable test conditions and search for suitable coating materials for AgNW TCE encapsulation.

2.14 Summary

This chapter presented the fundamental properties of AgNWs, metal oxides semiconductors and the basic operation principle of TFTs. TCEs based on various materials, applications and their post processing were reviewed. Typical TFT four possible device structures were discussed. Materials properties for optically transparent polymer/plastic film substrates, dielectric and TFT active layers were detailed. State-of-

the-art performances in metal oxide TFTs were reviewed. Process optimization techniques, Taguchi methods, Plackett-Burman design were discussed. Accelerated life testing and electrical effects in AgNW films were reviewed. The long-term stability of manufactured devices is very important and the best method to check or test their stability is to subject devices to different accelerated real-world environmental conditions.

Chapter 3. Experimental Methods Used for Fabrication and Characterization

This chapter presents details of the materials and methods used during this Ph.D. work. The methods and experimental setups used for manufacturing silver nanowire (AgNW) transparent conductive electrodes (TCEs), ZnO TFTs, IGZO TFTs, high-k Al₂O₃ dielectric and their measurements are described. Further, IV characterization of TCEs, TFTs and the density of states is described. Also, experimental setups for accelerated lifetime testing (ALT) of TCEs, TFTs and post processing of TCEs are presented.

3.1 AgNWs solution Preparation

Three types of AgNWs were used for the preparation of AgNWs solution, namely thin short nanowire; AgNWs-60S ($d_{mean} = 60$ nm, $L_{mean} = 20-30$ µm), thin long nanowire; AgNWs-30L ($d_{mean} = 30$ nm, $L_{mean} = 100-200$ µm) and thick long nanowires; AgNWs-100L ($d_{mean} = 100$ nm, $L_{mean} = 100-200$ µm). All AgNWs dispersions were prepared in the clean room at room temperature. All AgNWs materials were diluted with ethanol to the concentration of 0.4 mg/ml for preparation of AgNWs solution. All AgNWs solutions were ultrasonicated just prior to spray coating on PEN substrates and this was done in order to ensure that the AgNWs were dispersed properly in the suspension before spray coating. A short duration was used (<5 minutes) at low power in order to ensure no breakage of AgNWs.

3.1.1 Materials and AgNW Solution Preparation

AgNWs (silver purity 99.5%, concentration 20 mg/mL in ethanol) were purchased from ACS materials. Flexible transparent substrates of polyethylenenapthalate (PEN) with thickness 125 µm were used for manufacturing TCEs. Polymethylmethacrylate (PMMA, molecular weight 495K, 8% solution in anisole) was purchased from Microchem. For this work, a PMMA buffer layer was employed to reduce the surface roughness of the substrates. High performance liquid chromatography (HPLC) grade isopropyl alcohol (IPA), ethanol (EtOH) and acetone were purchased from Sigma-Aldrich and used for cleaning of substrates. Deionized (DI) water was used from the in-house facility of Bangor University. Nickel shims were used for the embossing process explained in section 3.2. Zinc oxide nanoparticles of average particle size less than 100nm were purchased from Sigma-Aldrich and used to further improve surface properties of TCEs.

Shown in Table 3.1, the list of all selected materials used for the TCE preparation.

Туре	Materials	Diameter	Length	Purity	Concentration	Thickness
		(nm)	(µm)	(%)	(mg/mL)	(µm)
Silver nanowires	AgNWs-60S	60	20-30	99.5	20	-
	AgNWs-30L	30	100- 200	99.5	20	-
	AgNWs- 100L	100	100- 200	99.5	20	-
Nanoparticle	ZNO NP	< 100	-	-	-	-
Solvents	IPA	-	-	98	-	-
	Ethanol	-	-	95	NA	-
	Acetone	-	-	99.9	NA	-
Substrates	PEN	-	-	-	-	125
	ITO on PET	-	-	-	-	127
Buffer layer	PMMA	-	-	-	-	10
Anti-sticking solution	BGLGZ83	-	-	-	-	-
Stamp	Ni Shim	-	-	-	-	200

Table 3.1: Selected materials used for TCEs preparation.



Figure 3.1: AgNW TCEs preparation flow chart, showing process steps of spray coating and post processing of AgNW TCEs. R_{sh} is sheet resistance, T is transmittance, and H is % haze of TCE and PV, Pa and Pq are roughness parameters peak to valley, average roughness, and RMS roughness respectively.

3.1.2 Fabrication Steps of AgNW Electrodes

Shown in Figure 3.1, is the flow chart of fabrication steps for AgNW TCEs. PEN film was cut into 15×18 mm² substrates and thoroughly cleaned with DI water, acetone and IPA, followed by drying with a jet of dry nitrogen. For this work, a PMMA buffer layer was spin coated onto the PEN substrates at 2000 rpm followed by annealing at 100 °C for 1 minute. AgNWs were spray coated onto the substrates. Finally, post processing steps; thermal embossing, N₂ plasma and photonic sintering were performed.

3.1.3 Experimental Setup for Spray Coating AgNWs

In this thesis, the spray coating technique has been used for AgNWs film deposition. This method is inexpensive, easy to use and does not require any complex equipment. Furthermore, using this method, the AgNWs solution can be distributed uniformly on the substrate. Initial spray deposition was performed using hand spray coater. Later spray coating was conducted using a custom-built system with ultrasonic spray atomiser supplied by MTI in South Korea. The substrates were placed on the substrate heater at 60 $^{\circ}$ C and the AgNWs dispersion was sprayed onto the heated substrates to evaporate the ethanol carrier. The deposition time for preparing the initial electrode film with R_{Sh} \approx 25 Ω sq⁻¹ was 5 minutes ±5 seconds. Composite ZnO NP/ AgNWs transparent electrodes were manufactured by spray coating nanoparticles onto post-processed AgNWs films.



Figure 3.2: (a) Schematic illustration of spray deposition setup, (b) hand spray deposition setup, (c) ultrasonic spray coater, power supply, and atomizer.

Shown in Figure 3.2 (a) is the schematic illustration of spray deposition setup, (b) hand spray deposition setup, (c) ultrasonic spray coater, power supply and atomizer. For the AgNWs spray coating, a back pressure of the air was kept at 1 bar. A piston-type oilless airbrush compressor was used to achieve a uniform gas/liquid flow rate (0.2 ml min⁻¹) at the spray nozzle. The spray nozzle diameter was 0.5 mm and the fluid cup capacity was 7 cm⁻³.

3.2 AgNWs Film Deposition and Post Processing Approach

In this work three post-processing methods were sequentially applied onto the AgNW TCEs to achieve high performance; thermal embossing, Nitrogen (N₂) plasma treatment and photonic sintering of electrodes. Figure 3.3 shows the fabrication procedure for applying these treatment processes. All three post processing techniques are undertaken sequentially, but these are all transferrable onto a R2R system.



Figure 3.3: AgNWs film deposition and post processing approach: thermal embossing, photonic sintering, N_2 plasma treatment are used for post-treatment.

3.2.1 Embossing Experimental Setups for AgNW TCEs

AgNW TCEs were thermally embossed to enhance electrical and surface smoothness properties. For this purpose, two experimental setups were used; firstly, an in-house embossing machine is shown in Figure 3.4 and secondly a nano imprint lithography (NIL) system manufactured by Obducat, as shown in Figure 3.5.



Figure 3.4: (a) In house embossing machine from left (1) Fan for cooling, (2) embosser with having sample holder pneumatic pressing arrangements and (3) temperature controller, (b) Stack of components used for embossing TCEs and (c) image of nickel shims.



Figure 3.5: (a) Nanoimprinter Obducat 2.5 used for embossing TCEs and (b) Nano imprinter typical profile of controlled embossing process parameters: time, temperature and pressure.

The in-house embossing machine consisted of a fan for cooling down the hot plate, two flat and heated aluminium plates for embossing the samples and a Clarke CBJ2B 2 tonne bottle jack having automatic overload protection was used for creating pressure on plates. The 'stack' needed for the thermal embossing process was optimised and consisted of the following components; (1) TCE coated PEN substrate, a glass square piece, a nickel shim, a PTFE square piece, which were placed on the lower plate of embossing machine.

All components on the hot plate were heated to 140 °C and a 20 minutes wait time (settling time) was applied. The pressure was held for 20 minutes. Finally, the hot plate was cooled to 60 °C using fan air cooling and the pressure was released from the stack. Individual components of the stack were gently separated from each other to avoid any damage to AgNW TCEs. Thermal embossing was used to compress AgNWs on the surface of the substrate. The post-processing leads to improvements in surface roughness, sheet resistance, however, there is no change in transmittance and haze. The thermalembossing process was conducted using a 2″ nickel shim (shown in Figure 3.4 (c)), supplied by Nanotypos in Greece, formed from a polished silicon wafer to ensure flatness. The anti-sticking layer BGL-GZ-83 was spin coated onto the nickel shims before use in the thermal embossing procedure. The nickel shim was cleaned thoroughly using solvents before use.

Specification	In-house system	NIL2.5
Max temperature	200 °C	250 °C
Max pressure	-	70 Bar
Heat up time to 140 °C	3 min	2 min
Cool down time from 140 °C to 60 °C	15 min	10 min
Max substrate size	4 inches	2.5 inches
Loading	Manual	Manual
Unloading	Manual	Manual
Substrate-cooling capability	Air-cooling	Liquid N ₂ / Air-cooling
Software control	No	Lab-view controlled

Table 3.2: Specifications of in house and Obducat NIL 2.5 embossing systems.

A summary of the two embossing systems is shown in Table 3.2. Sheet resistance, haze and transmittance were measured before and after application of thermal embossing using nano imprint lithography (NIL) system. Moreover, thermal embossing of AgNW TCEs was performed at predetermined process conditions according to L₉ OA as discussed in the next Chapter 4.

3.2.2 Post Processing using Nitrogen Plasma and Photonic Curing

The nitrogen plasma treatment process was conducted using a Diener second 'Nano' plasma treatment system shown in Figure 3.6 (a) at ambient pressure after 30 seconds of nitrogen purge. Three gases namely, nitrogen, argon and oxygen were used initially for plasma treatment on AgNWs TCEs. There was approximately a 50 percent reduction in

resistance after the plasma treatment of nitrogen as well as argon gas. However, in the case of oxygen plasma treatment, the AgNWs TCEs became non-conductive. Why N2



Figure 3.6: (a) Diener second 'Nano' plasma treatment system, (b) Photonic sintering setup and typical profile of photonic curing system.



Figure 3.7: (a), (b) Illustration of Photonic curing process, (c) Typical Single pulse (300 μ sec length) thermal profile.

A Novacentrix Pulse Forge 1200 photonic curing system shown in Figure 3.6 (b) was used for sintering only the top surface of AgNW films. Photonic curing was performed using carefully timed and controlled flash lamps to heat only the surface and not the entire thickness of the material. The required exposure time is usually less than 1 millisecond using this setup. It allows high temperature, high performance inks to be used on low temperature, low cost, flexible substrates. Pulse conditions (>10 parameters) are carefully tuned to each material application. Photonic curing can be performed using single or multiple pulses. Figure 3.7 (a) and (b) shows the photonic sintering process. Figure 3.7 (b) demonstrates that as the distance from the surface of a thin film is increased

the temperature of photonic curing decreases (thus affects only the top of the surface). A typical single pulse profile is shown in Figure 3.7 (c)

3.3 TFT Fabrication

During this Ph.D. work, various techniques were used for the fabrication of TFTs based on inorganic metal oxide materials. Two methods of active layer deposition were used, one of these is spray pyrolysis and other being a sputtering deposition. Gate dielectrics were produced using anodization and electron beam evaporation. TFTs were also manufactured on dry SiO₂ wafers (200µm thick) purchased from Micro Chemicals.

3.3.1 Materials used for TFT Preparation

In this work, ZnO and IGZO were used as the semiconductor materials while SiO₂ and Al₂O₃ were used as the gate insulator. Kapton polyimide (PI) film substrates were used for manufacturing flexible IGZO TFTs, discussed in chapter 6 in detail. Shown in Table 3.3 are the materials used for TFT manufacture.

Туре	Materials	Stoichiometry	Length	Purity (%)	Concentratio	Thickness
			(µm)		n (mg/mL)	(µm)
TFT active	ZnO	-	20-30	99.9	20	0.02
layer	IGZO	1:1:1	100-200	99.9	20	0.02
Dielectrics	Sio ₂	-	-	-	-	200
	Al ₂ O ₃	-	-	-	-	0.02 - 0.1
Solvents	IPA	-	-	98	-	-
	Ethanol	-	-	95	NA	-
	Acetone	-	-	99.9	NA	-
Substrates	Kapton (PI)	-	-	-	-	25
	ITO on PET	-	-	-	-	127
	PEN	-	-	-	-	125

Table 3.3: Materials selected for TFT manufacture.

3.3.2 ZnO Precursor Solution Preparation and Deposition

In the case of ZnO, Zinc acetate dihydrate is dissolved in methanol at a 3% (w/w) concentration for making the precursor solution. Spray coating deposition was carried out using an airbrush nozzle actuated by a micro-controlled servo motor to spray a zinc acetate dihydrate solution onto pre-heated (350 °C) substrates. The substrate heating promotes the solvent evaporation as the solution spray reaches the substrate surface and causes the zinc acetate pyrolysis and subsequent formation of a thin ZnO layer. Spray deposition was carried out at air pressure 0.7 bar and nozzle kept 20 cm distance from the substrate on a hotplate. For optimum film formation, the airbrush nozzle was actuated for 3 short intervals of 5 seconds within three 60s pause intervals. The resulting ZnO films

obtained by this deposition procedure were very uniform, with an average thickness of 20 nm (determined by AFM).

3.3.3 Fabrication Steps for Oxide Semiconductor with Solution Process and Radio Frequency (RF) Sputtering

In this section two deposition processes; spray pyrolysis and sputter coating are described. Shown in Figure 3.8, are the TFT fabrication steps as follows: 1) Gate metal deposition, 2) gate insulator deposition, 3) channel deposition, 4) post-deposition annealing, 5) source/drain contact deposition. Heavily doped silicon-SiO₂ substrates of area 10×10 mm² were thoroughly cleaned (to remove contamination on substrate surface) with DI water, acetone and IPA, followed by drying with a jet of dry nitrogen and dehydrated at $100 \, {}^{0}$ C for a minimum of 20 minutes. The substrates are then treated with an oxygen plasma for 5 minutes to eliminate any remaining impurities. ZnO active layer is then deposited on SiO₂ substrates using an airbrush spray-pyrolysis system onto the pre-heated substrates.



Figure 3.8: (a) Fabrication steps for oxide semiconductor solution process (b) Optical microscope image (top view) of bottom gate ZnO TFT manufactured in this work using solution process (c) ZnO solution formulation.

The sputter coating method was also deployed for the deposition of oxide semiconductors. Sputtering is a physical vapour deposition (PVD) technique and relies on

the bombardment of a target of the source material with high energy inert ions. The ions impact generates a series of collisions between atoms of the target, leading to the ejection of some of these atoms. Sputtered atoms condense on the substrate and coat a thin layer. Argon (Ar) was generally used as the primary process gas. A characteristic feature of sputtering is the glow discharge cloud which appears between the target and substrate as shown in Figure 3.9 (a). RF sputtering can be used with both conductive and insulating targets. The frequency used for RF sputtering used was 13.56 MHz. The process parameters that control grain structure are substrate type/temperature, base pressure, deposition temperature, deposition rate, and process pressure (controls number of collisions). The physical nature of this process allows its use with virtually any existing source material. Shown in Figure 3.9 (b) Image of Leybold UNIVEX 350 sputter coater (left) and E-beam thermal evaporator (right). In this thesis work, RF sputtering was used for ZnO, IGZO active layer deposition and E-beam thermal evaporator was used to deposit Al₂O₃ high-k dielectric layer.



Figure 3.9: (a) RF sputtering mechanism Ionized Ar atoms diffuse from the glow discharge region into the cathode sheath, then are accelerated toward the ZnO/IGZO target via a large electric field. (b) Image of Sputter coater (left) and e-beam thermal evaporator (right).

3.3.4 Source Drain Contact Thermal Evaporation

Source, drain and gate contacts were deposited using thermal UNIVEX 350 Leybold thermal evaporator. Thermal evaporation is also a PVD technique commonly used for depositing metals layers like Al and is a much simpler process than sputtering. Sublimation of the source atoms is accomplished by bringing a metal bit to its respective melting point with the application of a large current to a resistive thermal tungsten boat or filament coil. Shown in Figure 3.10 (a) are the typical prerequisites of a thermal evaporation setup. Figure 3.10 (b) shows the image of the Leybold thermal evaporator. Thermal evaporation was accomplished in a high vacuum (10⁻⁶ - 10⁻⁷ torr). Atoms evaporated from the bit producing vapour pressure travel on a ballistic trajectory from the molten source material to the substrate surface. A high vacuum is required to minimize collisions of source atoms with background species. It traverses the chamber and hits the 'cold' substrate surface thus desublimates and coats the substrate. For better uniformity of deposited layer low evaporation rate of 1 nm per second was used.



Figure 3.10: (a)Schematic illustration of a thermal evaporation system showing all important parts, (b) Image of Leybold thermal evaporator.



Figure 3.11: Lift-off process defining the metal electrodes. (a) Starting with IGZO layer on SiO_2/Si substrates, (b) spin-coating the bi-layer AZ1505 +ve photoresist on top of LOR-5A resist, (c) UV exposure through mask, (d) developing the bi-layer in AZ726 MIF, (e) thermally evaporating Al with Cr adhesion layer, and (f) lift-off the bi-layer to define the metal electrodes.

3.3.5 Lift-Off Process

In addition to shadow masks, small channel (5-100 μ m) TFTs were also fabricated on SiO₂ substrates using the photolithography process. The bi-layer lift-off process was used to define the electrodes, as depicted in Figure 3.11. First 500 nm LOR-5A resist then 500nm AZ1505 positive-tone photoresist films were spun-coated over the IGZO layer on top of the SiO₂/Si substrates. The bi-layer was then UV-exposed with ~112 mW/cm² to define the contact electrodes. The pattern was transferred to the bi-layer using the commercial AZ726 MIF developer. 100 nm of Al was thermally evaporated on the patterned photoresist using a Leybold 250 system, with 7nm of Cr as an adhesion layer. Finally, the bi-layer was removed in 1165 photoresist stripper using a sonication bath to create the Al/Cr contact electrodes.

3.4 Characterization of TCEs and TFTs

In this section, all electrical and optical characterization methods used for TFTs and AgNW TCEs are explained.



Figure 3.12: (a) Schematic diagram of the MIM CV/Cf and capacitance characterization setup. Test probes were connected to BNC connector to allow connection to the precision LCR meter.

3.4.1 TFT CV/Cf and MIM Capacitance Characterization

All metal insulator metal (MIM) CV/Cf and capacitance measurements were performed using Hewlett Packard 4284A 20Hz-1MHz precision LCR meter and Solartron SI 1255 HF frequency response analyser source measure unit connected to three probe tips mounted on micromanipulators. All measurements and data acquisition were automated using MATLAB software. The schematic diagram of MIM CV/Cf and capacitance characterization setup is shown in Figure 3.12.

3.4.2 UV Saturation Experiment Setup

All UV sensing saturation experiment (detailed in chapter 5) measurements of ZnO TFTs as photo sensor were performed using Agilent B2900A precision source measured unit connected to three probe tips mounted on micromanipulators. All measurements and data acquisition were automated using B2900A Quick IV measurement software and Keithley 237 was connected to source UV LED. The schematic diagram of UV saturation experiment is shown in Figure 3.13.



Figure 3.13: (a) Schematic diagram of UV saturation experiment characterization setup. Test probes were connected to BNC connector to allow connection to the Agilent B2900A precision SMU and Keithley 237 was connected to source UV LED.

3.4.3 TFT Current Voltage (IV) Characterization

All TFT IV measurements were performed using Agilent B2900A or Keithley 4200 precision source measure unit connected to three probe tips mounted on micromanipulators. All measurements and data acquisition were automated using B2900A Quick IV measurement software or TSPTM Express software through a LAN

connection. The schematic diagram of the transistor characterization setup is shown in Figure 3.14.



Figure 3.14: (a) Schematic diagram of the transistor characterization setup. Test pins were connected to BNC connector to allow connection to the SMU unit, (b) Image of probe station used for TFTs IV characterization.

3.4.4 IV Characterization and Accelerated Lifetime Testing (ALT) of AgNW TCEs

Shown in Figure 3.15 are the experimental setups for accelerated lifetime testing of AgNW TCEs, (a) environmental chamber (b) circuit board for ALT showing AgNW TCEs connected to crocodile clips, power supply, and Bo test SMU.



Figure 3.15: Experimental setup for accelerated lifetime testing of AgNW TCEs. (a) Environmental chamber (b) Circuit board for ALT showing AgNWs samples connected to crocodile clips, power supply, and Bo test SMU.

To evaluate the long-term stability of the post processing upon the AgNWs stability, a bespoke accelerated life testing (ALT) system was constructed. This enabled high current biasing of the AgNW electrodes and repeated resistance calculations every 30 minutes, during which the bias momentarily switched off (~5 seconds). To accelerate the degradation, samples were placed in an environmental chamber (Weiss UK ltd). Simultaneous high bias and temperature degradations were conducted at varying stress levels so that the stability could be assessed under 'normal operation conditions', which was defined as $I_{Bias} = 20 \text{ mA/cm}^2$ and an ambient temperature of 20 ^oC, which is typical settings for an OLED or OPV.



Figure 3.16: Circuit design for accelerated lifetime testing AgNW TCEs, first and last units are shown for simplicity because all units are the same.

The circuit of the developed ALT is shown in Figure 3.16 was designed and implemented on the printed circuit board. The main electronic components used were relays, bipolar junction transistors (as switching transistors) and diodes (as current

limiters). For simplicity only first and last units (out of eight) are shown because each unit is a replica of other units. The Botest software was controlled by an auto clicker, which starts a new measurement cycle every 30 seconds for eight TCEs connected to SMU via a multiplexer. The measurement cycle runs a full IV sweep on each TCE connected to the multiplexers in turn.

3.4.5 Setup for Sheet Resistance Measurement of AgNW TCEs

Sheet resistance measurements were conducted using the four-probe setup as shown in Figure 3.17 (a). This system is a combination of the multi height probe stands with the RM3000 Test Unit, with the added feature of a removable X-Y micro position table. Figure 3.17 (b) shows a schematic of 4-point probe configuration. Typical probe spacing s is approximately 1 mm. Each probe tip is supported by springs on the other end to minimize sample damage during probing. A high impedance current source is used to supply current through the outer two probes and a voltmeter is used to measure the voltage across the inner two probes (See Figure 3.17 (b)) to determine the thin layer resistivity. For a very thin layer (thickness t << s), the R_{sh} is given by equation 2.1 (Chapter 2).



Figure 3.17: (a) Jandel multipurpose four-point system and (b) Schematic of 4-point probe configuration.

3.4.6 Atomic Force Microscopy

Atomic force microscopy (AFM) was used to measure film thicknesses by scanning across a scratch made using the sharp edge of Knife in the layer. AFM was also used to measure average, RMS and peak to valley surface roughness of device layers. These measurements were performed using a Digital Instruments Nanoman V running in tapping mode.

3.4.7 Scanning Electron Microscopy

Scanning electron microscopy (SEM) was used to obtain high resolution surface and cross section imaging of fabricated TCEs layers. Scanning electron microscope (SEM) images were obtained using a Leo 1455VP SEM operating at 30 kV with 10 pA beam current at a typical magnification of ×10000.

3.4.8 Optical Characterization

Figure 3.18 (a) shows the Ocean Optics HR4000 UV-Vis-NIR spectrometer and (b) Shimadzu UV-3600 spectrophotometer used in this work. UV-Vis spectrophotometry was used in the absorption peak measurements of AgNW films, ZnO, IGZO active layer (Shimadzu UV-3600 spectrophotometer). The Shimadzu UV-3600 spectrophotometer has three detectors with a resolution of 0.1 nm and has a wavelength range of 185-3300nm. The spectrometer used for in-situ absorption measurements was an HR4000 UV-Vis-NIR supplied from Ocean Optics Inc. The unit has a wavelength range of 350-1100nm. This device requires the reference to be taken before the absorption measurement.



Figure 3.18: (a) Ocean Optics HR4000 UV-Vis-NIR spectrometer and (b) Shimadzu UV-3600 spectrophotometer.

3.4.9 White light interferometry (WLI) and haze measurements

The surface roughness is an important factor for the optimum performance of thin-film electronic devices such as TFTs, OLEDs, and OPVs. To characterise the surface roughness of thin films, white light interferometry was used. Shown in Figure 3.19 is the experimental setup for surface roughness measurement using Micro XAM. Surface roughness was estimated by white light interferometry (WLI) using a Micro XAM surface mapping microscope (KLA Tencor, USA).



Figure 3.19: (a) Experimental setup for surface roughness measurement using Micro XAM and (b) Image of haze meter used in this work and (c) Four scan configurations for haze measurement [194].

A haze meter was used to measure the amount of light that is scattered when passing through AgNW TCE. Shown in figure 3.19 (b) is the image of the haze meter used in this work. The haze found to be 0.3% for AgNW TCE with 24.1 Ω /sq sheet resistance.

Haze measurement involves four scans using the configurations shown in Figure 3.19 (c). In configuration T1 exit from an integrating sphere is closed with white standard, while in configuration T2 additionally a sample is placed at the entrance to the sphere. In configuration T3 light can pass through the sphere undisturbed and in configuration T4 it is scattered by the sample at the entrance to the sphere.

The integrate scans used in two spectral ranges: all visible range (380-780nm) and the range where samples do not absorb (600-780nm). The integrated area under each spectrum was used in equation 3.4 [192]:

$$Haze = \left[\frac{T4}{T2} - \frac{T3}{T1}\right] * 100\%$$
 Eq. 3.4

3.4.10 Bias Stress Characterization

The positive gate bias stress (PGBS) characterization procedure described here is relevant for the studies in Chapter 6. A typical flow diagram for the bias stress experiments is shown in Figure 3.18(a).

First, the initial transfer characteristic of IGZO TFT was measured. Then the device was subjected to various bias stress conditions for 11000s stress time. The bias stress was frequently interrupted to measure the output and transfer characteristics in saturation ($V_D = 40$ V) regime from which the transistor parameters were extracted.

Figure 3.20 (b) presents a typical transfer curve, plotted on linear and semi-log scales, both in the initial state and after bias stress. The shift of threshold voltage, ΔV_T , was determined as $[V_T (t) - V_T (0)]$, where t=0 corresponds to pre-stress values.



Figure 3.20: (a) Flow chart of a typical bias stress experiment. (b) Illustration of threshold voltage shift, ΔV_T , calculated as $[V_T (t) - V_T (0)]$, where t=0 corresponds to pre-stress values.

After completing a positive bias stress test, devices were subjected to a recovery procedure. The devices were left short-circuited in the dark for a 21minutes or under white light for a few seconds (180s) until V_T recovered to its original value. Due to fast recovery under white light, recovery of all devices was performed using white light.

3.5 Extraction of the Density of Trap State Using the Grünewald Model

The extraction of Density of States (DoS), from the transfer characteristic of a thin film transistor, was performed using the Grünewald *et al.* model [195]. This was accomplished by first developing the relationship between the gate dependent insulator/semiconductor interface potential, V_0 , and the effective forward voltage, $V_F = (V_G - V_{ON})$. This is achieved by numerically solving the equation 3.5.

$$exp(\beta V_0) + \beta V_0 - 1 = \beta \frac{C_i l}{\varepsilon_s \varepsilon_0 I_0} \left[V_F I(V_F) - \int_0^{V_F} I(V_F) dV_F \right] \qquad \text{Eq. 3.5}$$

where, *l* is the thickness of the semiconductor, $\beta = q/kT$, k the Boltzmann, T the absolute temperature, q the charge of the carrier, ϵ_S the relative permittivity of the semiconductor and ϵ_0 the permittivity of free space. I₀ is the off-current at V_{ON}. Following further manipulation, the total electron density n(V₀) and the density of states N(E) may be obtained from equation 3.6

$$n(V_0) = \frac{C_i^2}{\varepsilon_s \varepsilon_0 q^2} V_F \left[\frac{dV_0}{dV_F}\right]^{-1}$$
 Eq. 3.6

Finally, the density of states, N(E) from equation 3.7,

$$N(E) \approx \frac{dn(V_0)}{dV_0}$$
 Eq. 3.7

Chapter 4. Application of Thermal Embossing to Enhance the Properties of AgNW TCEs

This chapter reports a low-cost method for manufacturing silver nanowire (AgNW) transparent conductive electrodes (TCEs). The initial optimisation and production processes are explained at the start of section 4.1. The report of AgNWs TCEs shows remarkable electrical performance, surface planarity, and environmental stability. This process relies on three sequential steps, which are roll-to-roll (R2R) compatible; thermal embossing, infrared sintering, and plasma treatment. This process leads to the demonstration of a conductive film with a sheet resistance of 2.5 Ω sq⁻¹ and high transmittance, thus demonstrating the highest reported Figure-of-merit in AgNWs to date (FoM = 933). A further benefit of the process is that the surface roughness is substantially reduced compared to traditional AgNW TCEs processing techniques. Finally, consideration of the long-term stability is given by developing an accelerated life test process that simultaneously stresses the applied bias and high temperature. Regression line fitting shows that approximately 150-times improvement in the stability is achieved at 'normal operational conditions' when compared to traditionally deposited AgNW films. X-ray photoelectron spectroscopy (XPS) is used to understand the root cause of the improvement in long-term stability, which is related to reduced chemical changes in the AgNWs.

4.1 Initial AgNW TCE Fabrication and Optimisation

Initial AgNW TCEs manufacturing work and optimisation began with spray coating AgNW dispersions in ethanol with different volumes and different concentrations on flexible PEN substrates. To obtain high transmittance with low sheet resistance a number of input factors like AgNW dispersion flow rate, the angle between nozzle and substrate and distance of spray nozzle were varied as shown in chapter 3 Figure 3.2.

Figure 4.2 shows (a) transmittance as a function of wavelength for two separate AgNWs (D = 60nm) on PEN electrodes, (b) transmittance as a function of wavelength for two separate AgNWs (D =100-200nm) electrodes, (c) transmittance at 550nm as a function of sheet resistance for different diameter AgNWs types (d) sheet resistance as a function of amount of AgNWs deposited by spray coating on PEN, (e) transmittance as a function of amount of AgNWs deposited by spray coating on PEN and (f) % haze of TCEs

amount of AgNWs deposited by spray coating. It can be observed that both transmittance as well as sheet resistance decreases, and haze increases with an increase in volume or silver amount. This indicates that a trade-off between sheet resistance and transmittance is necessary, which is determined by the final application of AgNW TCEs. The data is supported by earlier studies in AgNWs, but Figure 4.2(c) is remarkable as the transparency of wider diameter AgNWs remains high even for lower sheet resistance. Where possible, it was decided to use these AgNWs in the work.



Figure 4.1: (a) Transmittance as a function of wavelength for two separate AgNWs (D = 60nm) on PEN electrodes, (b) Transmittance as a function of wavelength for two separate AgNWs (D = 100-200nm) electrodes, (c) Transmittance at 550nm as a function of sheet resistance for different diameter AgNWs types (d) Sheet resistance as a function of amount of AgNWs deposited by spray coating on PEN, (e) Transmittance as a function of amount of AgNWs deposited by spray coating on PEN and (f) % haze of TCEs amount of AgNWs deposited by spray coating.

4.2 Effect of Thermal Embossing on AgNW TCEs Properties

The effect of thermal embossing on electrical, optical and surface properties of TCEs was studied. Table 4.1 shows the results for sheet resistance, transmittance, haze, and surface roughness parameters obtained after measuring AgNW TCEs before and after application of thermal embossing. The percentage decrease in sheet resistance (R_{sh}) average

roughness (R_a), R.M.S roughness (S_q) and peak to valley roughness (P_v) were calculated. Average (Ave) and standard deviation (s.d) of various performance parameters for ten samples are shown in Table 4.1. Standard deviation is a number used to express how measurements for a group are spread out from the average, or estimated value. A low standard deviation means that most of the numbers are very close to the expected value. The standard deviation should be as low as possible.

Table 4.1: Average (Ave) and standard deviation (s.d) of Sheet resistance, transmittance and surface roughness parameters before and after embossing.

Statistics	R _{sh} before Ω ⁻ sq	R _{sh} after Ω ⁻ sq	T (%)	H (%)	Ra (nm)	Ra after (nm)	P _v (nm)	P _v after (nm)	R _q (nm)	R _q after (nm)	% age decrease in R _{sh}	% age decrease in R₁
Ave	39.38	22.48	87.12	0.4	14.68	8.03	163.49	68.32	18.59	9.79	42.59	44.70
S.d.	5.11	2.90	1.08	0.13	2.18	0.86	58.73	10.46	3.04	1.52	6.32	6.49



Figure 4.2: (a) Sheet resistance as a function of annealing temperature measured at different annealing times, (b) SEM image of AgNW showing polyvinylpyrrolidone (PVP) capping, (c) optical microscopic image (50 × magnification) of non-embossed AgNW film on PEN (d) optical microscopic image (50 × magnification) of embossed AgNW film on PEN, (e) optical microscopic image (20 × magnification) of non-embossed AgNW film on PEN and (f) optical microscopic image (20 × magnification) of embossed AgNW film on PEN.
It is worth mentioning that there is an average 42.59% reduction in sheet resistance and 44.70% reduction in the roughness average R_a after the thermal embossing step. Similarly, there is a reduction in other surface parameters, P_V and R_q as shown in Table 4.1. In AgNW TCEs the junction of the AgNWs is the source of high resistance and residual surfactant polyvinylpyrrolidone (PVP) capping can increase the sheet resistance of the AgNWs at these junctions. Shown in Figure 4.2 (a) is the reduction in sheet resistance as a function of annealing temperature measured at different annealing times. The reduction in sheet resistance on the application of heating and thermal embossing is probably due to the thinning down PVP layer shown in Figure 4.2 (b).

4.3 Taguchi Orthogonal Array Optimisation before and after Thermal Embossing Process

Pressure, time and temperature of the embossing step were all varied, and the process was optimised using the Taguchi Parameter Design method. All embossing operations were conducted under ambient conditions rather than under vacuum. Sheet resistance, haze, and transmittance were measured before and after the application of thermal embossing using nano imprint lithography (NIL) system.

A Design of experiment (DOE) software tool was used to assist and optimise the thermal embossing process. A Taguchi L₉ OA discussed in section 2.10.1 was chosen with 3 factors (ABC) and three level design was selected for estimating the effect of different factors; A=Pressure of embossing (5 bar, 15 bar, 20 bar), B=Time of embossing (30 seconds, 2 min, 10 min), C=Temperature of embossing (60 °C, 90 °C, 120 °C) on the sheet resistance of electrodes. AgNW TCEs were embossed at different conditions shown in Table 4.2 and 9 experimental runs. After performing all experimental runs sheet resistance and transmittance were measured. The sheet resistance of TCEs was changed, however no change in transmittance was observed, so this is not reported. The Taguchi OA Factorial settings and properties were divided based upon early trials and literature values and are shown in Table 4.2.

				Number	OA	Level	Level	Level
Factors	Name	Units	Туре	of Column		1	2	3
	Levels	Index	(Low)		(High)			
Р	Pressure	bar	Quantitative	3	1	5	15	20
Т	Temperature	⁰ C	Quantitative	3	2	60	90	120
t	Time	min	Quantitative	3	3	0.5	2	10

Table 4.2: Taguchi OA Factorial settings and properties of embossing factors.

The response, defined as the percentage change in sheet resistance (R_{sh}), was analysed using DOE software tool for the varying test runs. The main effect plots are shown in Figure 4.3 (a), (b), (c) and (d) the term effect plot extracted from DOE analysis. In the DOE, the main effect is the effect of an independent variable (input factors) on a dependent variable (output factors) averaging across the levels of any other independent variables.



Figure 4.3: Main effect plots(a) change in sheet resistance as a function of embossing pressure variation, (b) change in sheet resistance as a function of embossing temperature variation and (c) change in sheet resistance as a function of embossing time variation (d) term effect plot for change in sheet resistance as a function of embossing pressure, temperature and time.

Shown in Figure 4.4 (a) is the Pareto chart based on linear regression analysis of the main effects and (b) the Pareto chart for main effects and 2-way interactions calculated by considering the change in sheet resistance. The Pareto chart shows the standardized effect of each term (i.e. factor or combination of factors) for a particular response. The thick blue line shows the threshold value for the 'significance' test. If a bar is having a value lower than the significance value, we can define this factor as having a 'significant effect,' based upon a significance level of 0.05. It can be observed from 4.4 (a) Pareto chart regression that embossing pressure and temperature are the significant factors for reducing the sheet resistance in our process. It can be noted that pressure is most significant and of greater significance than temperature. This can be attributed to the better integrity of the electrical junctions created due to the mechanical pressing force. Figure 4.4 (b) shows the Pareto chart analysis of variance (ANOVA) for change in sheet resistance. It can be observed that there is no 2-way interaction between the factors.



Figure 4.4: (a) Pareto chart-regression (b) Pareto chart analysis of variance (ANOVA) for change in sheet resistance.

4.4 Effect of Photonic Curing and N₂ Plasma on AgNW TCEs

After thermal embossing, another post processing step of applying a N₂ plasma cleaning was carried out on AgNWs coated substrates. The treatment process was conducted using a Diener second 'Nano' plasma treatment system at ambient pressure after a 30 seconds nitrogen purge. In this system, a 13.56 MHz RF Generator was used at 100 watts power to generate N₂ plasma for 120 seconds. The effect of plasma cleaning was observed and sheet resistance and transmittance were measured before and after plasma cleaning. AgNW electrodes were Nitrogen (N₂) plasma treated for 2 minutes, which result in an average 39.8 % reduction of sheet resistance, although no change in transmittance was noted. This reduction in sheet resistance is attributed to the removal of the PVP layer [96]and the enhancement of the contact between AgNWs.

NIR photonic curing was performed using a Novacentrix system, mentioned in section 3.2.2 and Figure 3.6 (b) was used to rapidly anneal the surface of AgNW TCEs. This results in improvements in electrical conductivity due to the welding of AgNWs junctions. Various AgNW TCE samples were subjected to a photonic curing/sintering process to find appropriate voltage (450V) pulse. The sheet resistance across the AgNWs electrode was measured using four probe system described in chapter 3 before and after applying sintering voltage pulse. Samples were mounted to the system and exposure varied from 450V, 400µs to 750µs with single energy pulse 1488mJ/cm² to 2712mJ/cm². The process ensured sintering occurred within the top 1µm of the film surface, rather than the entire thickness of the substrate and therefore impacted mostly upon the AgNW film. By concentrating the sintering so close to the surface, flexible and polymeric substrates can be used, and surface damage can be minimised.

It was found that a pulse duration of 600 to 700 μ s was the most suitable for AgNW TCEs photonic curing because there is a maximum reduction in sheet resistance at around 700 μ s. Longer duration pulses than 700 μ s or multiple pulses (1800 μ s – 3 pulses) were not suitable because the polymeric substrate began to melt at sides of AgNW samples. Photonic sintering resulted in an average 40.8% reduction of sheet resistance, and no change in transmittance was noted. The individual and combined effects of post processing steps on properties of the AgNWs electrode are tabulated in the next section 4.5 Table 4.3.

4.5 Effect of Post Processing on Electrical and Optical Performance of AgNW TCEs

The overall impact of the individual post processing techniques and their combined effect is shown in Table 4.3. It can be seen that the optimised individual post-process of thermal embossing, sintering, and N₂ plasma treatment have a similar effect upon the reduction in R_{sh} ; leading to a reduction of ~50%, when compared to the control samples. By using all three post processing techniques sequentially, a substantial 93.4% reduction in the sheet resistance was obtained. Optimisation of the sequence of the different processing steps was conducted and the optimum sequence was (1) NIR sintering, (2) thermal embossing and (3) N₂ plasma treatment. Importantly, none of the post processing techniques show a compromise in the optical transmittance or the optical haze of the samples with no change observed in these parameters (shown in Table 4.3). As discussed earlier, a FoM can often be used in TCEs to compare performances (defined in equation 2.4). The FoM for the combined post processing technique shows a value of 933; this is substantially higher than the highest reported in the literature, 487.9, and represents the best performing TCE based upon AgNWs, or ITO-replacements. As a comparison, the performances are benchmarked against the incumbent technology (ITO) which is reported for both glass and PET substrates (Table 4.3). In both cases, the AgNW TCEs prepared with the post processing technique show significantly enhanced performance over the ITO.

Table 4.3: Electrical and optical performance before and after post-processing,	showing
changes in sheet resistance (Rsh) and optical properties and the 'Figure of Merit'	(FoM) is
shown and benchmarked against ITO.	

Process	R _{sh}	R _{sh}	Percentage	Optical	Optical	FoM	Average
	(Before	(After	reduction in	transmittan	haze	(After	Percentage
	post	Post	R _{sh}	ce @	(%)	Post	change in
	process)	process)	(%)	550nm		process)	FoM
	$\Omega \ { m sq}^{-1}$	$\Omega \ sq^{-1}$		(%)			
Embossing	40.4	18.1	55.2	92.9	0.2	278.9	120.8
Sintering	42.4	19.1	55.0	94.0	0.3	314.1	97.1
N ₂ plasma	41.4	20.1	51.4	92.7	0.3	242.8	90.2
Combined	37.6	2.48	93.4	89.5	0.2	932.9	518.4
Combined	37.1	2.98	81.7	85.2	0.4	758.7	498.6
and with							
ZnO NP							
coating							
ITO on PET	40.0	n/a	n/a	85.0	0.4	55.7	n/a
ITO on glass	12.0	n/a	n/a	84.0	0.5	172.4	n/a

4.6 Surface Topographic Properties

In this section, surface roughness parameters before and after post-processing are investigated. In the context of TCEs, the surface roughness of the sample is very important. In OPVs and TFTs, the low surface roughness of TCEs can lead to AgNWs penetrating through the active regions of the devices leading to electrical shorts.

4.6.1 Surface Topographic Properties Before and After Post Processing

The positive changes in surface roughness after post processing are reported in Table 4.4. The data shows that the root mean square surface roughness (R_q) is reduced to 3.6 nm from 6.4nm, the average roughness (R_a) is reduced to 2.7 nm from 4.9 nm and the maximum peak-to-valley to 60.8 nm from 105.9 nm after the post processing techniques. The surface roughness was benchmarked also against incumbent technology (also shown in Table 4.4) and it is evident that after post processing, there is a 21.5 to 45.4 percent reduction in the R_a , R_q , P_v surface roughness parameters.

Given the significant enhancements in electrical and surface topographic properties, it is necessary to discuss their origins. SEM images of AgNW TCEs before and after post embossing are shown in Figure 4.5 (a) and (b) respectively. Based on this, it can be said that the percolating network of AgNWs after embossing possess shows a more uniform height as the variation in SEM contrast is less obvious. Whilst topography can be somewhat difficult to evaluate using SEM, the images do show that the embossing step leads to improved physical contact between percolating AgNWs and is therefore likely to increase the number of electrical connections between AgNWs and integrity of the contact junctions. As the embossing temperature was low, it is unlikely there is any 'fusing' of AgNWs electrodes due to this alone. The reduction in SEM contrast indicates reduced secondary electron emission and is likely to be as the AgNWs are partially-embedded within the polymer substrate. The embedding of the AgNWs into the polymer surface is likely to contribute to the reduction in surface roughness as the top surface topography is determined by the smoothness of the nickel shim which determines the embossed surface smoothness, rather than the protrusions of the AgNWs. In Figure 4.5 (c), it is apparent that the AgNWs have compressed and are embedded into the PMMA layer and some appear to have fractured during the embossing stage. It is clear that the applied pressure and temperature need to be optimised as over pressurising during the embossing stage can lead to fractures which will diminish sheet resistance.

Table	4.4:	Surface	topographic	properties	before	and	after	post-pro	cessing,	showing
change	es in	average	surface roug	hness (Ra),	root-m	ean s	square	roughne	ess (Rq)	and Peak
to Vall	ey ro	oughness	(Pv) and ber	nchmarked a	against	ITO r	neasu	red using	g WLI.	

Sample	Ra	Ra	%	Rq	Rq	%	Pv	Pv	%
_	(Before	(After	Change	(Before	(After	Change	Peak to	Peak to	Change
	Post	Post	in Ra	Post	Post	in R _q	valley	valley	in peak
	process),	process),		process),	process),		(Before	(After	to
	nm	nm		nm	nm		Post	Post	valley
							process)	process)	Pv
							nm	nm	
AgNWs	6.8	3.7	45.4	8.5	4.9	43.0	101.4	79.6	21.5
Combined									
AgNWs	4.9	2.7	44.9	6.4	3.6	43.8	105.9	60.8	42.6
and with									
ZnO NP									
coating									
ITO on	3.3	n/a	n/a	4.7	n/a	n/a	52.2	n/a	n/a
PET									
ITO on	1.1	n/a	n/a	2.2	n/a	n/a	10	n/a	n/a
glass									

The physical changes that occur as a result of photonic sintering are also shown in the SEM image in Figure 4.5 (d, e). The data in Table 4.4 shows that the average R_{sh} reduces by ~50% after sintering. Based on the SEM images, the sintering process affects the AgNWs by melting and moderately deforming the AgNWs so that the formation of better electrical junctions between individual nanowires is achieved which leads to the enhanced conductivity in films. In previous work, high temperature processing has been shown to remove residue polyvinyl propyl (PVP) in the AgNWs, leading to improved chemical purity [58], which is also likely to contribute to the improvement in the R_{Sh}. Because the changes related to N₂ plasma treatment are most likely to be changed in the material chemistry, this is better discussed in section 4.11, where XPS studies of the surface chemistry are discussed. The N₂ plasma treatment is likely to also remove the PVP which results in the improvement in sheet resistance. The effect of zinc oxide nanoparticles to further improve surface properties is discussed in the next section 4.6.2.



Figure 4.5: SEM images of (a) AgNWs before embossing, (b) after embossing and (c) after embossing at higher resolution, (d) after embossing and sintered, (e) after embossing/sintering at a higher resolution. Cross sectional SEM images show (f) before embossing, (g) before embossing at higher resolution and (h) sample that has been embossed and sintered.

4.6.2 Application of Zinc Oxide Nanoparticles to Further Improve Surface Properties

After post processing, a zinc oxide (ZnO) nanoparticle dispersion in ethanol was spray coated directly on the post processed TCEs. Film thickness was measured using a cross-sectional SEM and was measured at 40±3 mm. By spray coating of the ZnO nanoparticles, a further planarization layer is deposited onto the AgNWs film, without significantly impacting on either R_{sh} or optical transmittance. This is best illustrated in the WLI data

shown in Figure 4.6, which shows surface roughness changes in a control sample, postprocessed sample and post-processed sample with ZnO overcoating. ZnO nanoparticles couple with AgNWs to form a composite electrode and it should further decrease in surface roughness. This is possibly due to filling up of vacate gap with ZnO nanoparticles and hence lowers the roughness parameter values.



Figure 4.6: Surface topography images using white light interferometry (WLI), (a) the control sample, (b) post-processed sample and (c) post-processed sample with ZnO overcoating.

The roughness is quantified in Table 4.4 (mentioned earlier) and the combined AgNWs-ZnO electrode is shown to possess roughness much lower than the control sample and is comparable to ITO on PET. It is clear from the WLI data that the ZnO fill up the voids created by the network of AgNWs and covers any remaining AgNWs that are protruding and thus improves the surface roughness. The result is significant for a second reason; for OPVs and OLEDs, ZnO is often used as an interlayer as the energy levels are favourable for electron extraction or hole injection (for OPVs and OLEDs, respectively). Therefore, the

ZnO layer has dual functionality as it planarises the AgNWs electrode and ensures energy alignment between the transparent electrode and the active region.

4.6.3 Demonstration of OPV Device

To demonstrate the suitability of the AgNWs electrode films, inverted Organic Photovoltaics (OPVs) were fabricated onto the developed TCEs. This work was performed with the help and support of my supervisor Dr. J. Kettle. To benchmark the AgNW TCEs, PSC devices were also made using ITO coated PET substrates ($R_{sh} = 60$ Ω /square, transmittance = 84% purchased from Sigma Aldrich) that were first cleaned using deionised water, acetone and isopropanol in an ultrasonic cleaner, then treated in a UV-ozone reactor with oxygen plasma for 10 minutes. The remaining device fabrication procedure was the same. The AgNW and ITO substrates were coated with ZnO with a film thickness of 80nm. Samples were transferred into a nitrogen atmosphere glovebox ([0₂], < 1ppm; [H₂O], <100ppm), where the active layers and contacts were applied. The absorber was deposited inside a glove box by spin-coating using the two-step procedure. Active layer BHJ blends using PTB7-Th (as a donor) and [6,6]-phenyl-C₇₁-butyric acid methyl ester (C71-PCBM) (as an acceptor) with weight ratios 1 : 1.5 were prepared and mixed with chlorobenzene solvent with a concentration of 30 mg mL⁻¹. Active layers were applied by spin-casting from a 45 °C solution. The active layer was annealed at 45 °C for 15 minutes before thermal evaporation of the cathode was performed through a shadow mask to define the device area and consisted of 8 nm of MoO₃ and 100 nm of silver (Ag) to form the cathode of the device.



Figure 4.7: Typical current–voltage characteristics for PTB7-Th:PCBM OPV using ITObased electrodes on PET (black) and glass substrates and AgNWs-based electrode (red) under 100mW/cm² AM1.5 illumination.

The measurement system used to characterize the devices consisted of a Newport solar simulator with 100 mWcm⁻² AM1.5G output (calibrated using a silicon reference cell from RERA in the Netherlands. The open circuit Voltage (V_{oc}), short-circuit current density (J_{SC}), fill factor (FF) and PCE values are averaged from twelve cells.

The performance was benchmarked against an ITO electrode manufactured on PET. Shown in Figure 4.7 are the IV characteristics under AM1.5G illumination from a set of 12 devices made with ITO or AgNW front electrodes, with the remainder of the device structure not changing during these tests. IV characteristics are shown for the device which was closest to the mean in both sets of devices, and the PV performance parameters and manufacturing yield are reported in Table 4.5

Table 4.5: IV characteristics of PTB7-Th: PCBM solar cells measured with 100 mWcm⁻² AM1.5G output. Performances and yield are measured from 12 devices.

Sample	Voc (V)	Jsc (mA/cm ²)	FF	PCE	Yield
ITO on PET	0.732	-16.3	0.56	6.70	100%
AgNWs on PEN	0.716	-16.8	0.58	7.00	100%

In terms of performance, the AgNWs shows a moderately improved PV performance; this is primarily due to an increase in short circuit current density (Jsc) of around 0.6 mA/cm² and Fill Factor of 2%, compared to the sample made on an ITO substrate leading to a drop in the PCE. One would expect the AgNWs device to have a higher FF than the device made on ITO-PET due to the lower Rsh, but the observed differences in the FF are minor. Considering the IV characteristics reported in Figure 4, the series resistance (Rs) of the AgNWs device is moderately better than that of the ITO-PET device, but there appears to be a decrease in shunt resistance (R_{sh}). This decreased R_{sh} is attributed to the increased shunts and parasitic pathways between AgNWs films and the PTB7-Th:PCBM active layer as the ZnO might not fully planarise the gap in-between the AgNWs and the active layer. In the case of a larger device or modules, the use of AgNWs would have an even greater impact upon PV performance as the R_{Sh} plays a much more significant role as the area increases. The open circuit Voltage (Voc) of the AgNWs electrode device is slightly lower as AgNWs possess a lower work-function and this accounts for the observed decrease of Voc [196]. In addition to the slight performance increase, the AgNWs do possess a number of additional advantages over ITO including the

improved mechanical robustness, no post treatment requirements and is not processed in vacuum.

Device performance has also been tested in resistive heaters and shown in Figure 4.8. The performances demonstrated for bias condition (5V, 300mA), showing that heat can be applied very quickly and over an extended temperature range (limited by the temperature compatibility of the substrate). This data shows that the post-processed electrodes can be used for a range of applications.



Figure 4.8: (a), (b) Images of transparent heaters, (b) Temperature profile of transparent heaters with different over-coatings.

4.7 Mixed AgNW TCEs

In this section individual AgNWs and mixed AgNWs type TCEs with a similar R_{Sh} of 25±1 Ω sq-1 results are presented. Individual types of AgNWs with different mean diameter D and length L used for the study, namely thin-short nanowire; AgNWs-60S (D = 60 nm, L = 20-30 μ m), thin-long nanowire; AgNWs-30L (D = 30 nm, L = 100-200 μ m) and thick-long

nanowires; AgNWs-100L (D = 100 nm, L = 100-200 μ m). The shorter nanowires are advantageous for achieving high conducting networks at low concentration and the longer nanowires provide higher conductivity performance. The combined advantages of short and long nanowires were exploited by coating them together. Two mixed AgNWs composition were prepared, one consisted of the type: thin-short nanowires (AgNWs-60S) with thin-long nanowires (AgNWs-30L), denoted as 'AgNWs-M1', and another one consist of the type: thin-short nanowires (AgNWs-60S) with thick-long nanowires (AgNWs-100L), denoted as AgNWs-M2. Mixed composite AgNWsAgNWs-M1 dispersion was prepared by mixing AgNW-60S and AgNWs-30L as supplied in 1:1 volume by volume (V/V) ratios and then diluted with ethanol to 0.4 mg/mL final concentration. Similarly, AgNWs-M2 was prepared with AgNWs-60S and AgNWs-100L. AgNW dispersions in ethanol of each type (0.5 mg/mL) were sonicated at low power for 5 minutes to minimized aggregates before air spray coating. Individual AgNWs and mixed AgNWs type TCEs were prepared using the same process discussed in chapter 3 section 3.3. The deposition time for preparing electrodes with $R_{Sh} \approx 25 \Omega \text{ sg}^{-1}$ was found to be approximately 5 minutes ±5 seconds. However, using mixed geometry electrodes, the deposition time was approximately 50% lower due to more inter connections made by short AgNWs to connect the gap between longer nanowires. This represents a significant advantage for future scale up of the technology. The mixed AgNWs results show moderate improvement in the electrical/optical performance and the surface roughness.

SEM images of the conducting surface of AgNW electrodes prepared from fixed dimension type AgNWsAgNWs-30L and AgNWs-60S is shown in Figure 4.9 (a) and Figure 4.9 (b) while a mixed composition type AgNWs-M2 is shown in Figure 4.9 (c). The inter mixing of short-thin nanowires AgNWs-60S with long-thick nanowires AgNWs-100L can be clearly seen from Figure 4.9 (c), which provides more event surface coverage and as a result of fewer aggregations at nanowire junctions.



Figure 4.9: SEM images of (a) AgNWs-30L (b) AgNWs-60, (c) AgNWs-M2.

4.7.1 Optical and Electrical Properties

The transmittance spectra of TCE prepared by the different types of AgNWs dispersion with a similar R_{Sh} of $25\pm1 \Omega$ sq⁻¹ are shown in Figure 4.10 (a). This value of R_{Sh} was chosen as it represents a sizeable improvement over ITO on a flexible substrate and is comparable to the value of ITO often used on glass substrates. It can be seen from Figure 4.10 (a) that longer AgNW electrodes (AgNWs-100L, AgNWs-30L) shows 5-10% transmittance improvement in the visible region of spectrum compare to the shorter nanowires AgNWs-60S. In addition, the thin-long nanowires (AgWS-30L) electrode was found to show higher transmittance than the thick-long nanowires (AgNWs-100L) electrode with similar R_{Sh} of 25Ω sq⁻¹.



Figure 4.10: (a)Transmittance spectra of AgNWs-30L, AgNWs-60S, AgNWs-100L, AgNWs-M1 and AgNWs-M2 composite electrodes with R_{sh} of 25 Ω sq⁻¹, (b)FoM and average FoM of AgNWs-30L, AgNWs-60S, AgNWs-100L, AgNWs-M1 and AgNWs-M2 composite electrodes with sheet resistance of 25 Ω sq⁻¹.

The mixed composition AgNWs-M1 electrode shows similar overall transmittance values to the longer nanowire component AgNWs-30L, but with moderate improvement in the region 400 nm to 550 nm of the visible spectrum. This could be a significant result for academics and industrialists working with blue OLEDs, who wish to maximize the light output in this region. Likewise, the AgNWs-M2 electrode displayed better transmittance compared to its parent AgNW electrodes of AgNWs-60S and AgNWs-100L and improved performance in the blue-green spectrum. Due to the reduced aggregation and better percolation of the mixed AgNWs films, the transparency is greater for the same sheet resistance, and the overall silver content of the film is lower. For the

AgNW TCEs, the silver content is calculated at 0.2 mg cm⁻² for mixed, which compares to 0.3 mg cm⁻² for long AgNWs and 0.4 mg cm⁻² for short AgNWs.

To quantify the optical and electrical properties a Figure of merit (FoM) is commonly defined in literature given in equation (2.4), discussed in Chapter 2. For comparison, commercial ITO film on PET used in our laboratory was measured and the FoM found to be 58 (R_{Sh} = 60 Ω sq⁻¹, transparency = 90%, source Sigma Aldrich, UK). The FoM values calculated for various AgNW electrodes of 25 Ω sq⁻¹ R_{Sh} with transmittance (%T) at 550 nm is plotted in Figure 4.10 (b). Among the different AgNWs type electrodes studied, the mixed composition AgNW electrodes AgNWs-M1 (FoM = 138.6, Table 4.6) has the highest FoM value while lowest for short-thin nanowires AgNWs-60S (FoM = 85.1, Table 4.6). The optical performance of the mixed-composition electrodes (AgNWs-M1, AgNWs-M2) in terms of FoM is found to be higher as compared to the parent electrodes (AgNWs-30L, AgNWs-60S, AgNWs-100L) having same R_{sh} of 25 Ω sq-1. In Table 4.6, the best and average performances of the three most promising combinations are shown i.e. AgNWs-M2, short AgNWs-60S, and long AgNWs-30L. When considering the FoM, the mixed AgNWs films show around 10-15% improvement over the best obtained results from an AgNWs film prepared from a single nanowire geometry (AgNWs-30L). In addition, the mixed AgNWs film shows 30-40% improvement over AgNWs films prepared with short nanowires. The results support the conclusion that short nanowires do not provide good percolating networks and therefore exhibit poorer performance than longer nanowires. However, they could still have industrial applications, should they be used in combination with longer nanowires.

Туре	R _{sh}	R _{sh}	R _{sh}	T (%)	T (%)	T (%)	FoM	FoM
	(Ω sq ⁻¹)	(Ω sq ⁻¹)	(Ω sq ⁻¹)	(Best)	(Average)	(s.d)	(Best)	(Average)
	(Best)	(Average)	(s.d)					
AgNWs- M1	25.7	25.9	0.2	90.2	89.8	0.5	138.6	131.6
AgNWs- M2	25.1	25.4	0.4	87.1	86.4	0.9	105.0	97.8
AgNWs- 60S	25.6	26.0	0.8	84.7	84.1	0.4	85.1	80.1
AgNWs- 30L	25.3	25.8	0.7	89.3	88.8	0.6	127.9	119.4
AgNWs- 100L	25.2	25.4	0.3	84.8	84.0	1.0	87.0	81.4

Table 4.6: Average (from 5), best and standard deviation (s.d) of sheet resistance and transmittance of (AgNWs-M2), short (AgNWs-60S) and long (AgNWs-30L) electrodes.

4.7.2 Surface Roughness

The surface roughness of the conductive surface of the AgNW electrodes was examined by white light interferometry (WLI) and AFM. Figure 4.11 shows the topological images of AgNWs-30L, AgNWs-60S, and AgNWs-M2 electrodes, respectively.





Table 4.7 shows the surface roughness of various types of nanowires electrodes, (AgNW-30L, AgNW-60S, AgNWs-100L, AgNWs-M1, AgNWs-M2) by WLI measurement over a large area of 278 x 207 μ m². In this work, the surface roughness was calculated from an average of five different samples and the average root mean square roughness (R_q) and maximum peak-to-valley (Pv) roughness was calculated.

For all samples, the highest average P_V roughness was obtained for the short nanowires ($P_V = 87.2$ nm), although the surface roughness is overall much lower than other reports. The low peak-to-valley roughness was attributed to the PMMA layer, which possibly facilitates the embedding of nanowires, thereby smoothening the surface.

Туре	Rq (nm)	Rq (nm)	Pv (nm)	Pv (nm)
	average	STDV	average	STDV
AgNWs-M1	3.2	0.2	75.2	2.1
AgNWs-M2	4.1	0.4	80.9	2.9
AgNWs-60S	8.5	1.2	87.2	3.9
AgNWs-30L	6.2	0.6	85.9	3.8
AgNWs-100L	5.9	0.3	80.1	2.0

Table 4.7: Surface roughness parameters of mixed (AgNWs-M2), short (AgNWs-60S) and long (AgNWs-30L) electrodes.

The root mean square roughness (R_q) for the mixed composition AgNW electrodes (AgNWs-M1, AgNWs-M2) are found in the range 3-4 nm while for the fixed dimension type nanowires electrodes (AgNWs-30L, AgNWs-60S, AgNW-100L) the values are in the range 6-8 nm. This indicates that the mixed composition AgNW electrodes possess a somewhat smoother surface compared to the fixed dimension type AgNW electrodes, attributed due to a more evenly distributed surface coverage. For illustration, topological images of a short, a long and mixed composition type nanowire electrode are shown in Figure 4.11. It can be seen by visual inspection of Figure 4.11 that AgNWs-M2 $(R_q = 3.7 \text{ nm}, PV = 79.6 \text{ nm})$, based electrodes topological image is smoother compared with AgNWs-30L (R_q = 5.5 nm, P_V = 77.3 nm) and AgNW-60S (R_q = 7.9 nm, P_V = 81.7 nm) electrodes. The high surface roughness of AgNW TCE is a result of multiple AgNWs overlapping one another or by aggregation. A low volume of material is used during this deposition, so the long AgNWs are dispersed quite evenly across the sample and the amount of overlap is relatively low. Long AgNWs tend to form more aggregates during the spray coating deposition, so by minimizing the volume deposited, the concentration of aggregates is reduced, thus lowering the overall surface roughness. Subsequently, the short AgNWs are deposited; these do not tend to form large aggregates. Therefore, they act as an 'in-fill' into the long AgNW mesh, providing the percolation and ensuring no major aggregation points are formed.

The advantage of mixed composition silver nanowires, comprising of short nanowires length (L = 25 μ m) with long nanowire (L = 150 μ m), include shorter spray deposition time, better electrical/optical properties (as evidenced by the FoM values) and lower surface roughness electrodes. We believe that the simple technique demonstrated in this work will be effective in saving time and cost for large-scale production of AgNW based transparent conducting electrodes and helpful for further development of metallic nanowire-based electrodes in general.

4.8 Effect of Post Processing on Stability of AgNW Films

To study the stability of AgNW films, AgNW TCEs were subjected to various high temperature and current conditions to access the lifetime, failure at 10 %, 25% of the initial value of electrode resistance (shown in Table 4.8). One of the examples to test the lifetime of AgNWs TCEs is shown in Figure 4.12. Embossed and non-embossed (normal or non-post processed) AgNWs TCEs were subjected to 300 mA at 65 °C temperature to observe the effect of embossing on high current and 65 °C stability. It can be clearly observed from Figure 4.12 (a) and (b) that embossed AgNW TCEs are much stable as compared to normal due to negligible change or degradation for embossed samples due to less heat dissipation.

	Post process	Stress condition	Failure at 10 % of initial value	Time for failure at 10 % of initial	Failure at 25 % of initial value	Time for failure at 25 % of initial
			(Ω)	value (hours)	(Ω)	value (hours)
	No	000 4 . 55	26.51	264.00	30.13	401.25
_	INO	300 mA at RT	20.13	362.25	22.88	645.00
	Embagad		16.94	960.00	19.25	2358.00
	Embossed		14.96	1788.00	17.00	4308.00
	No	300 mA at 65	27.83	5.46	31.63	13.69
	INO	°C	38.50	4.50	43.75	11.29
	Embagad		17.38	62.25	19.75	165.75
	Empossed		15.84	60.00	18.00	153.00

Table 4.8: Failure (300 mA, RT, 65 °C) at 10 %, 25% of initial value of electrode resistance.



Figure 4.12: Effect of embossing on high current 300 mA at 65 °C temperature stability (a) Change in resistance for normal and embossed AgNW TCEs up to 166.5 hours (b) Average change in resistance for normal and embossed AgNW TCEs up to 166.5 hours.

4.9 QALT Testing of the Reliability of AgNW Films

To study stability, quantitative accelerated life testing (QALT) can be applied to relate failure rates achieved at elevated stress levels to what might be under normal operational conditions. For this work, four experimental tests were conducted using a (2²) full factorial experimental design with the following two-stress, two-level conditions; 45 ^oC/150mA/cm², 45^oC/250mA/cm², 65^oC/150mA/cm² and 65^oC/250mA/cm² applied to accelerate the degradation. When the temperature and current density are the accelerated stresses of a test, then a modified Eyring model can be developed and used as the life test model to relate degradation at an elevated stress level to that at a lower, operational level. This is shown in equation 4.1 and is referred to as the Temperaturecurrent (TI) relationship. Equation 4.1 can be used for predicting life at lower operational conditions, where life is shown as a function of temperature (V) and current density (I) and all other components are fitting parameters and calculation of the acceleration factor (AF) (equation 4.2). Six samples were used each test condition and a Weibull 2-point distribution was applied for each stress level (defined in equation 4.3, where β is defined as the shape parameter, η is the scale parameter, t is the time and f(t) is the probability of failure. Shown in Table 4.9 are the fitting parameters of the TI model, which have been calculated for the time to increase in resistance by 10% (T10%) and 25% (T25%) of the original value. To compare the impact of post-processing, a consistent definition of failures is needed. For this work, we calculated the time for 63% of the population of AgNW electrodes tested to have declined a particular value [such as 10% increase in resistance (T10%) or 25% increase (T25%)]. This value is often referred to as B(63%). By considering equation 4.2, when $t = \eta$, the cumulative number of failures in the population, F(t) = 63%, so η is equivalent to B (63%).

$$L(I, V) = \frac{A}{I^n e^{-\frac{B}{V}}}$$
 Eq. 4.1

$$A_F = \frac{L_{USE}}{L_{Accelerated}} = \frac{I_A}{I_O}^n e^{D\left(\frac{1}{V_O} - \frac{1}{V_A}\right)}$$
Eq. 4.2

$$f(t) = \frac{\beta}{\eta} \left(\frac{t}{\eta}\right)^{\beta-1} e^{-\left(\frac{t}{\eta}\right)^{\beta}}$$
 Eq. 4.3

Based upon equation 4.1, life is a function of two stress conditions, so life versus temperature and life versus current density can be plotted on a logarithmic scale to show how varying temperature or current density affects the life of the AgNW films, whilst the other variable is kept constant. This is shown in Figure 4.13 (a) and (b) for life model fittings from Table 4. The general trend is to be expected; as the stress level of current density (a) or temperature (b) is increased, the anticipated life of the AgNW film decreases. It is evident from Figure 4.13 that the stability of the post processed samples is substantially higher than the non-post processed samples, by around three orders of magnitude under normal operational conditions (I_{bias}=20mA/cm² and temperature). This is because the post processed samples possess a lower resistance due to improved electrical contact. As the power dissipated across the electrode is related to bias current and resistance by $P_D = I^2$. R, then the 'as used' sample will experience 16x more power dissipation that the post-processed sample, leading to increased Joule heating, which causes the nanowires to break up and thus creates an electrical discontinuity in the nanowire film [197]. As more heat is created, the probability of failures increases. Furthermore, during the post-processing procedure, the AgNWs are partially embedded within the underlying polymer, which is likely to act as a barrier layer limiting the impact of oxidation and sulfidation.

Table 4.9: Fitted parameters (FP)obtained using mean likelihood estimation (MLE) for post-processed samples and as-used samples. β is defined as the shape parameter, η the scale parameter from the Weibull 2-point probability distribution in equation 4.2. The values of A, B, and n are also fitting parameters from the temperature-current model. in equation 4.1.

FP	Post-processed T10%	As used T10%	Post-processed T25%	As used T25%
А	7.79e-7	2.0e-6	9.51e-8	3.20e-7
В	9359	7824	10220	8440
n	1.74	1.92	1.68	1.82
β	2.47	2.08	2.42	2.11
η	63353	433	175002	1005

An important characteristic often assessed in life test models is the acceleration factor (AF), which shows the ratio of the AgNWs life at the 'operational' stress level to its life at an accelerated stress level and is defined in equation 4.1. Both the post processed and 'as used' samples show similar characteristics, but the 'as used' samples appear to degrade quicker with increases in current density.



Figure 4.13: Regression line fitting for post-processed and 'as-used' AgNW films of the expected time to reach T90% as a function of (a) current density (mA/cm²) and (b) temperature (K). Included in (a) and (b) are the fitted probability distribution functions for each stress level. The Acceleration Factor (AF) as a function of current density and temperature and RH humidity are shown in (c) and (d), respectively.

To evaluate the stability fully, further experiments have been conducted by applying different interlayers onto the AgNW electrodes to see the interaction between two layers. A single stress condition was applied (I_{bias} = 300mA/cm² and temperature = 65 °C), enabling a relative comparison of stability to be obtained, which is reported in Figure 4.14 The least stable electrode was the 'as used' electrode which had no post processing conducted. The poor stability is due to thermal oxidation of AgNWs as the sample is exposed to the environment. When AgNW films are exposed to air and water, they can be easily oxidized, leading to a sharp increase in sheet resistance and haze of the AgNWs films[198]. After the post-processing, the AgNWs are moderately embedded into PMMA and this appears to act as a barrier to inhibit the level of oxidation. By applying a further overcoating the stability is enhanced, although the most significant improvement is obtained by using the ZnO interlayer. The application of PEDOT+DMSO shows the

lowest stability as an overcoating layer and is likely to be related to the reactive nature of Ag to environmental sulphur which causes sulfidation.



Figure 4.14: Bar chart showing mean time to reach T10% (black) and T25% (red) for AgNW films including non-post processed ('as used'), post processed, and post processed with ZnO, PEDOT:PSS+DMSO and MoO₃ transport layers. All samples were tested under the same conditions with a current density of 300mA/cm² and a temperature of 60 °C applied.

It is worth noting that the optical properties did not change upon ageing. For example, the optical haze of the aged samples was compared to freshly prepared samples. For samples without any interlayers (i.e. ZnO/PEDOT:PSS), the optical haze of fresh samples was measured at 4.81%, which changed less than ±0.10% after ageing. This value is an average of three samples.

4.10 Adhesion and Mechanical Testing

The AgNWs adhesion was determined through a pull-off test using TQC SP302 standard tape (width 1cm). The procedure followed the ISO 2409 standard which is used for the determination of adhesion of polymeric layers such as paints and varnish [199]. Considering the AgNW films, no cracks or delamination was observed by visual inspection after the pull-off test which confirmed robust adhesion and compatibility between the PMMA and AgNWs. Measurements of the sheet resistance before and after showed no discernible difference for the post-processed samples (<5% relative difference). By contrast, for non-processed samples, the sheet resistance increased by 65%, showing that the post-processing increases the adhesion of the conductive coating also.



Figure 4.15: Effect of bending on sheet resistance of post processed AgNW and ITO TCE.

The bend testing was performed to demonstrate the enhanced mechanical performance of the post-processed AgNW based electrodes during repeated bending. Under a 100-cycle test with a bending radius of 20mm, almost no effect was observed while dramatic failure appeared in the ITO electrode in the same test. The test shows that the mechanical flexibility of the samples is not affected by the post-processing of the AgNW films as shown in Figure 4.15, where R and R₀ represent the sheet resistance of films before and after the bending test, respectively.

4.11 XPS Analysis of the AgNW Films

XPS analysis was conducted on the AgNW films to investigate the impact of embossing, N₂ plasma treatment and accelerated ageing on the surface chemistry. All XPS measurements were performed at Chester University with the help of Dr. J. Kettle and Prof. G. C. Smith. Surface compositions as determined by XPS are shown in Table 4.10, after correction for the presence of small amounts of silicon found as a surface contaminant. All samples showed high levels of C and O, in most cases relatively strong Ag signals also present. All samples also showed low levels of S, Cl and N. The data in Table 4.10 shows the chemical stoichiometry of the different samples. For a deeper analysis, high-resolution scans were made of the individual photoelectron lines and curve fitted to a known reference data for silver and polymers.

The C1s spectra were typical of those obtained from a poly (methyl methacrylate) PMMA surface, with the hydrocarbon component arising from the backbone and the pendant methyl groups, and the oxygenated components arising from the two carbonoxygen bonded components of the ester group. The variation in the presence of carbon in this form indicates that there are different levels AgNW coverage between different samples. Samples that have been embossed show higher levels of carbon surface, indicating they have been submerged into the underlying PMMA.

Element	As	Thermal	N ₂ plasma	Thermal	As	Thermal embossing
	prepared	embossing	ueaument	ennossing	prepared	
				and N ₂ plasma	and aged	treatment and aged
				treatment		
C 1s	66.66	72.92	61.73	62.31	67.42	70.95
0 1s	19.72	22.15	27.10	25.12	22.20	24.91
Ag 3d	7.99	1.58	7.11	8.19	5.09	1.01
S 2p	1.66	0.29	1.86	2.07	2.13	0.14
Cl 2p	1.00	0.17	0.96	0.91	0.83	0.22
N 1s	2.97	2.89	1.24	1.40	1.79	2.32
Na 1s	-	-	-	-	0.55	0.45

Table 4.10: Surface composition obtained by XPS, showing the impact of N_2 plasma treatment, embossing and ageing upon the chemical changes at the surface.

The levels of Ag did not change significantly upon N₂ plasma treatment; however, the embossed samples showed much lower levels of Ag (consistent with the SEM images and Carbon XPS data). Nevertheless, even after thermal embossing, Ag was clearly present at the surface, indicating that the Ag was not fully embedded into the substrate during this processing step. By comparison, the N₂ plasma treated sample showed a decrease in the surface concentration of C and an increase in the surface concentration of O compared to the control. We speculate that the N₂ plasma removes hydrocarbon impurities from the AgNW films, namely the PVP used in end-capping during the synthesis, and this leads to the reduction in C and the improved electrical properties. This is supported by the reduction in the surface concentration of N after N₂ plasma treatment, which is also likely to be due to the removal of PVP. The increase in O content after thermal embossing is likely to be because of the raised temperature in ambient air leading to increased oxidisation.

On all samples, the Ag $3d_{5/2} - 3d_{3/2}$ doublets were sharp, intense, symmetrical in shape and typically 1.2 - 1.3 eV in width. In all cases, the binding energy of the $3d_{5/2}$ component was found in the range 368.8 - 369.3 eV, somewhat higher than the internationally-accepted reference value of 368.27 eV for clean metallic silver [200]. In all cases, a doublet separation of 6.02 ± 0.01 eV was found. Ag is unusual among metallic elements in that its oxides tend to give photoelectron peak binding energy shifts to lower binding energy than the metallic state. Here, a shift to higher binding energy is seen. This

is occasionally seen in the presence of alloying elements, but that cannot be the case here. Instead, it is more likely that the shift is due to charge transfer between the nanowires and the (insulating) PMMA/PEN substrates. This would be consistent with the results reported by Lin & Wang [201]and from other studies of deposited Ag layers. It could alternatively be a function of a more tightly bound crystal structure of the Ag in the form of nanowires compared to bulk metal.

Ageing the as-prepared sample cause small increases in the surface concentrations of O, S and C. Ageing of the post-processed sample caused an increase in the surface concentration of C and small reductions in the surface concentrations of the other elements present, except for Ag where a significant reduction from ~8% to ~ 1%, suggesting coverage of the Ag by a carbon-containing material as a consequence of the ageing process.

Sulphur was detected at low levels on all samples, with a possible correlation with the level of Ag (correlation coefficient $r^2 = 0.79$). This is consistent with the reactive nature of Ag to environmental S and the consequent formation of silver sulphides. Inspection of the S 2p photoelectron peaks showed the majority of S present to be in a sulphide state, S²⁻, consistent with AgS. The thermal embossed and aged samples show low levels of S indicating why these samples remain stable. By ensuring a greater proportion of the AgNWs are submerged into the underlying PMMA, they remain less susceptible to environmental ageing.

4.12 Summary

This chapter reports the high performing AgNW electrodes. High performance achieved in terms of the low sheet resistance and surface roughness as well as high stability. The process relies upon three sequential steps; thermal embossing, infrared sintering and a nitrogen plasma treatment. This process leads to the demonstration of a conductive film with a sheet resistance of $2.5\Omega \Omega$ sq⁻¹and 85% transmittance, thus demonstrating the highest reported Figure-of-merit in AgNW TCEs to date (FoM = 933). A further benefit of the processes surface roughness is substantially reduced compared to traditional AgNWs techniques. Accelerated life testing has been applied to show that the process leads to enhanced long-term stability in AgNW films. X-ray photoelectron spectroscopy (XPS) is used to understand the improvement in long-term stability. The overall properties of the transparent electrodes developed show enhanced properties over the incumbent technology; ITO.

Chapter 5. Development of ZnO TFTs Using Design of Experiments (DOE) Techniques

In this chapter, the development of low voltage metal oxide thin film transistors (TFTs) based upon Zinc Oxide (ZnO) is reported. Firstly, a comprehensive study of the influence of the processing parameters on the performance of anodized Al₂O₃ insulating films for ZnO TFTs built is conducted. In particular, a combinational science approach is undertaken in order to provide a rapid methodology to optimize the layer deposition. This approach could provide a step-change for process optimization for researchers in the area of flexible electronics. To accomplish this, 'screening' of the anodization parameters was carried out by using a Plackett–Burman Design (PBD), which enables several parameters to be varied simultaneously, with the reduced number of experiments, enabling identification of the most significant factors which impact the TFT performance. We studied eight different process parameters that influence the performance parameters of ZnO TFTs (mobility, threshold voltage) by using analysis of variance (ANOVA). This procedure could be adopted more widely in flexible electronics to increase the speed of product development.

Taguchi orthogonal array (OA) has then been applied for optimisation of ZnO TFTs performance under different input conditions. In addition to the combinational science approach, a detailed comparative study of sputtered and sprayed ZnO film properties and their impact upon TFT performance has been studied. Further for UV sensing applications UV saturation experiment on ZnO TFTs for both methods of deposition are explained using TFT behaviour on saturation after UV exposure. State of the art performances for sprayed ZnO TFTs mobility of 15.5 cm²V⁻¹s⁻¹ and high on/off ratio of 2.1x10⁶ is achieved and. However, for sputtered TFTs mobility of 1.19 cm²V⁻¹s⁻¹ and on/off ratio of 7.67×10^3 is achieved. It has been investigated how annealing temperature affects the structural and optical properties of TFTs. The band gap values are evaluated and approximately agrees with that of bulk ZnO at room temperature. However, bandgap (Eg) decreases as the annealing temperature increases due to the increase in grain size. AFM images of the samples show that there is an increase in roughness as annealing temperature increases, indicating an increase in grain size. X-ray photoelectron spectroscopy (XPS) is used to understand the root cause of improved electrical performance.

5.1 Development of High-k Dielectric (Al₂O₃)

Most transistors and ZnO-based circuits operate at relatively high voltage, exceeding 20V. This is a disadvantage for practical low power applications such as for battery powered portable devices that operate at voltages below 5V. To achieve lower operational voltage, optimization of the TFT dielectric layer is necessary. This is not without challenges as the control over the defect density in the dielectric is key to ensure reproducible performances in ZnO-based TFTs. These defects depend on the deposition technique of ZnO but are also highly dependent on the properties of the gate insulator and the quality of the gate insulator/semiconductor interface. A possible way to overcome both challenges is the use of thin layers (<20 nm) of dielectric, with high dielectric constant (> 6). The dielectric must have simultaneously (i) high capacitance per unit area, (ii) low leakage currents to minimize power loss, (iii) low surface roughness to minimize dispersion and thus improve carrier mobility and (iv) a low density of the interfacial trap states to minimize instability of the threshold voltage. Aluminium oxide (Al₂O₃) is a highk dielectric that can be used as the gate material for TFTs. An approach used to deposit high-quality Al₂O₃ films is anodization. Importantly, anodization is a self-limiting process as thin film growth occurs in different directions during the process, resulting in pinholefree, homogenous oxides. In this process, the aluminium layer is normally thermally deposited on substrates which are then immersed in an electrolytic aqueous solution, for example, tartaric acid, and then a constant current is applied at the Al electrode until a pre-determined voltage is achieved, which defines the film thickness. The thickness (t in nm) of the Al₂O₃ films is determined by the anodization voltage (V_A) during the anodisation process by the relation, $t = c \times V_A$, c being anodization constant of Al and can vary depending on the experimental conditions for the oxide growth [202]. The thickness of dielectric layer was estimated from above relation because Al₂O₃ is difficult to measure Al₂O₃ underneath part towards Al layer. The chemical equations involving anodisation of aluminium are given below:

$$4Al + 3O_2 \rightarrow 2Al_2O_3 \qquad \qquad \text{Eq. 5.1}$$

$$2Al + 3H_2O \rightarrow Al_2O_3 + 3H_2 \qquad \qquad \text{Eq. 5.2}$$

where, the formation of Al₂O₃ is directly proportional to the amount of water or oxygen molecules in the electrolyte. There is evidence that some excess oxygen is retained in the bulk of Al₂O₃, which can promote the tunneling effect which leads to higher leakage current.

For high performing TFTs, the dielectric must possess (i) a large capacitance per unit area, (ii) a low leakage current and (iii) a low surface roughness to minimize carrier scattering, allowing optimal carrier mobility in the transistor channel during accumulation; and (iv) a low interfacial trap density to minimize threshold voltage instabilities.



Figure 5.1: (a) Dielectric constant of Al_2O_3 and AC conductivity (log scale) versus frequency, (b) C-f curve of the anodic capacitor, (c) C-V curve at different frequencies of the anodic capacitor, (d) AFM image of Al_2O_3 gate dielectric layer.

Shown in Figure 5.1(a) are the dielectric constant (K=9.8) of anodic Al_2O_3 was calculated from the measured values of capacitance versus frequency in metal insulator metal (MIM) capacitors. The dielectric constant does not vary with the frequency from 20 Hz to 10 kHz, but the ac conductivity increases linearly with an increase in frequency as

shown in Figure 5.1(a). The ac conductivity is related to the energy loss obtained from the polarization mechanism. The increase in ac conductivity can be understood by the equation 5.3.

$$\sigma_{ac}(\omega) = \sigma_{total}(\omega) - \sigma_{dc} = A\omega^s$$
 Eq. 5.3

where, ω is the angular frequency, s is the frequency exponent and A is a constant independent of frequency.

Figure 5.1 (b) shows that capacitance is constant over wide range of frequency (20 to 10 kHz). RMS and average roughness of Al₂O₃ gate dielectric layer is 0.75 nm, 0.57 nm respectively shown in AFM image in Figure 5.1 (d).

5.2 Optimisation of Anodic Al₂O₃ using Plackett-Burman Design

The optimisation of anodic Al₂O₃ dielectric was performed because in most of the previously reported works no detailed study was carried out for all input factors of anodisation process. In this work, a two-level multivariate analysis based on a Plackett-Burman design has been applied to optimise the fabrication of anodised (Al₂O₃) dielectric films for ZnO TFTs. The electrical performance parameters of the devices were measured and Analysis of Variance (ANOVA) of the combined responses has been applied to identify how the Al₂O₃ dielectric fabrication process influences the electrical properties of the TFTs. Using this approach, the levels for the manufacturing factors to achieve optimal overall device performance parameters have been ranked. The crosschecked analysis of the TFT performance parameters demonstrated that the appropriate control of the anodisation process can have a higher impact on TFT performance than the use of traditional methods of surface treatment of the dielectric layer.

In this work, eight different factors with two levels considered for the anodisation process are listed in Table 5.1. Aluminium films were evaporated onto glass substrates and anodisation process was carried out according to factors and levels are given in Table 5.1. After anodisation, the films were removed from the solution and the impact of post annealing considered; samples were either left to dry at room temperature or annealed at 150 °C for 1 hour. The performance of the Al₂O₃ films as a dielectric layer was tested by manufacturing metal-insulator-metal (MIM) capacitor structures and bottom-gate/top-

contact ZnO TFTs. For TFTs, ZnO films were deposited onto the Al₂O₃ film by RF sputtering method explained in Chapter 3. The thickness (40nm), as well as the evaporation rate (0.5 Å/s) of the ZnO films, were controlled by a quartz crystal sensor. To complete the transistors and the MIM capacitors, a top layer of aluminium (70 nm) was thermally evaporated through a shadow mask to form a circular electrode, which has a central electrode with 1 mm of diameter surrounded by a concentric ring, with a 400-µm gap. This external ring was grounded, working as a guard ring for the MIM capacitors) were evaluated for each test run. Fourteen parameters were used for the analysis, which were separated in three different groups: i) parameters for evaluation of the Al₂O₃ oxide film, obtained from impedance/capacitance spectroscopy measurements in MIM capacitors and AFM images; ii) parameters obtained from the TFT characterisation curves, associated with the carrier transport in the semiconducting film or at the semiconductor/dielectric interface; and iii) parameters obtained from the source-to-gate current curves, focusing on the quality of the dielectric layer during a TFT measurement.

Table 5.1: Factors chosen for use in the screening test of the two-level factorial factors l	οy
the PBD and correspondent values.	

Factors	Factors name	Unit	Low value	High value
А	Thickness of Al-layer	nm	60	200
В	Al evaporation rate	Å/s	5	15
С	H ₂ O content	%	16	30
D	Electrolyte temperature	⁰ C	40	60
Е	pH of the electrolyte	-	6	5
F	Current density	mA/cm ²	0.45	0.65
G	Annealing	С	No	150 °C
Н	Final voltage	V	30	40

The first group includes the Al₂O₃ dielectric constant (ϵ), the ac conductivity at low frequencies (σ_{LF}) and the AC conductivity at high frequencies (σ_{HF}). As an additional response parameter of the dielectric film quality, the root mean square (RMS) roughness of the anodised surface, determined from atomic force microscopy (AFM) images was considered. The second group comprises the ZnO TFT electron mobility (μ), the threshold voltage (V_T), the hysteresis of the transfer curve (ΔV_H), the "off" state current (I_{off}), the "on" state current (I_{on}) and the on/off ratio (I_{on}/I_{off}). The third group used as response parameters the gate current in the depletion state (I_{dep}), the gate current in the accumulation state (I_{acc}), the voltage of minimal gate current in forward scan (V_{for}) and the voltage of minimal gate current in reverse scan (V_{rev}). In this thesis, focus on the response parameters from the second group, which are more important to the TFT performance, was kept.

The design of experiments used for screening the anodisation factors introduced in Table 5.1 is based on a matrix sequence of the twelve runs according a Plackett-Burman design (PBD) presented in Table 5.2 (discussed in Chapter 2). In Table 5.2, the letter L and H refers to the "low value" and "high value" for each process factor, respectively, whereas the letters A to H refers to the anodisation factors. For each experimental run, a set of measurements (TFT curves, AC impedance and AFM) was carried out for an average of eight samples used for each run. It is worth to mention that the combination of experimental runs and their respective low and high levels of the factors specified in Table 5.2 were determined in a way that is possible to apply Analysis of Variance (ANOVA) for screening the most significant ones. Moreover, another characteristic of the PBD matrix in Table 5.2 is its orthogonality, i.e., for each level (high or low) in one factor, there is an even number of "highs" and "lows" for all the remaining factors of the matrix. ANOVA has been used to identify which processing condition of the dielectric layer contributes more significantly to a determined best TFT response parameter (e.g. μ , V_T, I_{on}, I_{off}, Δ V_H, I_{dep}).

Run	Α	В	С	D	Ε	F	G	Н
1	L	L	L	L	L	L	L	L
2	Н	L	Н	Н	L	Н	Н	Н
3	L	L	L	Н	L	Н	Н	Н
4	Н	L	L	L	Н	L	Н	Н
5	Н	Н	L	Н	Н	Н	L	L
6	L	Н	L	Н	Н	Н	L	L
7	Н	Н	Н	L	L	L	Н	L
8	L	L	Н	L	Н	Н	Н	Н
9	Н	L	Н	Н	Н	L	L	L
10	Н	Н	L	L	L	Н	L	Н
11	L	Н	Н	L	Н	Н	Н	L
12	L	Н	Н	Н	L	L	L	Н

5.3 ZnO TFT using anodic Al₂O₃ as the gate dielectric

To analyse the results TFT performance parameters were extracted from each test run, which were used as output responses for the Plackett Burman design. Figure 5.2 (a) and (b) show, respectively, the output and transfer curves for a TFT manufactured using Test Run 3 from Table 5.2. Figure 5.2 (b) inset shows a cross-section schematic of the TFT structure used for this work. The $\sqrt{I_{DS}}$ vs. V_G curve is also depicted in Figure 5.2 (b), allowing the evaluation of the TFT mobility (μ) from the slope of the curve and the threshold voltage (V_T) from the extrapolation of the linear region to the horizontal axis.

The drain current and saturation mobility was calculated using equations 2.15 and 2.17 as described in Chapter 2 respectively.



Figure 5.2: a) Output curves sweeping V_G from 0 to 12 V, in of 2 V steps; b) transfer curve and $I_D^{1/2}$ vs. V_G for V_{DS} = +10 V. Inset: schematic of the ZnO TFT structure; c) transfer curve, for V_{DS}=10 V, showing the hysteresis between forward and reverse voltage scan; All curves were obtained from a ZnO TFT with anodised Al₂O₃ prepared as specified in run 3, Table 5.2.

In Figure 5.2 (c), the transfer curve shows clearly the hysteresis for a device constructed in accordance with Run 3 (where $\Delta V_{\rm H} = 1.5$ V).

From the transfer curves for run 3, a mobility of $1.27 \text{ cm}^2 \text{ v}^{-1} \text{ s}^{-1}$, an on-off ratio of 1.4×10^4 and the threshold voltage of 3.5 V were obtained. The performance parameters, namely, "on" current, "off" current and on/off ratio were defined as the maximum channel current in accumulation, the lowest channel current in depletion, and their ratio, respectively. The TFT transfer curves normally show hysteresis of the channel current (I_D) when the voltage at the gate electrode (V_G) was swept in forward or in reverse

direction. The evaluation of the hysteresis was undertaken using the method reported elsewhere [203], in which the hysteresis magnitude (ΔV_H) is defined as the voltage difference in the transfer curves, between forward and reverse scan, when the channel current is half of its maximum value.

5.4 Influence of the Dielectric Layer Fabrication Factors on the ZnO TFTs Characteristics

To study how variations in the dielectric layer manufacturing process affect the TFT characteristics, we analysed the characterization parameters (μ , V_T, Δ V_H, I_{on}, I_{off} and I_{on}/I_{off}) from TFTs which were built using the same ZnO layer process conditions and different dielectric layers fabricated according to the PBD defined by Tables 5.1 and 5.2. The mean values ($\overline{Y_n}$) and the standard deviation \overline{s}_n values of the TFT response parameters (each run has an average of 8 devices) are presented in Table 5.3. The mean value from all experiments (between-runs) for a particular response is denoted by $\overline{Y_T}$ whereas s_T is the correspondent between-runs standard deviation.

Table 5.3: Mean and standard deviation values of response parameters from the performance of ZnO TFTs using anodized Al_2O_3 for the 12 experimental runs using PBD method.

Run	μ (cm².V ⁻¹ .s ⁻¹)		V _{тн} (V)		Ι _{οΝ} (μΑ)		I _{OFF} (nA)		I _{ON} /I _{OFF} (x10 ³)		ΔV _H (V)	
	$\overline{Y_n}$	Sn	\overline{Y}_n	Sn	\overline{Y}_n	Sn	$\overline{Y_n}$	Sn	$\overline{Y_n}$	Sn	\overline{Y}_n	Sn
1	1.30	0.11	4.09	0.65	4.31	1.13	5.58	0.77	0.76	0.14	3.54	0.58
2	0.15	0.02	0.31	1.00	1.97	0.77	0.48	0.21	4.73	2.96	3.50	0.56
3	1.59	0.17	3.99	0.64	3.44	0.64	1.82	0.61	2.09	0.75	2.05	0.33
4	0.11	0.06	3.90	0.96	2.90	0.74	0.27	0.12	12.83	6.39	1.93	0.41
5	0.33	0.06	2.90	0.67	3.52	0.63	0.66	0.25	6.53	3.99	2.17	0.41
6	0.21	0.06	3.26	0.88	3.08	0.46	0.48	0.19	7.51	3.69	2.58	0.38
7	0.09	0.05	2.63	0.82	0.49	0.30	0.42	0.10	1.15	0.56	4.10	0.35
8	0.19	0.02	2.66	0.58	0.57	0.14	0.26	0.10	2.63	1.41	4.44	0.72
9	0.39	0.04	2.73	0.58	2.07	0.35	0.36	0.13	6.44	2.49	2.34	0.37
10	0.06	0.03	3.06	1.02	0.34	0.11	0.26	0.12	1.80	1.51	4.74	0.54
11	0.15	0.02	4.53	1.00	0.48	0.15	0.41	0.14	1.27	0.50	1.31	0.29
12	0.14	0.02	3.46	1.23	0.42	0.11	0.58	0.24	0.83	0.34	6.05	0.55
$\overline{\boldsymbol{Y}_T}$	0.39		3.13		1.97		0.97		4.05		3.23	
ST	0.50		1.09		1.46		1.51		3.69		1.41	
$\overline{\boldsymbol{s}}_n$		0.06		0.84		0.40		0.17		1.46		0.41

From Table 5.3, it is possible to qualitatively estimate the influence of the variation on the factors from the comparison of the between-run standard deviation (sT) to the

mean of the in-run standard deviation (\bar{s}_n). The value of the between-run standard deviation compared to the total mean of the response also suggests that the manufacturing factors influence the observed response parameters. For instance, focusing on the TFT mobility, the between-runs SD (0.50 cm².V⁻¹.s⁻¹) is even higher than the total mean value (0.39 cm².V⁻¹.s⁻¹), and several times higher than the mean in-run SD ($\bar{s}_n = 0.06 \text{ cm}^2.V^{-1}.s^{-1}$), indicating that factor variation has strong influence on the TFT mobility.

Table 5.4: Ranking of the factor effects considering the TFT mobility response parameter. The corresponding computed sum of squares of the effects, F-value, p-value, significance, percentage contribution, and cumulative contribution of each factor on the effects are also tabulated.

Response: TFT mobility									
Factor	Effect	Sum of squares	F-value	p-value	significant	Contribution (%)	Cumulative contribution (%)		
Н	-0.50	0.250	1218.8	< 0.001	yes	26.93	26.93		
В	-0.46	0.210	1024.3	< 0.001	yes	22.63	49.56		
С	-0.41	0.172	834.8	< 0.001	yes	18.45	68.01		
Α	-0.41	0.167	813.1	< 0.001	yes	17.97	85.98		
E	-0.32	0.104	507.5	< 0.001	yes	11.21	97.19		
D	0.16	0.024	118.2	< 0.001	yes	2.61	99.81		
F	0.04	0.002	7.5	0.0077	no	0.16	99.97		
G	-0.02	0.000	1.4	0.2482	no	0.03	100.00		

To understand if the manufacturing factors have a statistically significant impact upon the quality of the dielectric layer, ANOVA calculations were performed considering the response parameters obtained from all samples in all experimental run. The calculated effects from all sources of variation (factors) on the TFT mobility, the sum of squares and the percentage of contribution are presented in Table 5.4, showing the ranking of the factors considering the effect on the TFT mobility. Six factors (H, B, C, A, E, and D) are significant within the confidence interval. However, as a convention, we decided to consider as relevant only factors whose contribution to the effect is superior to the fraction correspondent to a uniform distribution among all the 8 factors (i.e., superior to 12.5%). This criterion was used to discriminate the factors which contribute more to the effect of factors, which, even though are significant, contribute less to the observed effect. Factors F and G were found to be non-significant, which means that the calculated effect of these parameters on the mobility could not be distinguished from noise. The factors which contribute most to the TFT mobility are (in rank order): the final voltage for the anodization process, the Al-evaporation rate, the electrolyte water content, and the thickness of the Al layer. All these factors presented a negative effect on the mobility, which means that when the factor is set on the "high" value, a decrease in the TFT mobility was observed.



Figure 5.3: Half Normal Plot of effects using the TFT mobility as the response parameter.

The corresponding half-normal plot for the absolute effects is also shown in Figure 5.3. The half-normal plot highlights that the factors H, B, C, A, E and D does not follow the normal distribution, considering a confidence interval of 99% (*p*-limit of 0.01). For the considered number of degrees of freedom and confidence interval, the t-value (defined as the square root of the F-ratio from equation 1) limit is 2.642. The vertical axis of Figure 5.3 is in "probability scale", which represents the inverse of a cumulative Gaussian distribution, that is, Y'= norminv(Y/100). The function norminv(x_p) computes the deviate x_p associated with the given lower tail probability p(0.01, in the present case) of the standardized normal distribution. x_p is calculated for the given p such that.

$$p = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{x_p} e^{-u^2/2} du \quad \text{for} - \infty < x_p < \infty$$
 5.4

Therefore, the cumulative Gaussian distribution is represented by the straight line of Figure 5.3. Factors distancing the cumulative Gaussian distribution contribute to the rejection of the null hypothesis (H0).

To discriminate the effects by varying the different manufacturing factors, we performed ANOVA on all the results, leading to the Pareto charts of effects presented in Figure 5.4. In these charts, the most significant factors are ranked and the threshold limit of significance (p = 0.01) is represented by a vertical dashed line. A factor in which absolute effect surpasses the p-limit is considered significant.

The effect of a factor upon a response parameter is defined as positive (negative) if the response presents an absolute increase (decrease) by changing the factor level from "low" to "high". Colours are used to distinguish if the effect is negative or positive in the Pareto charts of Figure 5.4. The Pareto charts show that several factors influence each response parameter considering a confidence interval of 99%. For the "off" current, all factors were found to be significant, whereas 7 factors were significant for both the mobility and the on-off ratio, 6 factors for the hysteresis, 5 factors for the "on" current, and 4 factors for the threshold voltage. Moreover, the factors ranking and effect sign (positive or negative) change considerably among the observed responses. For improved factor screening, we considered only the factors, which have a higher relative contribution to each response parameter, as described before during the analysis of Figure 5.3 and Table 5.4.

The anodization current density (factor F) is not among the most relevant factors in any of the cases, and the annealing treatment (factor G) and the electrolyte temperature (factor D) appear just once each among all cases, having smaller contributions to the observed effects. Table 5.5 also presents the sign of the influence of the factor (positive or negative) on the effect for each response, indicating the sign of the effect for a particular factor value ("high" or "low").

The final voltage (H) is directly associated to the thickness of the Al₂O₃ and it is expected that the higher the thickness, the lower the TFT mobility, "on" current and "off" current because of the lower electric field at the semiconductor/dielectric interface; this is supported by the data obtained in Table 5.5.



Figure 5.4: Pareto charts of the absolute effects on a) TFT mobility; b) TFT threshold voltage; c) voltage hysteresis; d) TFT "on" current; e) TFT "off" current; f) TFT on/off ratio.

The results are presented in Table 5.5, where the percentage contribution and the cumulative contribution from the most significant factors are computed. Considering as relevant the factors whose relative contribution to the effect is superior to 12.5%, we notice a better screening of the manufacture factors (cells highlighted in yellow on Table
5.5). The exception is the threshold voltage, which has 4 significant factors and still has 4 relevant factors, summing up a cumulative contribution superior to 85%.

Table 5.5: Relative Contribution to the Effects and Cumulative Relative Contribution from the Manufacture Factors Considering the (i) TFT Mobility, (ii) TFT Threshold Voltage, (iii) Voltage Hysteresis, (iv) TFT "On" Current, (v) TFT "Off" Current, and (vi) TFT On-Off Ratio as the Response Parameters.

		F	Response: carrier mot	oility		Response: threshold voltage					
		Factor	Contribution (%)	Cumulative Contribution (%)			Fa	ctor	Contribution (%)	Cumulative Contribution (%)	
-	у	Н	26.93	26.93		-	У	А	35.26	35.26	
-	у	В	22.63	49.56		-	у	С	20.17	55.42	
-	у	С	18.45	68.01		-	у	D	14.98	70.40	
-	у	А	17.97	85.98		-	у	н	14.80	85.21	
-	у	E	11.21	97.19]	-	n	F	5.76	90.97	
+	у	D	2.61	99.81]	+	n	E	5.02	95.98	
+	n	F	0.16	99.97]	+	n	В	3.95	99.93	
-	n	G	0.03	100.00]	-	n	G	0.07	100.00	
		Respo	onse: TFT channel "on	" current				Res	ponse: TFT channel "o	ff" current	
		Factor	Contribution (%)	Cumulative Contribution (%)		Factor Contribution (%) Cumulati				Cumulative Contribution (%)	
-	у	С	53.59	53.59	1	-	у	н	18.97	18.97	
-	у	В	19.12	72.71		-	у	Е	17.72	36.70	
+	у	D	11.72	84.44]	-	у	А	17.62	54.31	
-	у	Н	10.13	94.56]	-	у	С	17.08	71.39	
-	у	F	3.46	98.02		-	у	В	14.09	85.48	
+	n	E	1.09	99.11]	-	у	G	5.74	91.21	
+	n	G	0.49	99.60]	-	у	F	5.71	96.92	
-	n	A	0.40	100.00		-	у	D	3.08	100.00	
		F	Response: TFT on/off	ratio		Response: transfer curve hysteresis					
		Factor	Contribution (%)	Cumulative Contribution (%)			Fa	ctor	Contribution (%)	Cumulative Contribution (%)	
+	у	E	38.17	38.17		-	у	E	34.44	34.44	
+	у	А	19.30	57.47		-	у	G	24.75	59.20	
-	у	С	11.98	69.45		+	у	н	24.14	83.34	
+	у	Н	8.35	77.80		+	у	С	9.11	92.45	
+	у	G	6.38	84.18		+	у	В	4.05	96.50	
-	у	F	6.27	90.45		-	у	F	2.17	98.67	
-	у	В	6.16	96.62		-	n	D	0.75	99.43	
+	n	D	3.38	100.00		-	n	A	0.57	100.00	
+	pos	itive effe	ect	A: Thickness of Al-la	ayer				E: pH of the electrolyt	e	
-	neg	ative ef	fect	B: Al-evaporation ra	te				F: Current density		
у	sigr	nificant f	ator	C: Water content					G: Annealing		
n	nor	n-signific	ant factor	D: Temperature of e	elect	rolyt	е		H: Final voltage		
	con	tributio	n > 12.5%								

Hickmott reported that the maximum amount of charges introduced into the Al_2O_3 layer during the anodization process can be more than 100 times larger when aqueous electrolytes are used [204]. The effect of H_2O content in the electrolyte can be explained by considering the equations 5.1 1nd 5.2 used to describe the Al_2O_3 growth onto metallic Al in section 5.2. While water residue is likely to be present on the surface, some water molecules or oxygen atoms can also be trapped in the Al₂O₃ bulk, creating defects that affect the electrical properties of the oxide layer, as well as the oxide surface roughness.

Conduction in the TFT channel initiates when the gate electric field at the interface induces the filling of gap states, moving the Fermi level closer to the conduction band and increasing the charge carrier concentration. Therefore, for voltages below V_T, most of the charge carriers are in localized states and the transport is dominated by hopping, which results in low mobility values. The threshold voltage itself is mainly influenced by the presence of shallow states in the bandgap, which is affected by the thickness of the dielectric layer and the oxide dielectric constant. Moreover, the higher density of traps resulting from a thicker dielectric layer can be also responsible for the observed higher transfer curve hysteresis. The water content (C) in the electrolyte causes a higher Al₂O₃ surface roughness, which causes more carrier scattering at the dielectric interface, leading to lower mobility, as well as "on" current and "off" current. The lower threshold voltage for higher water content may be because most of the defects formed in the dielectric layer are due to dipoles and not to charged states.

The Al-layer thickness (A) and the Al-evaporation rate (B) are also factors which affect the anodized oxide layer morphology and conductivity in such a way that the higher these factors, the lower are the TFT mobility and "on" current (higher carrier scattering at the interface) and the "off" current (lower channel intrinsic current, probably to defect states). The data in Table 5.5 also indicates that the pH of the electrolyte (E) impacts negatively the transfer curve hysteresis, denoting that, for slower Al₂O₃ deposition rates (a more conductive electrolyte results in slower oxide growth), less charge traps are formed at the TFT semiconductor/dielectric interface. Hysteresis originates from the delay in releasing charges from traps. During the forward sweep of VG, traps with energy lower than the Fermi level are filled. During the reverse sweep, however, the charges are not released from the traps immediately, shifting the transfer curve compared to forward sweep. Therefore, interface states and traps in the bulk can be considered the main cause of hysteresis.

Factor E (pH of the electrolyte) is also the most important contribution to the TFT on/off ratio. Considering that both the "on" and "off" currents are negatively affected by the pH of the electrolyte, one concludes that the decrease of the "off" current is higher than the decrease of the "on" current, causing an increase on the on/off current ratio when

the factor is at its "high" level. A similar reasoning can be applied to explain why factor A (Al-film thickness) has a positive impact on the on/off ratio, despite its negative impact on both the TFT "on" and "off" currents. Another factor that has a high negative contribution to the transfer curve hysteresis (and, consequently to the charge traps density) is the annealing of the Al₂O₃ films before the ZnO sputtering. The substrate annealing promotes the desorption of adsorbed species (molecular oxygen, moisture, etc.) to the dielectric layer surface which can cause charge trapping at the semiconductor/dielectric interface during transistor operation.

5.5 ZnO TFT Overall Performance

A useful purpose of this work would be to identify the impact of the process factors on the overall performance of the ZnO TFTs. To analyse the effects of the manufacturing process, it is necessary to identify targets for each response parameter. We considered that, for the mobility, the "on" current and the on/off ratio, the best response is the highest value as possible, whereas, for the threshold voltage, the "off" current and the hysteresis, the best response is as low as possible. Using this criterion, Table 5.6 has been constructed, where the most appropriate target level (only H or L) was selected for each response. For each TFT response parameter, as different process factors could potentially lead to an optimized response, a method to identify the optimal process factor was needed. Therefore, we considered as a major criterion the highest number of occurrences of each target level, as observed, for instance, for factor A in Table 5.6 (3× H vs 1× L, resulting in a target level H). However, for factor H (final voltage), each target level occurs twice, requiring the use of a second criterion. We considered the highest sum of the percentage contribution of factor H to each response parameter (which can be obtained from Table 5.5), resulting in a target level L for factor H. As a result, the factor levels have an optimal combination of transistor performance in the last row of Table 5.6 shown below.

It can be clearly seen that that factor F (current density) does not contribute significantly to the overall TFT performance and can be ignored, even though ANOVA showed initially that it was significant for some response parameters. The scores of the factors for all parameters studied, calculated by ANOVA, are summarized in Table 5.7. The lines of the Table are organized by separating the manufacturing factors in three different groups: (i) factors prior to the anodization step (the Al layer thickness and evaporation rate), (ii) factors directly associated with the anodization process (H₂O content, temperature of the electrolyte, pH of the electrolytic solution, current density and final

voltage), and (iii) the postprocessing factors: in this case just the "annealing" step. Each cell of the Table representing a significant factor (based upon the 12.5% relative contribution stated before) contains a number which represents the ranking of the factor influence on the associated response parameter.

Table 5.6: Optimal Choice for the Level of Each Manufacturing Factor Considering the Individual Responses and the Overall Optimized Performance Considering the Six Response Parameters from the TFT Transfer Curve^a

		FACTORS							
	Α	В	С	D	Ε	F	G	Н	
RESPONSES									
Mobility	Н	L	L	n	n	n	n	L	
Threshold Voltage	L	n	Н	н	n	n	n	Н	
On current	n	L	L	n	n	n	n	n	
Off current	Н	Н	Н	n	н	n	n	Н	
On/off ratio	Н	n	n	n	Н	n	n	n	
hysteresis	n	n	n	n	L	n	Н	L	
Optimum value	Н	L	Н	Н	Н	n	Н	L	

^a Yellow cell with a "n" indicates that the factor influence can be neglected.

The scores of the factors for all parameters studied, calculated by ANOVA, are summarized in Table 5.7. The lines of this table are organized by separating the manufacturing factors in three different groups: (i) factors prior to the anodization step (the Al layer thickness and evaporation rate), (ii) factors directly associated to the anodization process (H₂O) content, temperature of the electrolyte, pH of the electrolytic solution, current density and final voltage), and (iii) the postprocessing factors: in this case just the "annealing" step. Each cell of the table representing a significant factor (based upon the 12.5% relative contribution stated before) contains a number which represents the ranking of the factor influence on the associated response parameter.

From Table 5.7, it is possible to identify that factors F and G (current density and annealing, respectively) are the factors that have the lowest overall impact upon performance considering all TFT response parameters. On the other hand, factors A, C, and H (Al-film thickness, H₂O content in the electrolyte, and final voltage, respectively) are the factors, which have the highest impact on the overall performance. To rank the factors which impact most the overall performance, we can consider two criteria: (i) the

number of times each factor appears as significant in Table 5.7 and (ii) the lowest sum of the order of significance in each line of Table 5.7. According to the first criterion, factors A, C, and H are tied. However, by the second criterion, factor H (final voltage) is the most relevant factor, since it has the lowest sum of the order of significance. Factors A and C are still tied by such criterion. Moreover, factor H affects most TFT mobility, which is commonly attributed as the most important TFT performance parameter. Therefore, Table 5.7 can be used to determine which process factors are the most important to the device performance, and which ones have a lower effect or can be neglected, in the case of the need of a full factorial DOE (when the interactions of the factors play an important role in the response parameters).

P	ARAMETERS		μ	Vth	Ion	Ioff	Ion/Io ff	∆V _H	
D	Thick Al layer	Α	4	1		3	2		
1)	Al Evap. Rate	В	2		2	5			
	H ₂ O content	С	3	2	1	4			
	Temp.Electrolyte	D		3					
II)	pH Electrolyte	E				2	1	1	
	Current density	F							
	Final voltage	н	1	4		1		3	
III)	Annealing	G						2	
I) Pre	-anodisation factors					Neg	Negative factor		
II) an	odisation factors	r				Positive factor			

Table 5.7: Comparative Table of the Significance Scores of Process Factors Considering All Studied Response Parameters^a

^a The number in each cell represents the order of significance found by ANOVA for the respective pair of factor and parameter. Only factors that presented relative contribution superior to 12.5% were considered.

5.6 Optimisation of Sputter Coated ZnO TFTs using Taguchi OA

The Plackett Burman approach was useful for optimisation of the anodisation process as there were many factors influencing the performance. Taguchi orthogonal array L9 (3⁴) was also used to optimize the ZnO TFT manufacture process parameters via the design of experiment (DOE). Taguchi method was used because a smaller number of factors are needed to optimise the ZnO TFTs manufacture process with greater non-linearity as factors are changed and moreover, it gives better accuracy.

The Taguchi OA factorial settings and properties of factors shown in Table 5.8. Four factors (ABCD) and three level (1,2,3) Taguchi design was selected for estimating the effect of different factors; A= sputter rate (0.5 Å/s, 1 Å/s, 2 Å/s), B = Temperature of post annealing (0 °C, 250 °C, 450 °C), C = Thickness of ZnO active layer (20nm, 40nm, 60nm), D =Environment of annealing (D = 100 % N₂, 50 % N₂and D = in air) on response parameters like mobility, threshold voltage, on-off drain current of ZnO TFTs.

Table 5.8: Taguchi OA Factorial settings and properties of RF sputtering, annealing process factors.

				Number	OA	Level	Level	Level
Abbreviation	n Name	Units	Туре	of	Column	1	2	3
				Levels	Index	(Low)		(High)
А	Sputter rate	Å/s	Quantitative	3	1	1	2	3
В	Annealing	⁰ C	Quantitative	3	2	1	2	3
С	Thickness	nm	Quantitative	3	3	1	2	3
D	Environment	NA	Quantitative	3	4	1	2	3

Shown in Table 5.9 the Taguchi orthogonal array L9 (3⁴) with 9 runs. Run is the experiment number performed for Optimisation of process parameters.

Table 5.9: Taguchi OA L₉ (3⁴) showing 9 experiment runs with 4-factors and 3-levels.

	L ₉ (3 ⁴) Orthogonal array									
Run Number		Performance								
	A B C D									
1	1	1	1	1	0.01					
2	1	2	2	2	0.22					
3	1	3	3	3	0.23					
4	2	1	2	3	0.07					
5	2	2	3	1	0.20					
6	2	3	1	2	0.30					
7	3	1	3	2	0.11					
8	3	2	1	3	0.14					
9	3	3	2	1	0.37					

5.6.1 Pareto Chart and Term Effects for Mobility

Shown in Figure 5.5 (a) is the Pareto chart showing the impact of changes in the input process factors on mobility. The thick blue line indicates the threshold value of the significance test. If the term effect is greater than the significant level, the factor is deemed significant. The factors lower than the significance level, the noise in the experiment is as greater as the effect of changing the factor, so it isn't considered as significant. It can be

clearly observed from this plot that annealing at 450 °C (B[3]) is most significant factor causing increase in mobility of ZnO TFTs. This is because post annealing removes physical defects, such as sputter damage, and stabilizes the oxygen concentration.



Figure 5.5: Pareto chart regression for (a) mobility, (b) Term effect plot for mobility.

It can be seen from Figure 5.5 (b) that there is an increase in mobility from lower sputter rates (0.5 to 1 Å/s-better uniformity). However, at a high sputter rate of 2 Å/s the mobility falls due to non-uniformities and higher amorphous nature of the deposited active layer. The structure and morphology of the ZnO films are dependent on temperature, gas flow rate and time of annealing. High temperature annealing larger then than 400 $^{\circ}$ C led to bigger grain size and better crystallinity. The significance of the effect of annealing on the increase in the saturation mobility can be seen in Figure 5.5 (b). Maximum mobility of 0.37 cm² V⁻¹ s⁻¹ at 450 $^{\circ}$ C was achieved due to better crystallinity of ZnO film caused by post annealing. Moreover, annealing treatment minimizes the crystalline defects, and hence, enhances the field effect mobility.

Also, it can be observed from the term effect plot that with the increase in channel thickness mobility increases. This is due to an enhancement in the crystalline quality and channel conductance of the ZnO films as the channel thickness increased.

Annealing under 100 % N₂ atmosphere was performed and it can be seen from Figure 5.5 (b) that mobility under this environment is the highest. This is because nitrogen is inert and reacts with virtually nothing under the conditions used for semiconductor processing. Since the nitrogen is a non-reactive gas (neutral) that does not participate in any oxidation or reduction reactions, the trap states at the interface were minimised. Overall, the results show that post annealing in the inert environment leads to better performance of ZnO TFTs and thus is the most important factor to control during the fabrication.

5.7 Comparison of Sprayed and Sputtered ZnO TFTS

In this section comparative study of sputtered and solution-processed ZnO films and the impact upon TFT performance is descried. Sprayed TFTs on SiO₂ substrates were manufactured by Prof. Lucas Fugikawa Santos group in UNESP, Brazil and all testing was carried out at Bangor University. Sputtered TFTs on SiO₂ substrates were manufactured at Bangor based on Taguchi OA optimisation process. This work focuses on comparing spray and sputtered ZnO TFTs, differences in the microstructure and chemical composition are studied and reasoning for these discrepancies is given based on XPS analysis of the ZnO films.

5.7.1 Optical Spectroscopy of ZnO Films

In this section optical properties of the ZnO films deposited from sputtering and spray deposition are compared. A Tauc plot provides means for determining the optical band gap (E_g) in ZnO and can be used to understand if different processing routes and annealing temperatures impact on the E_g , which is shown in Figure 5.12. The optical band gap values were obtained by extrapolating the linear part of the curves in Figure 5.12 (a) and (b) with the intercept of this linear part with the x-axis. The absorption coefficient (α) can be expressed by applying the Tauc model in the high absorption region, equation (5.5) [205].

$$(\alpha h\nu)^2 = A(h\nu - E_a)$$
 Eq. 5.5

Figure 5.6 (a), (b) shows the plot of $(\alpha h\nu)^2$ versus photon energy (hv) for sputtered and spray coated ZnO films, respectively with the calculated values of E_g summarised in Table 5.10. It is apparent that annealing and processing conditions only moderately affect the film's optical bandgap but that spray coated films possess a lower optical bandgap than sputtered ones. The value of bandgap energy (E_g) agrees approximately with the room temperature bandgap of ZnO (3.3 eV) [206][207]. The decrease in optical E_g with increasing annealing temperature can be attributed to improvements in the atomic structure order which also results in bandgap narrowing; a view supported by Jeon *et al* [208]. Based upon the data in Table 5.10, it is plausible to state that large improvements in atomic structure can be obtained from the annealing of sputter deposited films than spray coated ones by annealing.



Figure 5.6: Plot of $(\alpha h\nu)^2$ versus photon energy $(h\nu)$ of ZnO film with different annealing temperatures (a) Sputtered ZnO film and (b) Sprayed ZnO film.

Table 5.10: Band gap energy, E_g of the ZnO films at different annealing temperatures for sputtered and sprayed samples.

Annealing Temperature	Bandgap energy Sputtered [eV]	Bandgap energy Sprayed [eV]
200	3.30	3.28
350	3.28	3.26
450	3.26	3.25

5.7.2 Atomic Force Microscopy of ZnO Films

The surface properties of ZnO films were investigated using Atomic Force Microscopy (AFM). AFM topography image analysis reveals that smoother films are achieved when they are deposited by sputtering at room temperature, as confirmed in Figure 5.7. However, there is a very small increase in roughness at the higher temperatures of annealing for both sputtered as well as spray deposited films. The surface roughness of the films was evaluated in terms of root mean square (R_q), average (R_a) and peak to valley (R_{max}) and are shown in Table 5.11. This small increase in roughness can be attributed to increases in grain sizes, which can be verified from the AFM images in Figure 5.7. It appears that as the temperature of the substrate is increased, particularly for values greater than 200 °C, there is a tendency for ZnO to form larger crystallites, resulting in a more irregular surface. The dynamics and the irregularity of the size of the drops from

solution sprayed onto substrate also contributing to increase in roughness due to the low homogeneity in the distribution of the drops sprayed along the substrate [174].



Figure 5.7: (a), (b), (c) sputtered AFM topography images of sputtered and (d), (e), (f) sprayed ZnO films annealed at different temperatures shown above.

Table 5.11: Surface roughness parameters of sprayed and sputtered active layers of ZnO TFTs measured from an average of 6 samples.

Annealing	Sprayed Ro	oughness Paran	neters	Sputtered Roughness Parameters				
Temperature [°C]	R _q [nm]	Ra [nm]	R _{max} [nm]	R _q [nm]	Ra [nm]	R _{max} [nm]		
200	1.43	1.09	14.3	0.60	0.48	5.30		
350	1.51	1.15	14.6	0.62	0.49	5.40		
450	1.96	1.55	17.6	0.74	0.56	11.15		

5.7.3 X-Ray Diffraction of ZnO Films

X-Ray diffraction (XRD) has been used to analyse the differences between the microstructures of spray and sputtered samples. In Figure 5.14, the XRD scans for the higher temperature samples which were annealed at 350 °C temperature to 450 °C are shown. The XRD patterns show substantially different diffraction patterned with a single prominent peak in the case of sputtered films.



Figure 5.8: (a) Sputtered and (b)sprayed ZnO films diffraction patterns annealed at different temperatures shown above.

For the case of the spray coated films, four diffraction peaks are present which corresponds to the planes (100), (002), (101) and (110), indicating the presence of film crystallites with different orientations. In the case of spray coated samples, the intensity of these peaks increases as the annealing temperature increases, thus indicating that the crystallinity of the film was improved. Average crystallite size obtained using Scherrer equation (5.6) [209] 14.88 nm for 300 °C and 17.32 nm for 450 °C for sprayed samples and 11.67 nm for 300 °C and 13.72 nm for 450 °C for sputtered samples.

$$d = \frac{0.9\lambda}{B\cos\theta}$$
 Eq. 5.6

Where, λ , θ and B are the X-ray wavelength, Bragg diffraction angle and line of full width at half maximum, respectively.

Transistor devices did not work at lower temperatures such as 200 °C annealing because the crystallisation of ZnO films only takes place at temperatures approximately annealing temperature is greater than 300 °C. Wurtzite structure is the dominant polycrystalline phase when the deposition/annealing temperatures are in the range of 300–450 °C. The rise of (002) peak intensity and shift in the position of (002) peak can be attributed to better crystallinity with greater relaxation caused by annealing. All the diffraction peaks of these XRD patterns are in good agreement with the (JCPDS data Card No. 36-1451).

5.7.4 XPS Analysis of the ZnO Films

All sputtered samples used for XPS studies were prepared at Bangor University cleanroom. XPS measurements for sputtered ZnO films were done at Chester University and analysis was performed by Dinesh Kumar, Dr. Jeff Kettle and Prof. G.C Smith.



Figure 5.9: XPS survey scans from sputtered (a) and spray-coated ZnO films.

The XPS data from sputtered ZnO films were acquired using a bespoke ultra-high vacuum system fitted with a Specs GmbH Focus 500 monochromated Al Ka X-ray source, Specs GmbH Phoibos 150 mm mean radius hemispherical analyser with 9-channeltron detection, and a Specs GmbH FG20 charge neutralising electron gun. XPS data from spraycoated films were obtained using a K-Alpha, Thermo Scientific spectrometer from National Laboratory for Nanoscience in Campinas, Brazil. Survey spectra were acquired over the binding energy range 1100 – 0eV using a pass energy of 50eV and high-resolution scans were made over the C 1s, O 1s and N 1s photoelectron lines (where detected) using a pass energy of 20 eV. Under these conditions, the full width at half maximum of the Ag $3d^{5/2}$ reference line is < 0.8eV. In each case, the analysis was an area-average over a region approximately 2mm in diameter on the sample surface. The energy scale of the instrument is calibrated according to ISO standard 15472, and the intensity scale is calibrated using an in-house method traceable to the UK National Physical Laboratory. Data were quantified using Scofield cross sections corrected for the energy dependencies of the electron attenuation lengths and the instrument transmission. Data interpretation was carried out using CasaXPS software v2.3.16. The spot area was a 400 x 400 μ m² region.

XPS spectra from all samples (both sputtered and sprayed) showed strong Zn and O peaks, with C also present. The annealed sputtered sample also showed a very low level of Cl. As an example, the survey scan for the as-prepared sputtered sample is shown in Figure 5.9 (a) and for the spray-coated films, in Figure 5.9 (b). The results of the quantification of the data are shown in Table 5.12.

Element	Position	Surface concentration, atom %							
element		Sputter	ed Films	Spray-coated films					
anu inie	[ev]	As-prepared	Annealed	300 [°C]	450 [°C]				
C 1s	284.8	39.12	27.79	22.75	22.53				
0 1s	530-532	38.59	42.33	48.07	48.69				
Zn 2p ^{3/2}	1022	22.29	29.71	29.18	28.77				
Cl 2p		-	0.16	-	-				

Table 5.12: Surface composition for sputtered and spray-coated ZnO films.

Table 5.13: Surface compositions after application of the hydrocarbon contamination overlayer correction (excluding the very minor l contribution in the case of the annealed sample.

		As prepared	Annealed
	Hydrocarbon layer	1.75	1.15
	thickness (nm):		
Element and line	0 1s	52.94	51.78
	Zn 2p ^{3/2}	47.06	48.22
Ratio O/Zn		1.13	1.07

The data shows that annealing of the sputtered films reduced the amount of surface carbon present and also reducing the ratio of oxygen to zinc from ~ 1.7 to ~ 1.4. Stoichiometric ZnO is expected to show a ratio of O/Zn of 1. The difference is probably due to preferential attenuation of the Zn $2p^{3/2}$ signal compared to the O 1s signal by the surface carbonaceous overlayer, as a consequence of the lower kinetic energy of the Zn $2p^{3/2}$ photoelectrons. To correct the measured surface compositions for the effect of the hydrocarbon overlayer, the method of Smith (2005) is applied. In this method, the carbon composition is used to estimate the thickness of the hydrocarbon contamination overlayer, the effect of a layer of this thickness on the attenuation of signals from the underlying material is estimated, and the results are re-normalised to 100 % atoms. The results of applying this correction for sputtered samples are shown in Table 5.13.

The results show a much closer approach to the stoichiometric ZnO, as expected, but a smaller Zn content in the spray coated film, indicating a lower level of C impurities. The slight excess of O is most likely due to the presence of some carbon-bonded oxygen. Hydrocarbon contamination in spray-coated films was not so significant and the correction procedure was no applied.

5.7.5 Surface Chemistry

The Zn 2p spectra from the spray coated samples are shown in Figure 5.10. The Zn 2p spectra present a spin-orbit splitting (2p^{3/2} and 2p^{1/2}) distancing approximately 23.1 eV, with the low binding energy peak around 1021.2 eV. These peaks are consistent with Zn²⁺ ions in the crystalline lattice, which is the case for ZnO. No significant changes were observed for spray-coated and sputtered samples, irrespective of annealing treatment. Further clear evidence that the observed Zn peaks are associated with the presence of ZnO comes from the intense components of the Auger peaks found at kinetic energies of approximately 988.4 eV (binding energy equivalents of approximately 498.2 eV).



Figure 5.10: Zn 2p^{3/2} and LMM X-ray excited Auger spectra for the as-prepared samples.

From these results, the samples consisted of near-pure stoichiometric ZnO with a hydrocarbon overlayer due to absorption from the environment, as is usually seen in XPS analysis of air-exposed samples. Annealing did not appear to change the chemistry or stoichiometry of the ZnO; however, it did result in a reduction in the amount of hydrocarbon contamination. If the hydrocarbon was assumed to be present as a uniform thin overlayer, then its thickness was reduced from approximately 1.75 nm to approximately 1.15 nm on annealing.

For spray-coated films, where hydrocarbon contamination did not play an important role, annealing seems to change slightly the sample chemistry and

stoichiometry, leading to a higher amount of oxygen in metallic bonds in the crystalline lattice and a higher amount of oxygen vacancies. Both cases corroborate the improved electrical properties of the devices, discussed later.



Figure 5.11: (a) sputtered untreated high-resolution spectra for O 1s, (b) sputtered annealed at 450 °C high-resolution spectra for O 1s (c) sprayed at 300 °C high-resolution spectra for O 1s (d) sprayed at 450 °C high-resolution spectra for O 1s.

High resolution scans over the O1s peaks (for sputtered and sprayed samples) are shown in Figure 5.11 also, curve-fitted to known chemical state reference data. The spectra from the sputtered films show a strong component at approximately 530.4 eV and a broader component at approximately 532 eV. The sharper component at lower binding energy is attributed to oxygen in metallic bonds (in ZnO in this case), with the broader component at higher binding energy due to oxygen in a range of unresolved carbon bonding configurations, consistent with those observed in the C 1s peak. The difference between as-prepared and annealed samples is consistent with the reduction in the hydrocarbon contamination layer (and its associated oxygen bonds) on annealing.

High resolution scans over the C 1s peaks from sputtered samples are shown in Figure 5.12, curve-fitted to known chemical state reference data. Both C 1s spectra show strong hydrocarbon peaks at 285.0 eV, with weaker components at approximately 286.5

eV and 289.0 eV due to C-O (hydroxyl) and COO- (surface acid groups). After annealing, there was a small drop in the relative proportion of acid groups compared to the non-annealed samples (11% of the C 1s intensity, compared to 15% of the C 1s intensity as prepared). However, any differences in the chemical state after annealing appear minor. Similar peaks (but less intense) were observed in spray-coated samples (HR scans not shown).



Figure 5.12: (a) Zn 2p spectra for spray-coated films annealed at 450 °C, (b) Zn 2p spectra for spray-coated films annealed at 300 °C, Carbon 1s spectra for sputtered samples (c) untreated, (d) annealed at 450 °C.

Spray-coated films have much narrower O1s spectra, probably due to the longer annealing time (1 hour). The spectra were fitted considering three major contributions, as shown in Table 5.14. The low-energy and more intense component at 530 eV is attributed to oxygen in metallic bonds, forming crystalline ZnO. Two smaller components are observed at 530.8 eV and 531.7 eV. The higher energy component can be attributed to oxygen in ZnOO₃ bonds since the spray-coated ZnO films are formed by the pyrolysis of an organic precursor salt, zinc acetate dihydrate. The intermediate component at 530.8 eV is commonly attributed to oxygen atoms in oxygen deficient regions caused by oxygen vacancies [210]. Table 5.14 shows that the annealing at 450 °C increases the amount of oxygen in the crystalline lattice, simultaneously by the decrease of the amount of oxygen in ZnOH and ZnCO₃, which is consistent with the picture of higher conversion from zinc acetate to ZnO. Moreover, the increase of the amount of oxygen vacancies for the annealed sample is in agreement with the observed improvement of the electrical properties of the films, since the n-type conductivity of ZnO can be associated to oxygen vacancies [211].

Component	Spray Annealed at 300) [ºC]	Spray Annealed at 450 [ºC	t 450 [ºC]		
component	Position	%At Concentration	Position	%At Concentration		
0 1s (a)	530.04	60.81	529.93	69.69		
0 1s (b)	530.80	10.47	530.79	15.40		
0 1s (c)	531.75	28.73	531.58	14.91		

Table 5.14: Component assignment for the O 1s spectra from the spray-coated samples.

Table 5.14 can also be used to correct the surface composition concerning the stoichiometric composition of spray-coated ZnO films. By considering only the amount of oxygen in metallic bonds in the crystalline lattice compared to the Zn 2p peak, we find a stoichiometric O:Zn ratio of 1.00 and 1.18 for the samples annealed at 300 °C and 450 °C, respectively.

Overall, the higher saturation mobility in case of sprayed as compared to sputtered TFTs is attributed to the higher oxygen levels in as prepared as well as annealed samples.

5.8 Electrical Characterization of ZnO Thin-Film Transistors Fabricated using Spray and Sputter Deposition

The electrical properties of the sputtered and spray deposited ZnO films were also investigated using optimised bottom-gate, top-contact thin film transistor. A comparative set of the transfer and output characteristics for ZnO TFTs (W/L = 5.55) deposited from spray and sputter processes at various processing temperatures are shown in Figure 5.13 and 5.14 respectively.

As shown in Table 5.15, the sprayed TFTs exhibit excellent operating characteristics with the maximum electron mobility of $15.55 \text{ cm}^2 \text{V}^{-1} \text{S}^{-1}$ and current on/off

ratio of 2.1×10⁶. However, for sputtered TFTs, lower mobility of 1.19 cm²V⁻¹S⁻¹ and on/off ratio of 7.67×10³ is reported. The better performance in case of spray deposition is probably due to the better initial crystal formation of the ZnO active layer, whereas sputtering occurs at room temperature and is subsequently post-annealed, as supported by the XRD data.

Table 5.15: Performance paramet	ers of spra	yed and	sputtered	Zn0	TFTs a	t differen
annealing temperatures.						

Annealing	Spra	yed performa	nce	Sputtered performance				
temperature		Parameters]	Parameters			
(°C)	μ _s [cm ² V ⁻¹ s ⁻¹]	V _T [V]	I _{on} /I _{off} [ratio]	μ _S [cm ² V ⁻¹ S ⁻¹]	V _T [V]	I _{on} /I _{off} ratio		
300	1.34	10.38	1.34E+06	0.21	9.11	3.52E+03		
350	1.63	8.92	4.6E+05	0.68	6.78	1.45E+03		
450	15.55	4.48	2.1E+06	1.19	5.51	7.67E+03		

The ZnO TFT transfer and output curves were stable when measured after 8 week in the dark storage. In Table 5.15, the performance parameters are shown as a function of annealing temperature. For both sputtered and spray coated devices, the threshold voltage (V_T) decreases, and the saturation mobility (μ_s) increases with increasing annealing temperature. This is likely to be due to a reduction in impurities at insulator/semiconductor interface and removal of any moisture in device structure which can degrade the device performance. Also, at higher annealing temperatures, more electrons can escape from the localized states and contribute to the free carriers in the conduction band, which causes higher mobility and smaller threshold voltage [212]. Increased mobility after annealing is attributed to increased oxygen levels as confirmed by the XPS analysis. Improved device performance by increasing the annealing temperature might come from better crystallinity (or higher average crystallite size), improving the inter-grain charge transport. Moreover, there would be less influence from defects at the grain boundaries, which can cause charge-trapping and/or scattering. XPS data have also shown an increase in the Zn amount in metallic bounds, which also means that there is more crystalline ZnO at elevated annealing temperatures.



Figure 5.13: Transfer characteristics of sprayed (a) annealing at 300 °C, (b) 350 °C, (c) 450 °C and sputtered (d) annealing at 300 °C, (e) 350 °C, (f) 450 °C ZnO TFTs annealed at different temperatures shown above.



Figure 5.14: Output characteristics of sprayed (a) annealing at 300 °C, (b) 350 0C, (c) 450 0C and sputtered (d) annealing at 300 0C, (e) 350 0C, (f) 450 0C ZnO TFTs annealed at different temperatures shown above.

5.9 UV Saturation Experiment

UV saturation experiment on ZnO TFTs manufactured using spray, as well as sputtered deposition methods to see the effect of UV light on the TFT characteristics have conducted.

Figure 5.15 (a) shows transfer characteristics of a ZnO TFT manufactured by spray coating and 5.15 (b) sputter coated before (black curve) and after (in blue) exposure to a UV-355 nm LED for 5 minutes at an irradiance of 0.214 mW/mm². In both

cases, the characteristics changed dramatically and shown is the behaviour after 5 minutes of light exposure, by which time the devices had saturated in terms of their electrical changes. Such behaviour is commonly known as "persistent photoconductivity". The ZnO TFT performance parameters including μ s, V_T and on-off ratio were evaluated before and after exposure to UV light.



Figure 5.15: Transfer curves for a ZnO TFT with the active layer deposited by (a) spraycoating, (b) sputter coating before and after UV exposure for 5 minutes at an irradiance of 0.214 mW/mm^2 .

Both types of TFTs show an overall increase in these parameters after UV exposure. However, with a more accentuated effect in the "off-current" region, where the gate bias is smaller than the threshold voltage. The relative change in these performance parameters is higher in the case of sprayed TFTs, suggesting that sprayed TFTs are more sensitive to UV irradiation. This can be attributed to higher surface roughness in case of sprayed TFTs as confirmed by AFM data, increased surface area might be responsible for higher adsorption of molecular oxygen onto the film surface. This increased conductivity effect occurs due to the photo-stimulated desorption of adsorbed oxygen molecules in the ZnO film. In the absence of UV light, oxygen molecules can be adsorbed on the surface as negatively charged ions by capturing the free electrons from the *n*-type semiconductor, thereby create a depletion layer with low conductivity near the surface. When the sample

is exposed to UV light with photon energy in the UV, it generates electron-hole pairs which increase the conductivity.



Figure 5.16: Decay time of the sprayed ZnO TFT (a) mobility, (b) OFF-current, (c) Drain on-current and sputtered ZnO TFT (d) mobility, (e) off-current, (f) on-current at different temperatures.

Shown in Figure 5.16 (a), (d) is saturation mobility, (b), (e) TFT OFF-current, and (c), (f) TFT on-current, the decay behaviour of sprayed (left), sputtered (right) the TFT with respect to time elapsed after the UV excitation was switched off for temperatures varying from 40 °C until 102 °C. The mobility for sprayed TFTs increases with temperature and presents a slow decay time at 40 °C. However, the sputtered TFTs show similar mobility values at all temperatures of measurements. By increasing the temperature, the decay of the photo-stimulated mobility becomes faster and nearly no temperature dependence can be observed. The sprayed TFT on-currents Figure 5.14 (c), however, presents a larger variation, with a faster decay than the carrier mobility Figure 5.16 (a) at low temperature (40 °C) and almost invariant behaviour at higher temperatures (above 75 °C), indicating that the on current increase is not only due to the

increase in the carrier mobility by UV illumination but also by the increase in the free charge carrier density in the transistor channel due to elevated temperature excitation energy. The off-current, which is a direct measure of the ZnO film conductivity, presents an even higher variation with temperature and a faster time decay Figure 5.16 (b) at lower temperatures. The initial value of the off-current is strongly dependent on the temperature, however, for temperatures above 75 °C, a temperature dependence cannot be observed for longer times.

From the observed results, it can be confirmed that, even though the increase on the temperature causes an increase on the UV-stimulated carrier mobility, it is also responsible for accelerating the adsorption of the oxygen molecules to the ZnO TFT active layer, which contributes to the decrease in the carrier mobility and the semiconductor conductivity. The TFT off-current has a faster decay time than the on-current because the contribution to the off-current comes from the conductivity of the whole ZnO layer, which has a larger area for oxygen adsorption, whereas the contribution to the on-current comes from the conductivity of the ZnO/dielectric interface, which represents a small portion of the whole ZnO film.

Figure 5.11 shows the temperature dependence of the sprayed TFT mobility (a), off current (c), and of the sputtered ZnO TFT mobility (b), off-current (d) immediately after the UV light was switched off. The TFT mobility increases with temperature, with an activation energy of about 140 meV for sprayed and 53 meV for sputtered TFTs. The OFF-current, as mentioned above, decreases with temperature, due to the adsorption of oxygen molecules from the ZnO film. This adsorption process is thermally activated and has an activation energy of 9.6 meV for sprayed and high activation energy 229.4 meV for sputtered samples. The adsorption process is thermally activated, introducing trapping centers that cause a decrease in the TFT device current. The density of free carriers (proportional to the TFT off current) can be expressed as equation (5.7), where, ΔE is the activation energy of the adsorption process.

$$n = n_0 \left[1 - exp\left(-\frac{\Delta E}{k_B T}\right)\right]$$
 Eq. 5.7

Figure 5.17 shows that the adsorption process for sprayed samples has a very low thermal activation energy, which causes a strong dependence of the off current on the

temperature. On the other hand, the activation energy for the adsorption process for sputtered devices has higher activation energy, meaning that the off current has little influence on the temperature, as observed in Figure 5.17. Overall the results show that the sputtered samples are less sensitive to UV irradiation because the charge transfer is quicker.



Figure 5.17: Temperature dependence of the sprayed TFT mobility (a), off current (c), and of sputtered ZnO TFT mobility (b), off-current (d) immediately after the UV light was switched off.

5.10 Summary

The optimisation of Al_2O_3 dielectric and ZnO deposition process using a combinational, DOE approach is reported. Correlations between the manufacturing process parameters and the device characteristics were investigated. An important observation from the results is that the manufacturing parameters of the dielectric layer of the TFTs have a strong influence on the device performance parameters, presenting a comparable effect (or even higher) on the device characteristics than the variation of the dielectric layer surface treatment parameters, which are commonly investigated in the literature. This procedure is particularly interesting for optimization of other devices in printed/flexible electronics to improve the fabrication process and to speed up the development of new low-cost products.

ZnO TFTs are well suited to UV sensing applications because they absorb predominantly in the UV region due to the wide bandgap ($E_g = 3.37$ eV). From the perspective of UV sensing, TFT on-current in the saturation region appears to be the best performance parameter in a ZnO TFT for examining differences in UV exposure.

The effect of the substrate temperature during ZnO deposition by spray pyrolysis and annealing temperature after sputtering to the structural, optical, and electrical transport properties has been investigated. Sprayed and sputtered ZnO film structural characterization demonstrated the formation of polycrystalline ZnO only at deposition temperatures above 200 °C. Further increase of the substrate deposition/annealing temperature to 450 °C resulted in films with enlarged crystal size. Consequently, thin-film transistors based on ZnO films grown at substrate temperatures in the range of 300 to 450 °C exhibit the highest electron mobility. It has been shown that the electrical characteristics of both ZnO TFT sprayed and sputtered are altered after their interaction with UV radiation. However, the magnitudes of these changes are larger in sprayed ZnO TFT than sputtered as their surface structure has a greater affinity with the absorption and desorption of oxygen molecules in its surface. The present results demonstrate the possibility of depositing high optical and electronic quality ZnO films onto large area substrates using spray pyrolysis, which is attractive for the electronic industry, since the pyrolysis spray technique is inexpensive; it is not need vacuum steps and is easily applied to large-scale electronics.

Chapter 6. Effect of Bias Stressing on Indium Gallium Zinc Oxide TFTs and Flexible TFTs

In this chapter, the development of Indium-gallium-zinc-oxide (IGZO) thin film transistors (TFTs) is discussed. IGZO TFTs have advantages of high transparency in the visible range due to being wide bandgap semiconductor, high field-effect mobility and low temperature for device fabrication [213], [214], which are mandatory requirements for high performance display applications [215]. IGZO TFT performance shows great potential for future low power and flexible electronics applications. However, the stability of amorphous oxide semiconductor (AOS) TFT under the various stress has been an issue for practical applications [216]. It is well known that the total trap density affects the behaviour of the threshold voltage shift (ΔV_T) in IGZO TFTs [217], [218]. Therefore, the effects of positive bias stress are investigated in inorganic TFTs based on IGZO as the semiconductor and silicon dioxide as the gate insulator, with measurements carried out over a range of positive bias stress. Bias stressing was performed by applying positive gate bias stress for a particular time, measure output, transfer characteristics, and recover the device to initial state using white light illumination and again applying bias for next period

In this chapter, the effect of positive gate biasing and CYTOP (a fluoropolymer) passivation on TFTs is discussed. An application of gate bias stress leads to a shift in the threshold voltage of IGZO TFTs and after it is removed, the threshold voltage (V_T) eventually returns to its original value. The underlying physical mechanisms for the shift in threshold voltage during the application of the bias and after the removal of the bias stress are investigated.

To increase the carrier mobility of the IGZO TFTs, a high dielectric constant (highk) material has been used as the gate insulator to improve the electrical performance of IGZO TFT[219], [220]. Therefore, the performance of low voltage IGZO TFT devices using high-k dielectric aluminium oxide (Al₂O₃) on plastic substrates is described. Moreover, based on AgNW TCEs developed in chapter 4, AgNW film gated IGZO TFTs electrical, optical performance was also investigated and compared with TFTs on incumbent ITO transparent electrodes.

6.1 Experimental details

In this work, a flexible Kapton type HN polyimide film (25µm thick) was used as the substrates, which were supplied from RS components ltd, UK. Immediately following cleaning and drying, the samples were transferred to Leybold thermal evaporator and Al metal (80nm) was deposited on these film substrates. E-beam evaporation system was used to deposit gate dielectric. The evaporation source materials were Al₂O₃ (1mm to 3mm clear pieces with 99.99% purity). The chamber was then pumped down to a base pressure of 8×10⁻⁶ torr and the deposition rate was kept below 2 Angstrom/second for uniform layer deposition. The film thickness (168nm) was measured using the Dektak surface profilometer. After manufacturing a 20 nm thick IGZO channel layer by using radio frequency (RF) sputtering technique, TFTs were post annealed at 250 °C for 1 hour and cooled down to room temperature. Post annealing is performed because it removes physical defects, such as sputter damage, and stabilizes the oxygen concentration.

IGZO TFTs with bottom-gate, top-contact structure were fabricated on SiO₂ substrates obtained from Micro Chemicals GmbH. All steps of fabrication including sputtering, thermal evaporation and annealing were carried out in the class 100 cleanroom. Firstly, substrates were thoroughly cleaned with solvents followed by drying with a jet flow of nitrogen and oxygen plasma treated for 5 minutes. Thin-film transistors were fabricated by depositing a thin IGZO active layer on top of commercial SiO₂/Si substrates using RF sputtering. The dry thermally grown 100 nm thick SiO₂ layer acts as the TFT dielectric layer whereas the p-doped (boron) Si substrate (500 µm thick) is used as the TFT gate electrode. A thin layer of aluminium (100 nm thick) was thermally evaporated on the bottom of the Si substrate to assure good contact with the gate electrode. On top of the IGZO active layer, Al contacts were thermally evaporated using a Leybold 250 Univex thermal evaporator through a metallic shadow mask to make drain and source top electrodes. The post-thermal-annealing (450 ° C) process is used to repair the interface traps in the IGZO thin film and to reduce the bulk traps/defects in the channel. An organic- passivation layer (CYTOP) was spin coated inside glove box inert environment for the passivating active layer of TFTs and annealed at 160 °C for 30 minutes. CYTOP-organic-passivation has the following characteristic: high transparency, electric insulation, water and oil repellence, chemical resistance, and moisture-proof property-"wettability control". Passivating IGZO TFTs with CYTOP in the inert environment (Glove box) leads to moisture control over the device. CYTOP is a hydrophobic material and its passivation minimizes environment-related reactions. Hydrophobic molecules are usually nonpolar, meaning the atoms that make the molecule do not produce a static electric field.



Figure 6.1: (a) Transfer characteristics of unpassivated and CYTOP passivated IGZO TFTs, (b) Output characteristics of CYTOP passivated IGZO TFT and (c) Output characteristics of unpassivated IGZO TFT.

6.2 Effect of CYTOP Passivation on Performance of IGZO TFTs

Shown in Figure 6.1 are the output and transfer characteristics of pristine and passivated IGZO TFTs. The TFTs with the CYTOP passivation layer exhibited a relatively good electrical characteristic ($\mu_s = 20 \text{ cm}^2 \text{V}^{-1} \text{s}^{-1}$) compared with that of the unpassivated TFT ($\mu_s = 17 \text{ cm}^2 \text{V}^{-1} \text{s}^{-1}$). CYTOP passivation improves electrical characteristics, reduces the threshold voltage and increases mobility. The CYTOP passivation was performed at low annealing temperature (160 °C), and therefore, it is compatible with polymer substrates.

6.3 Bias Stressing of IGZO TFTs

In this section, results for fabricated IGZO TFTs employing a organic-passivation layer

(CYTOP) are explained and compared with un-passivated IGZO TFTs. CYTOP is an amorphous fluoropolymer with a high light transmittance of more than 95%.

6.3.1 Comparison of Unpassivated Positive Gate Bias Stress

The effect of positive bias stress (PGBS) on the electrical performance of an IGZO TFTs on SiO₂ was investigated by measuring the transfer characteristics in the dark before and after stress. In this case, the initial state was achieved by applying the recovery procedure using white light discussed next section. Shown in Figure 6.2 the evolution of transfer characteristics of un-passivated and passivated IGZO TFTs as a function of stress time, demonstrating the time-dependence of the bias stress resulting from applying a constant gate voltage for 11000 seconds, (a) un-passivated V_G = 5 V and drain voltage, V_D of 0V, (b) un-passivated V_G = 20 V, V_D of 0V, (c) un-passivated V_G = 40 V, V_D of 0V, (d) passivated V_G = 5 V and drain voltage, V_D of 0V, (e) passivated V_G = 20 V, V_D of 0V, (f) passivated V_G = 40 V, V_D of 0V, (her the transfer V_G = 40 V, V_D of 0V, (her the transfer V_G = 40 V, V_D of 0V, (her the transfer V_G = 40 V, V_D of 0V, (her the transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer V_G = 40 V, V_D of 0V, (her transfer transfer V_G = 40 V, V_D of 0V, (her transfer tran

As seen in Figure 6.2, the transfer characteristics display a similar pattern i.e. a simple translation along the V_G axis in the direction of the applied positive electric field. There is negligible hysteresis between forward and reverse scans, neither before nor after any applied stress condition. It is observed throughout the stress period, that the subthreshold region continues to shift positively while off-current is reduced in both unpassivated, as well as passivated devices. However, there is a reduction in the magnitude of saturation current, I_D , after stressing for un-passivated devices whereas this reduction in saturation current I_D after stressing for passivated devices is negligible. This is presumably due to low trap levels in passivated devices as compared to un-passivated ones.

To better understand the physics behind the positive bias stress induced threshold voltage shift in RF sputter IGZO TFT, we performed a detail numerical analysis based on the stretched-exponential model. The carrier injection from conducting channel and the subsequent charge trapping plays an important role in the IGZO TFT positive bias instability. The dependence of ΔV_T on time is generally described by the stretched exponential function given equation 6.1.

$$\Delta V_T(t) = |V_T(\infty) - V_T(0)| \left[1 - e^{-\left(\frac{t}{\tau}\right)^{\beta}} \right]$$
 Eq.6.1

Where, $V_T(\infty)$ is the threshold voltage after long stress times, $V_T(0)$ its value prior to bias stress, τ is a characteristic time constant and β the stretching factor describing the distribution in trap times. The solid curves in Figure 6.3 (a) represents the stretched exponential fitting of equation (6.1) to the data using the extracted fitting parameters listed in Table 6.1 zero drain voltage, $V_D = 0$ V and different gate-bias stress conditions (V_G = 40 V and 10 V).



Figure 6.2: Transfer characteristics of un-passivated and passivated TFTs showing the timedependence of the bias stress resulting from applying a constant gate voltage, (a) Unpassivated $V_G = 5 V$ and drain voltage, V_D of 0V, (b) un-passivated $V_G = 20 V$, V_D of 0V, (c) un-

passivated $V_G = 40$ V, V_D of 0V,(d) passivated $V_G = 5$ V and drain voltage, V_D of 0V, (e) passivated $V_G = 20$ V, V_D of 0V, (f) passivated $V_G = 40$ V, V_D of 0V, in all cases the characteristics were measured with $V_G = 40$ V, $V_D = 40$ V.

Table 6.1: Parameters obtained by fitting equation (6.1) to the measured bias stress-induced ΔV_T in Figure 6.3 (a).

V _G (V)	$ V_{T}(\infty)-V_{T}(0) (V)$	τ (s)	β
10	3.27	$8.47 \ge 10^3$	0.36
40	12.66	9.08 x 10 ³	0.51



Figure 6.3: Changes in threshold voltage shift and mobility as function of gate bias stress time for gate bias stress (a) un-passivated threshold voltage shift at 10V, 40V (b) un-passivated saturation mobility shift at 10V, 40V, (c) passivated threshold voltage shift at 10V, 40V and (d) passivated saturation mobility shift at 10V, 40V.

Shown in Figure 6.3 are the changes in threshold voltage shift and mobility as function of gate bias stress time for gate bias stress (a) un-passivated threshold voltage shift at 10V, 40V (b) un-passivated saturation mobility shift at 10V, 40V, (c) passivated threshold voltage shift at 10V, 40V and (d) passivated saturation mobility shift at 10V, 40V. The ΔV_T is highly affected by the stress for un-passivated devices which is caused by electrons trapping at the interface between semiconductor/insulator. However, in the case of passivated TFTs, ΔV_T is only slightly affected by the positive gate bias stress due

to small electrons trapping at the interface between semiconductor/insulator. This is due to the hydrophobic nature of the CYTOP passivation layer. The red line in Figure 6.3 (a), (c) indicates that the time dependence of ΔV_T can be described by the stretched-exponential function given by equation (6.1). The gate-voltage dependent saturation mobility decreased for unpassivated TFTs as shown in Figure 6.3 (b), however for passivated initially decreased and then remained unchanged by positive voltage stressing as shown in Figure 6.3 (d).

6.4 Recovery Procedure

First gate voltages (V_G) ranging from 5V to 40 V were applied with the drain voltage V_D=0V. All stress conditions were applied for 11000s in the dark under ambient (uncontrolled) laboratory conditions, typically T=15 °C-25 °C and RH=40%–60%. After completing a bias stress test, selected devices were subjected to a recovery procedure. The recovery was interrupted at fixed times to record the transfer characteristics of the transistor with a drain bias of V_D =40V by sweeping the gate bias from 0 to 40V.



Figure 6.4: (a) Transfer characteristics recorded before, after positive bias stress and recovery at 180s with light, (b) time dependence of ΔV_T , recovery under white light illumination and (c) time dependence of ΔV_T , recovery under dark. The device was subjected to PGBS of V_G = 40V for 11000s.

Before proceeding to the next stress condition, V_T was reset to its initial value by exposing the device to white light for about 180s using white LED (3.2 V White LED 5mm through hole, Nichia NSPW500DS) purchaged fron RS components. Figure 6.4 (a) shows the transfer characteristics recorded before and after positive bias stress and full recovery after illumination 180s with white light, the time dependence of ΔV_T under white, Figure 6.4 (b) and Figure 6.4 (c) under devices short-circuited condition in the dark. The device was subjected to bias stress V_G = 40 V, V_D = 0 V in the dark for 11000 s. The recovery process was performed in the dark and under white light illumination with V_G = V_D = 0 V. Shown in Figure 6.4 (b) full recovery time 180s under white light and (c) 21 min for full recovery under dark when devices were left short-circuited in the dark.

It can be clearly observed in Figure 6.4(a), that the transfer curves shift back towards the one obtained before stressing. Changes in the threshold voltage (ΔV_T) over time both during stress and under recovery are shown in Figure 6.4 (b), (c). It is observed, that the full recovery of ΔV_T in the dark took 21 min. However, with white light exposure for a few seconds, ΔV_T fell to zero and the transfer curve was restored within ~ 180 s to its original value, thus providing a rapid method of re-setting the device ahead of further measurements. Such a fast recovery property is technologically relevant and could be utilized in practical applications such as in a memory device.



Figure 6.5: Change in threshold voltage, (a) un-passivated and passivated TFTs threshold voltage shift at 5V, 20V, 40V (b) un-passivated TFTs threshold voltage shift at 5V, 20V, 40V with white light (WL) and without white light as a function of gate bias stress for 11000s time.

6.5 Effect of CYTOP Passivation, White light (WL) and Positive Gate Bias stress

The influence of CYTOP passivations and white light illumination on the stability of IGZO thin film transistor is investigated in this section. Figure 6.5 shows the change in threshold voltage, change in threshold voltage, (a) un-passivated and passivated TFTs threshold voltage shift at 5V, 20V, 40V (b) un-passivated TFTs threshold voltage shift at 5V, 20V, 40V with white light (WL) and without white light as a function of gate bias stress for 11000s time.



Figure 6.6: The Change in threshold voltage, (a) passivated TFTs threshold voltage shift at 5V, 20V, 40V with white light (WL) and without WL, (b) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 1000s time, (c) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 5000s time, (d) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 5000s time, (d) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 11000s time.

Figure 6.6 shows the change in threshold voltage, (a) passivated TFTs threshold voltage shift at 5V, 20V, 40V with white light (WL) and without WL , (b) un-passivated,

un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 1000s time, (c) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 5000s time, (d) un-passivated, un-passivated with white light, passivated and passivated with white light threshold voltage shift as a function of gate bias stress for 11000s time.

It can be clearly seen from Figure 6.5 (a) that ΔV_T shift the CYTOP-passivated device exhibited relatively good stability ($\Delta V_T = 1V$) under positive gate bias stress of 5V while the unpassivated TFTs showed a 4.1V ΔV_T shift, respectively. The effect of white light on un-passivated TFTs can be seen from Figure 6.5 (b). Under long positive gate bias stress, the device illuminated by white light exhibits smaller positive threshold voltage shift than the device stressed under dark as one can observe from Figure 5.5 (b), 6.6 (a). The reduced threshold voltage shift under WL illumination can be explained by a competition between the bias-induced interface carrier trapping effect and photon-induced carrier de-trapping effect [221].

Overall, it is observed that CYTOP passivation reduces the traps and removes any moisture at the CYTOP/IGZO interface. However, WL illumination on the device reduces the threshold voltage shift due to photon-induced carrier de-trapping at the dielectric/active layer interface.

6.6 CV Measurements Before and After Bias Stressing

In the study of metal-oxide-semiconductor (MOS) structures, the capacitance voltage (CV) analysis is one of the simplest approaches for probing defects. To further clarify the instability effect of PGBS on passivated and unpassivated IGZO TFTs, the gate to source capacitance (C_{GS}) of IGZO TFTs was investigated. The C-V measurement of IGZO TFT was used to prove the positive effect of the proposed CYTOP passivation layer on reducing the traps at the interface. Whilst this does not quantify the number of trap states (as these will start to decrease as the measurement begins), trends in the data can be identified. The difference of capacitance between unpassivated and passivated TFTs as shown in Figure 6.8 is due to low interfacial traps in passivated ones as compared to unpassivated.

The low frequency TFT CV measurements were done at room temperature before and after PGBS using a Solatron SI 1255 HF frequency response analyser. The DC and the small-signal AC voltages were applied at the gate electrode, while changes to the stored charges due to the AC signal were detected through the shorted source and drain electrodes. A low frequency was selected to allow the charging current to arrive from the source and drain regions. At higher frequencies, the sub-gap states could potentially be filled in time, and the CV curves would show no distinct accumulation and depletion regions. Shown in Figure 6.7 is the variation in magnitude of capacitance, C_{GS}, with respect to applied gate voltage for unpassivated as well as passivated devices before the application of 40V positive gate bias voltage stress (PGBS). In the C_{GS}–V_G measurement, capacitance measurement terminal high is applied to the gate electrode and the source electrode is connected to capacitance measurement low terminal. After 11000 s PGBS stress, V_T variation with no degradation of subthreshold slope (SS) is observed, which is similar to the I_D–V_G transfer curves as shown in Figure 6.2. Shown in Figure 6.7 ΔV_{mg} is the shift of mid-gap voltage before and after PGBS of 40V for 11000s. It can be noted that this shift for CYTOP passivated is very small (1.5V) due to fewer trap states as compared to unpassivated (5.5V) TFTs for PGBS of 40V after 11000s.



Figure 6.7: C-V curves for sputtered IGZO TFT before and after PGBS. ΔV_{mg} is the shift of mid-gap voltage before and after PGBS of 40V for 11000s.

6.7 Density of States (DOS) Analysis

In this section, Density of States (DOS) analysis is used to understand the difference between unpassivated and passivated IGZO TFT devices under PGBS. In the previous section 6.3, it was suggested that the main effect of PGBS is simply to cause a threshold voltage shift, ΔV_T , arising from electron trapping in IGZO film or at SiO₂-IGZO interface. To confirm this, the DoS of the semiconductor was extracted to identify any evidence of
charge trapping within the IGZO bandgap. The DoS spectrum of IGZO was extracted from the transfer characteristics by applying the Grünewald method as outlined in Chapter 3.

Firstly, the relationship between the gate-dependent insulator/semiconductor interface potential, V₀, and the effective forward voltage, V_F = (V_G-V_{ON}) was obtained from equation (3.4) and plotted as shown in Figures 6.8 (a) and 6.8 (b). It can be observed that as stress time increases, plots of V₀ rise rapidly with V_F. Then the DoS, N(E) can be obtained with equations (3.5) and (3.6) explained in chapter 3 and plotted as a function of energy relative to the Fermi level, E_F , (V₀ = E - E_F) as shown in Figures 6.8 (c) and 6.8 (d).



Figure 6.8: Plots of V₀ versus V_F showing the effect of zero drain bias, V_D = 0 V and gate bias stresses of (a) V_G = 40 V unpassivated and (b) V_G = 40 V passivated. Density of states for zero drain bias, V_D = 0 V and gate bias stress of (c) V_G = 40 V unpassivated and (d) V_G = 40 V passivated.

From the plot of gate voltage dependent mobility at initial condition, it is assumed that the effective mobility becomes constant at $\sim 10 \text{ cm}^2/\text{Vs}$ (Figure 6.9 (a). This coincides

with a transport band edge (Figure 6.9 (b)) or *Ec*. Upon evaluating the relevant energy, the DoS may be replotted as a function of (*E-Ec*) as shown in Figures 6.10 (a) and (b). Under all bias stress conditions, the initial DoS distribution decreases from 1×10^{21} cm⁻³eV⁻¹ to 1×10^{18} cm⁻³eV⁻¹). As can be seen that the effect of gate bias stress is minimal with the plots obtained before and after bias stress coalescing to a single curve.



Figure 6.9: Constructions used to estimate the onset of mobility edge. The voltage, V_G , at which the tangent departs the experimental curve is noted and corrected for V_{ON} to give V_F , i.e $V_F = V_G - V_{ON} = 28 - (2) = 30 \text{ V}$.



Figure 6.10: Density of states plotted as a function of E-E_C for a zero-drain voltage, following gate bias stress with V_G of (a) 40 V unpassivated and (b) 40 V passivated.

The DoS distribution was fitted with the single exponential function of a localized states DoS according to:

$$N(E) \approx \frac{N_t}{E_t} \exp\left(-\frac{E}{E_t}\right)$$
 6.2

where N_t is the total trap density while E_t is the characteristic energy decay of the distributions. After stressing passivated devices for 11000 s under bias stress V_G = 40 V, V_D = 0 V, fitting parameters related to the shallow states ≤ 0.1 eV, is N_t $\sim 2.45 \times 10^{22}$ cm⁻³ with E_t ~ 43.8 meV. However, for unpassivated devices with same stress level, the total trap density increases slightly to N_t $\sim 3.02 \times 10^{22}$ cm⁻³ with E_t ~ 40.4 meV. This supports the view that shift in threshold voltage is due to interfacial traps existence and not from the formation of traps in the bandgap of IGZO

6.8 Low Voltage Flexible TFTs

IGZO thin-film transistors gated with E-beam processed thin Al₂O₃ have been fabricated on a plastic substrate. One method to achieve a low operating voltage is to decrease the thickness of the gate dielectric layer [222]. However, the gate leakage current can become too large if a conventional gate dielectric layer, e.g., SiO₂ is used. Otherwise, high dielectric constant (high-k) materials, such as HfO₂, Al₂O₃ and Ta₂O₅, have been used as a dielectric layer since they can provide a large specific gate capacitance without dramatically increasing the gate leakage current. Among the different types of high-k dielectric materials, Al₂O₃ is used because of its advantages such as low cost, low deposition temperature, low leakage current density and good compatibility with oxide semiconductors [223]. Shown in Figure 6.11 (a) a schematic cross-section of IGZO TFT with various device layers, (b) fabricated IGZO TFT device on Kapton film showing the bendability of devices.

Table 6.2: Surface roughness parameters of the dielectric and active layer.

Sample	R _q (nm)	R _a (nm)
Al ₂ O ₃ on Kapton	12.4	9.8
IGZO on AL ₂ O ₃	5.1	4.1

Atomic force microscopy (AFM) was used to study the surface roughness of the IGZO and Al_2O_3 films on Kapton substrates. Shown in Figure 6.11 (c) is the atomic force microscopic (AFM) image of the sputtered IGZO active layer. The important surface

roughness parameters values like RMS roughness (R_q), and average roughness (R_a) are shown in Table 6.2. The surface roughness of the dielectric and active layer surface affects the performance parameters and it should be low as possible. Roughness values were high, but devices were working after repeated measurements.



Figure 6.11: (a) A schematic cross-section of IGZO TFT biasing, (b) Fabricated IGZO TFT device on Kapton showing the flexibility and (c) AFM image of sputtered IGZO active layer.



Figure 6.12: Dielectric constant and oxide capacitance (log scale) of E-beam evaporated Al₂O₃ versus frequency.

The dielectric performance parameters of the Al₂O₃ films were measured in MIM capacitor configuration. Shown in Figure 6.12 is the dielectric constant (k) and the oxide

capacitance versus frequency for a manufactured MIM capacitor. In the inset of Figure 6.13 is shown a schematic of the MIM capacitor. The dielectric constant calculated is found to be around 8.52 and this value is closely matching the value of Al_2O_3 dielectric constant (8.32) using E-beam evaporation [224]. As shown in Figure 6.12 the dielectric constant does not change in the frequency range from 2KHz to 100 kHz and start increasing at higher frequencies. It can be clearly observed from Figure 6.13 that oxide capacitance is constant in the frequency range of 500Hz to 250KHz.

The TFT channel length (L) was 400 μ m and width (W) was 6300 μ m. The oxide capacitance density value in a flat band was 5.1 Fm⁻². The various performance parameters like mobility, threshold voltage and on/off current ratio were extracted from transfer characteristics and are tabulated in Table 6.3.

Table 6.3: Before bending the performance parameters of flexible IGZO TFTs at different bias voltages.

Operating voltage V_D (V)	V _T (V)	on-off ratio	Mobility cm ² V ⁻¹ s ⁻¹
0.5	0.07	0.92E+02	12.29
1	0.16	1.01E+03	28.11
2	0.18	0.89E+04	31.72

The fabricated flexible TFTs were characterized on curved holder, as shown in Figure 6.11 (b), where no considerable performance degradation was observed for curvature radii up to 15mm, as shown in Table 6.4. This result is promising for the use of the demonstrated TFTs in flexible electronics applications and combined with the ultra-low operation voltage (0.5V) these TFTs are suitable for battery-powered applications.

Table 6.4: After bending the performance parameters of flexible IGZO TFTs at different bias voltages.

Operating voltage V_D (v)	V _T (v)	ON-OFF ratio	Mobility cm ² V ⁻¹ s ⁻¹
0.5	0.08	0.90E+02	12.27
1	0.17	1.03E+03	28.09
2	0.19	0.88E+04	31.71

Typical transfer and output IV characteristics of IGZO TFTs measured under dark are shown in Figure 6.13. The leakage current is relatively high for these devices which do need to be improved in future devices. The leakage current in these devices can be improved by using bi-layer/composite organic-inorganic dielectric.



Figure 6.13: Transfer characteristics (a), (b) and (c) and output characteristics (d), (e) and (f) characteristics of IGZO-TFT with operating voltages: $V_D = 0.5 V$, 1V and 2V respectively.

Shown in Figure 6.14 the transfer characteristics of flexible IGZO-TFT before and after bending with operating voltage, V_D of 1V. It can be clearly observed from transfer characteristics that before and after bending there is not considerable degradation in the performance of flexible IGZO TFTs. The manufactured devices show a low operating voltage (V_D) of 1 V, a high drain current on-off ratio >10³, threshold voltage V_T of 0.16V,

saturation mobility μ_{sat} of 28 cm²V⁻¹s⁻¹ and a low subthreshold swing <200 mV/decade. The effects of bending on the gate dielectric in terms of important performance parameters have been studied and it is shown that the devices maintain their high performance even when flexed to a curvature radius of 15 mm. As a result, such devices possess a great potential for low-power, flexible electronics future applications like flexible displays.



Figure 6.14: The transfer characteristics of flexible IGZO-TFT before and after bending with operating voltage, $V_G = V_D$ of 1V.



Figure 6.15: (a) Image of manufactured IGZO TFT on ITO substrate, (b) Image of manufactured IGZO TFT on flexible AgNW electrodes.

6.9 Transparent IGZO TFTs on AgNW and ITO Electrodes

In this section transparent IGZO TFTs on flexible AgNWs TCEs are presented and their electrical and optical properties are compared to ITO based devices. AgNW TCEs were manufactured and post processed by using the same process as described in Chapter 4.

Shown in figure 6.15 (a) the image of manufactured IGZO TFT on an ITO substrate and figure 6.15 (b) the image of manufactured IGZO TFT on flexible AgNW electrodes. The E-beam evaporated Al_2O_3 layer (168nm) was used as the gate dielectric. The channel length of these devices was 90 μ m and width 5mm.

Operating	ITO gated				AgNW gated			
voltage V _D (V)	Performance Parameters				Performance Parameters			
	μs	VT	Ion/Ioff	SS	μs	VT	Ion/Ioff	SS
	[cm ² V ⁻¹ s ⁻¹]	[V]	[ratio]	mV/dec	[cm ² V ⁻¹ s ⁻¹]	[V]	ratio	mV/dec
2	5.04	0.57	1.04E+03	310	1.72	0.48	1.52E+03	185
3	10.50	0.77	1.44E+04	388	2.3	0.91	1.34E+04	188

Table 6.5: Performance parameters of AgNW gated and ITO gated IGZO TFTs.

Both ITO, as well as AgNW TCE, gated IGZO TFTs were working at low operating voltage as low as 2V. Performance parameters of AgNW gated and ITO gated IGZO TFTs are compared in Table 6.5. The devices show high mobility, high on-off ratio, low threshold voltage and small subthreshold slope as can be observed from Table 6.5. K. S. Kim et al, reported IGZO TFT on ITO electrode with mobility of 7.86 cm²V⁻¹s⁻¹, V_T = -0.23 V and SS =340 mV/dec [225].



Figure 6.16: Transmittance as a function of wavelength for (a) ITO gated IGZO TFT and (b) AgNW TCE gated IGZO TFT.

Shown in Figure 6.16 the transmittance as a function of wavelength for (a) ITO-IGZO film and (b) AgNWs-IGZO film. The transmittance at 550 nm of 91.2% (R_{sh} of 60 Ω sq⁻¹) for ITO-IGZO film, however for AgNWs-IGZO film 89.6% (R_{sh} of 12 Ω sq⁻¹) was measured.

Shown in Figure 6.17 the transfer and output characteristics of IGZO transparent TFTs gated with ITO and AgNW conducting electrodes. It can be clearly observed from the transfer characteristics that ITO gated TFTs shows higher off state current of the order of one decade as compared to AgNW film gated TFTs. However, ON state current is higher of the order of one decade for ITO gated TFTs as compared to AgNW film gated TFTs.



Figure 6.17: Transfer characteristics of ITO gated (black) and AgNWs gated IGZO-TFTs.

6.10 Summary

Positive gate bias-induced threshold voltage instability phenomenon of IGZO TFTs is investigated. A combination of device data and DOS analysis is explained in order to observe the difference between passivated and unpassivated TFTs. The influence of CYTOP passivation, white light illumination on the stability of IGZO thin film transistor is also explained. The work reports IGZO TFTs which have been fabricated using E-beam deposited high-k dielectric (Al₂O₃) onto a plastic substrate. The devices show a low operating voltage (V_D) of 1 V, a high current on-off ratio >10³, threshold voltage V_T of 0.16V, mobility of 28 cm²V⁻¹s⁻¹ and a low subthreshold swing <200 mV/decade. The performance shows great potential for low-power, flexible electronics. The performance of ITO gated and AgNW film gated IGZO TFTs is explained and results are compared.

Chapter 7. Conclusions and Further Work

7.1 Conclusions

The work reported in this thesis is on the development of flexible and transparent conducting electrodes (TCEs) such as silver nanowire (AgNW) TCEs, metal oxide thin film transistors (TFTs) with the aim of improving their stability and performance in real world conditions. This is important as for devices to become commercially viable they must make the transition from laboratory to large area electronics products that can be of real use. In-Depth literature of TCEs and metal oxide TFTs was given, where fundamental properties of AgNWs, metal oxides semiconductors and the basic operation principle of TFTs are explained. TCEs based on various materials, applications and their post processing methods used in literature are discussed. State-of-the-art performances for AgNWs TCEs and metal oxide TFTs were reviewed. Process optimization techniques, Taguchi methods and Plackett-Burman design were also discussed.

At the onset of this Ph.D., an indoor thermal embossing and NIL systems were setups for post processing of AgNW TCEs. This enables AgNW TCEs post processing to improve the electrical and surface properties after thermal embossing. The post processing relied upon three sequential steps; thermal embossing, infrared sintering and a nitrogen plasma treatment. This process leads to the demonstration of AgNWs films with a sheet resistance of 2.5 Ω sq⁻¹and 85% transmittance with world leading FOM = 933, reported FOM is 487.9 using expensive DC magnetron sputtering (Vacuum process). However, our manufacture of AgNW TCEs was based on low-cost spray coating in ambient conditions.

To evaluate the long-term stability of the post processing upon the AgNWs stability, a bespoke accelerated lifetime testing (ALT) system was constructed. For this new circuit, ALT was designed and implemented on the printed circuit board with a switch matrix composed of relays, diodes and BJTs. This allowed high current biasing of the AgNW electrodes and repeated resistance calculations after a regular period, to quantify the degradation of AgNW TCEs in different accelerated real-world environmental conditions. X-ray photoelectron spectroscopy was used to understand the improvement in long-term stability. The thermally embossed and aged samples show low levels of sulphur indicating why these samples remain stable for a long time. A greater proportion of the AgNWs are submerged into the underlying PMMA buffer layer, hence

AgNWs remain less susceptible to environmental ageing. The overall properties of the developed AgNW electrodes show enhanced properties over the incumbent technology; ITO.

The optimisation of high-k dielectric manufacture and ZnO deposition process using a combinational, DOE approach is reported in Chapter 5. Correlation between the manufacturing process parameters and the device characteristics was investigated. An important observation from the results is that the manufacturing parameters of the dielectric layer of the TFTs have a strong influence on the TFT device performance. This procedure can be applied for the optimization of other devices in printed/flexible electronics to improve the fabrication process and to speed up the development of new low-cost products.

Wide bandgap ($E_g = 3.37 \text{ eV}$) ZnO material is used for the manufacturing of low voltage TFTs and these TFTs are well suited to UV sensing applications. It was observed that the electrical characteristics of both ZnO TFT sprayed and sputtered are altered after exposure to UV light. However, the magnitudes of these changes are larger in sprayed ZnO TFT than sputtered as their surface structure has a greater affinity with the absorption and desorption of oxygen molecules on its surface. From the perspective of UV sensing, ZnO TFT on-current in the saturation regime appears to be the most suitable performance parameter for examining differences in UV exposure.

The effect of the substrate temperature (during ZnO deposition by spray pyrolysis and annealing temperature after sputtering) on the structural, optical, and electrical transport properties of ZnO TFTs has been investigated. An increase of the substrate deposition/annealing temperature to 450 °C resulted in ZnO films with enlarged crystal size. Therefore, TFTs based on ZnO films grown at substrate temperatures in the range of 300 to 450 °C exhibit the highest electron mobility. The present results demonstrate the possibility of depositing high optical and electronic quality ZnO films onto large area substrates using spray pyrolysis, which is attractive for the electronics industry, since the pyrolysis spray technique is inexpensive, it does not need vacuum steps and is easily applied to large scale electronics.

Understanding the origin of electrical instability in metal oxide TFTs over long periods of time is also essential to realize high performance circuits. Therefore, a positive gate bias-induced threshold voltage instability phenomenon of IGZO TFTs was investigated. The influence of CYTOP passivation and white light illumination on the stability of IGZO thin film transistors is also reported. A density of states analysis was performed in order to observe the difference between passivated and unpassivated TFTs; demonstrates the trap states at the semiconductor/dielectric interface.

Furthermore, the performance of the flexible IGZO TFTs which have been fabricated using E-beam deposited high-k dielectric (Al₂O₃) onto a plastic substrate, demonstrating low operating voltage (V_D) of 1 V, a high current on/off ratio >10³, threshold voltage V_T 0.16V, mobility of 28 cm²V⁻¹s⁻¹ and a low subthreshold swing <200 mV/decade are reported. However, the gate leakage current was high and this needs significant improvements. The performance of ITO gated and AgNW film gated IGZO TFTs is reported and results are compared.

Flexible electronics applications are expected to grow substantially over the next 10 years. These types of devices utilise new materials and production processes, which have not been previously used in an industrial context. The need to understand and optimise performance in a rapid manner has become imperative. Given the complexity and challenges of new flexible electronics components, the 'multivariate' approach reported in Chapter 5 could be adopted more widely by the industry to improves the reliability and performances of such devices.

7.2 Future Work

It was envisaged that we would fabricate a CMOS inverter using organic p-type and IGZO n-type devices but time imposed a limitation on completing this. Nevertheless, resistive mode inverters are reported. IGZO TFT based N-MOS resistive inverters were fabricated. Photolithography and lift-off process were used to pattern the source/drain metallization as shown in Figure 3.11 in chapter 3. The individual transistors with 5 to 100- μ m channel lengths were measured to have the highest saturation mobility of 17 cm²V⁻¹s⁻¹, the subthreshold swing of 388 mV/decade, and current on-off ratios in excess of 107. The basic design of a resistive load inverter is shown in Figure 7.1 (a). Here, IGZO TFT acts as the driver transistor and the load consists of a simple linear resistor R_L. The resistive load inverter voltage transfer characteristics were measured using a number of load resistances varying from 100 Ω to 1M Ω . Logic operations were satisfactorily demonstrated for a bias voltage of 5 to 25 V.



Figure 7.1: (a) Resistive load NMOS inverter biasing circuit, (b) IGZO TFTs on SiO_2/Si substrates with different channel lengths, (c) Transfer characteristics of IGZO TFTs with several channel lengths.

Digital inverter quality is often measured using the voltage transfer characteristics (VTC), which is a plot of output voltage as a function of the input voltage. Shown in Table 7.1, the important inverter parameters including noise margin, and operating logic levels obtained from VTC. The two corners of the VTC are very important, as the slope or gain (A) is –1 here and are termed as V_{IL} and V_{IH} points, shown in Figure 7.2 (b). The point at which $V_{in} = V_{out} = V_M$ is the transition point and called the inverter threshold voltage or switching threshold voltage. The noise that may get attached and vary these voltages levels. Noise tolerances for digital circuits are termed as Noise Margin NM (Low, High). Therefore, Noise Margin NM_L = $V_{IL} - V_{OL}$ and $NM_H = V_{OH} - V_{IH}$. The NM should be wide for stable operation of the inverter and high noise immunity.



Figure 7.2: (a) Experimentally measured voltage transfer characteristics (VTC) of NMOS resistive inverter with different loads from 100 K Ω to 1 M Ω , (b) VTC of NMOS resistive inverter showing important voltages of the inverter.

Important	Resistive loads								
voltages of	100 KΩ	110 KΩ	150 KΩ	180 KΩ	0.5ΜΩ	1 MΩ			
Inverter									
(V)									
V _{OH}	24.94	24.82	25	24.51	23.31	22.07			
V _{OL}	3.31	3.12	2.68	2.37	1.42	1.22			
V _{IL}	2.14	2.08	2.2	1.95	1.39	0.82			
V _{IH}	13.87	13.43	12.81	12.12	8.77	7.24			
V _M	10.71	10.42	10.02	9.48	7.26	5.73			
NML	-1.17	-1.04	-0.48	-0.42	-0.03	-0.4			
NM _H	11.07	11.39	12.19	12.39	14.54	14.83			

Table 7.1: Important inverter parameters obtained from the VTC of resistive load inverter using IGZO TFT.

Shown in Figure 7.3 is experimentally measured NMOS resistive inverter response to square wave input signal, (a) 5V supply voltage, $1M\Omega$ resistance, and 1000pF capacitance (b) 5V supply voltage, $2M\Omega$ resistance, and 1000pF capacitance,(c) 5V supply voltage, $5M\Omega$ resistance, and 1000pF capacitance, (d) 5V supply voltage, $10M\Omega$ resistance, and 1000pF capacitance, and 1000pF capacitance, and 1000pF capacitance, and 1000pF capacitance and 1000pF capacitance and 1000pF capacitance and 1000pF capacitance.



Figure 7.3: Experimentally measured NMOS resistive inverter response to square wave input signal, (a) 5V supply voltage, $1M\Omega$ resistance, and 1000pF capacitance (b) 5V supply voltage, $2M\Omega$ resistance, and 1000pF capacitance,(c) 5V supply voltage, $5M\Omega$ resistance, and 1000pF capacitance, (d) 5V supply voltage, $10M\Omega$ resistance, and 1000pF capacitance, for 1000pF capacitance, and 1000pF capacitance.



Figure 7.4: (a) Experimentally measured voltage transfer characteristics (VTC) of flexible NMOS resistive inverter (b) Experimentally measured flexible NMOS resistive inverter response to the square wave input signal, 5V supply voltage, $1M\Omega$ resistance, and 1000pF capacitance.

Rise time (t_r) is calculated, during the transition, when output switches from 10% to 90% of the maximum value and fall time (t_f) when output switches from 90% to 10% of the maximum value. According to our testing oscilloscope accuracy and the constructed circuit using 1M Ω resistance, 1000pF capacitance, the best rise, and fall times were extracted to be ~7.5 ms (Figure 7.3 (a)).

Similarly, using low voltage flexible IGZO TFTs described in earlier section 6.7, VTC and transient response of low voltage resistive inverter was measured using $1M\Omega$, 1000pF load and shown in Figure 7.4.

Flexible metal oxide TFTs are close to becoming a viable commercial technology, as the obstacles of stability and cost are being overcome. Metal oxides and organic dinaphtho[2,3-b:2',3'-f]thieno [3,2-b]thiophene (DNTT) both are stable materials, flexible CMOS circuits based on these materials are of importance. As active layer materials become more stable and achieve higher mobilities, the importance of preventing the ingress of water and oxygen by the use of effective encapsulation to reduce interface traps come to the forefront.

To gain more understanding of the trapping mechanisms and finding the origin of the degradation, it would be necessary to investigate the chemical changes in the metal oxide TFTs using techniques such as charge accumulation (CAS) spectroscopy. Access to such equipment would allow changes due to the chemical interaction of the semiconductor with O₂ and H₂O to be identified and compared to the results obtained from electrical measurements.

New accelerated life test procedures, for example, time-varying step-stress tests are proposed for testing the stability of AgNW TCEs and metal oxide TFTs. In step-stress accelerated testing, the test devices are subjected to successively higher stress levels in predetermined periods, and thus follow a time-varying stress profile. The devices usually start at a lower stress level at a predetermined time, the stress is increased and the test continues. The test is terminated when all devices have failed or when a certain number of failures are observed. Step-stress testing can substantially reduce the reliability test's duration.

REFERENCES

- [1] S. Gupta, W. T. Navaraj, L. Lorenzelli, and R. Dahiya, "Ultra-thin chips for high-performance fl exible electronics," *npj Flex. Electron.*, no. January, 2018.
- [2] W. Gao, H. Ota, D. Kiriya, K. Takei, and A. Javey, "Flexible Electronics toward Wearable Sensing," *Acc. Chem. Res.*, vol. 52, pp. 523–533, 2019.
- [3] R. Dass, "Printed, Organic and Flexible Electronics 2020-2030: Forecasts, Technologies, Markets," 2019. Available: https://www.idtechex.com/en/researchreport/flexible-printed-and-organic-electronics-2020-2030-forecasts-playersand-opportunities/687.
- [4] A. Klein and K. Christoph, "Transparent Conducting Oxides for Photovoltaics: Manipulation of Fermi Level, Work Function and Energy Band Alignment," pp. 4892–4914, 2010.
- [5] O. Bierwagen, "Indium oxide a transparent, wide-band gap semiconductor for (opto) electronic applications," *Semicond. Sci. Technol.*, vol. 30, no. 2, p. 24001, 2015.
- [6] Z. Lou and G. Shen, "Flexible Photodetectors Based on 1D Inorganic Nanostructures," pp. 1–19, 2015.
- [7] A. Kaidarova *et al.*, "Wearable multifunctional printed graphene sensors," *npj Flex. Electron.*, no. February, pp. 1–10, 2019.
- [8] Qijin Huang, Karam Nashwan Al-Milaji and Hong Zhao, "Inkjet Printing of Silver Nanowires for Stretchable Heaters," ACS Appl. Nano Mater., vol. 1, pp. 4528–4536, 2018.
- [9] M. Lagrange, D. P. Langley, G. Giusti, C. Jiménez, and Y. Bréchet, "Optimization of silver nanowire-based transparent electrodes: effects of density, size and thermal annealing," *Nanoscale*, vol. 7, pp. 17410–17423, 2015.
- [10] Dinh-Phuc Tran, Hung-I Lu and Chih-Kuang Lin, "Conductive Characteristics of Indium Tin Oxide Thin Film on Polymeric Substrate under Long-Term Static Deformation," Coatings, 8, 212. 2018.
- [11] J. Yeon Kwon and J. Kyeong Jeong, "Recent progress in high performance and reliable n-type transition metal oxide-based thin film transistors," *Semicond. Sci. Technol.*, vol. 30, no. 2, p. 024002, 2015.
- [12] L. Zhang, W. Xiao, W. Wu, and B. Liu, "Research Progress on Flexible Oxide-Based Thin Film Transistors," Appl. Sci., 9, 773, 2019.
- [13] M. Grundmann *et al.*, "The 2016 oxide electronic materials and oxide interfaces roadmap," *J. Phys. D. Appl. Phys.*, p. 433001, 2016.
- [14] B. Wang, W. Huang, L. Chi, M. Al-hashimi, T. J. Marks, and A. Facchetti, "High k Gate Dielectrics for Emerging Flexible and Stretchable Electronics," *Chem. Rev.*, vol. 118, pp. 5690–5754, 2018.
- [15] R. Das, "Flexible & Foldable OLED Displays 2019-2029," 2019. Available: https://www.idtechex.com/en/research-report/flexible-and-foldable-oleddisplays-2019-2029/629.
- [16] M. Nakata and S. I. D. Member, "Development of flexible displays using backchannel-etched In – Sn – Zn – O thin-film transistors and air-stable inverted organic light-emitting diodes," pp. 3–11, 2016.

- [17] Luisa Petti, Niko Münzenrieder, Christian Vogt, Hendrik Faber, Lars Büthe, Giuseppe Cantarella, Francesca Bottacchi, Thomas D. Anthopoulos, and Gerhard Tröste, "Metal oxide semiconductor thin-film transistors for flexible electronics," Applied Physics Reviews, vol. 021303, no. April 2016, 2017.
- [18] J. Sheng, H. Jeong, K. Han, T. Hong, and J. Park, "Review of recent advances in flexible oxide semiconductor thin-film transistors," vol. 0316, 2017.
- [19] J. Sheng, D. Choi, S. Lee, J. Park, and J. Park, "Performance modulation of transparent ALD indium oxide films on flexible substrates: transition between metal-like conductor," pp. 7571–7576, 2016.
- [20] W. Muhammad *et al.*, "Flexible IGZO TFT SPICE Model and Design of Active Strain-Compensation Circuits for Bendable Active Matrix Arrays," vol. 39, no. 9, pp. 1314– 1317, 2018.
- [21] R. Singh, Eric Singh and Hari Singh Nalwa, "Inkjet printed nanomaterial based flexible radio frequency identification (RFID) tag sensors for the internet of nano things," pp. 48597–48630, 2020.
- [22] Q. Li, L. Zhang, X. Tao, and X. Ding, "Review of Flexible Temperature Sensing Networks for Wearable Physiological Monitoring," vol. 1601371, pp. 1–23, 2017.
- [23] Y. H. Lee, M. Jang, M. Y. Lee, O. Y. Kweon, and J. H. Oh, "Flexible Field-Effect Transistor-Type Sensors Based on Conjugated Molecules," *CHEMPR*, vol. 3, no. 5, pp. 724–763, 2017.
- [24] R. Martins, E. Fortunato, and P. Barquinha, "High-Gain Transimpedance Amplifier for Flexible Radiation Dosimetry Using InGaZnO TFTs," *IEEE J. Electron Devices Soc.*, vol. 6, no. May 2018, pp. 760–765, 2020.
- [25] N. Münzenrieder, Daniil Karnaushenko, Luisa Petti, Giuseppe Cantarella, Christian Vogt, Lars Büthe, Dmitriy D. Karnaushenko, Oliver G. Schmidt, Denys Makarov and Gerhard Tröster, "Entirely Flexible On-Site Conditioned Magnetic Sensorics," Adv. Electron. Mater. 2016, 2, 1600188.
- [26] H. Lai, Z. Pei, J. Jian, and B. Tzeng, "Alumina nanoparticle / polymer nanocomposite dielectric for flexible amorphous indium-gallium-zinc oxide thin film transistors on plastic substrate with superior stability," vol. 013503, no. 2009, pp. 1–5, 2016.
- [27] T. Transistor, H. Hsu, C. Chang, C. Cheng, and S. Chiou, "An Oxygen Gettering Scheme for Improving Device Mobility and Subthreshold Swing of InGaZnO-Based Thin-Film Transistor," *IEEE Trans. Nanotechnol.*, vol. 13, no. 5, pp. 933–938, 2014.
- [28] I. F. M. Tfts *et al.*, "Influence of Mechanical Bending on Flexible InGaZnO-Based Ferroelectric Memory TFTs," vol. 61, no. 4, pp. 1085–1092, 2014.
- [29] Hsiao-Hsuan Hsu, Chun-Yen Chang, Chun-Hu Cheng, Po-Chun Chen, Yu-Chien Chiu, Ping Chiou, and Chin-Pao Cheng, "High Mobility Field-Effect Thin Film Transistor Using Room-Temperature High- Gate Dielectrics," vol. 10, no. 10, pp. 875–881, 2014.
- [30] Jeong Eun Lee, Bhupendra K. Sharma, Seoung-Ki Lee, Haseok Jeon, Byung Hee Hong, Hoo-Jeong Lee, and Jong-Hyun Ahn, "Thermal stability of metal Ohmic contacts in indium gallium zinc oxide transistors using a graphene barrier layer," Appl. Phys. Lett., vol. 113112, no. March 2013, 2014.
- [31] J. S. Discrete, K. Nomura, H. Ohta, A. Takagi, and T. Kamiya, "Room-temperature

fabrication of transparent flexible thin-film transistors using amorphous oxide semiconductors," vol. 432, no. November, pp. 3383–3386, 2004.

- [32] R. Zhang, "Recent Progress on the Fabrication and Properties of Silver Nanowire-Based Transparent Electrodes," pp. 1–17, 2018.
- [33] D. Langley, D. Bellet, and J. Simonato, "Flexible transparent conductive materials based on silver nanowire networks : a review."
- [34] A. Kumar and C. Zhou, "The Race To Replace Tin-Doped Indium Oxide : Which Material Will Win ?," *ACS Nano*, vol. 4, no. 1, pp. 11–14, 2010.
- [35] S. Huang, Y. Liu, Y. Zhao, Z. Ren, and C. F. Guo, "Flexible Electronics : Stretchable Electrodes and Their Future," vol. 1805924, pp. 1–15, 2019.
- [36] P. Zhao, Q. Tang, X. Zhao, Y. Tong, and Y. Liu, "Journal of Colloid and Interface Science Highly stable and flexible transparent conductive polymer electrode patterns for large-scale organic transistors," *J. Colloid Interface Sci.*, vol. 520, pp. 58–63, 2018.
- [37] Y. S. Woo, "Transparent Conductive Electrodes Based on Graphene-Related Materials," no. Cvd, pp. 24–28, 2019.
- [38] E. J. López-naranjo, L. J. González-ortiz, L. M. Apátiga, E. M. Rivera-muñoz, and A. Manzano-ramírez, "Transparent Electrodes : A Review of the Use of Carbon-Based Nanomaterials," vol. 2016, 2016.
- [39] Y. Tang, H. Ruan, Z. Huang, and D. Shi, "Fabrication of high-quality copper nanowires fl exible transparent conductive electrodes with enhanced mechanical and chemical stability," 2018.
- [40] S. Sorel, P. E. Lyons, S. De, J. C. Dickerson, and J. N. Coleman, "The dependence of the optoelectrical properties of silver nanowire networks on nanowire length and diameter," vol. 185201.
- [41] S. Sorel, U. Khan, J. N. Coleman, S. Sorel, U. Khan, and J. N. Coleman, "Flexible, transparent dielectric capacitors with nanostructured electrodes Flexible, transparent dielectric capacitors with nanostructured electrodes," Appl. Phys. Lett. , vol. 103106, no. 2012, 2016.
- [42] Sukanta De, Thomas Higgins, Philip E Lyons and Evie M Doherty, "Silver Nanowire Networks as Flexible, Transparent, Conducting Films: Extremely High DC to Optical Conductivity Ratios," vol. 3, no. 7, pp. 1767–1774, 2009.
- [43] D. J. Finn, M. Lotya, and J. N. Coleman, "Inkjet Printing of Silver Nanowire Networks," ACS Appl. Mater. Interfaces 2015, 7, 9254–9261.
- [44] Franz Selzer, Nelli Weib, David Kneppe, Ludwig Bormann, Christoph Sachse, Nikolai Gaponik, Alexander Eychmüller, Karl Leoa and Lars Müller-Meskamp., "A spraycoating process for highly conductive silver nanowire networks as the transparent top-electrode for small molecule organic photovoltaics," Nanoscale, 2015, 7, 2777.
- [45] Dechan Angmo, Thomas R. Andersen, Janet J. Bentzen, Martin Helgesen, Roar R. Sondergaard, Mikkel Jorgensen, Jon E. Carle, Eva Bundgaard and Frederik C. Kreb, "Roll-to-Roll Printed Silver Nanowire Semitransparent Electrodes for Fully Ambient Solution-Processed Tandem Polymer Solar Cells," Adv. Funct. Mater.2015, 25, 4539–4547 pp. 4539–4547, 2015.
- [46] D. Langley, C. Celle, D. Bellet, and J. Simonato, "Flexible transparent conductive materials based on silver nanowire networks : a review," Nanotechnology vol. 24, 2013.

- [48] K. Zilberberg and T. Riedl, "Metal-nanostructures-a modern and powerful platform to create transparent electrodes for thin-film photovoltaics," *J. Mater. Chem. A*, vol. 4, no. 38, pp. 14481–14508, 2016.
- [49] T. Sannicolo, M. Lagrange, A. Cabos, C. Celle, J. P. Simonato, and D. Bellet, "Metallic Nanowire-Based Transparent Electrodes for Next Generation Flexible Devices: a Review," *Small*, vol. 12, no. 44, pp. 6052–6075, 2016.
- [50] F. Selzer *et al.*, "A spray-coating process for highly conductive silver nanowire networks as the transparent top-electrode for small molecule organic photovoltaics," *Nanoscale*, vol. 7, no. 6, pp. 2777–2783, 2015.
- [51] S. Sorel, U. Khan, and J. N. Coleman, "Flexible, transparent dielectric capacitors with nanostructured electrodes," *Appl. Phys. Lett.*, vol. 101, no. 10, 2012.
- [52] B. Wang, E. Lee, D. Lim, H. W. Kang, and Y. Oh, "Roll-to-roll slot die production of 300 mm large area silver nanowire mesh films for flexible transparent electrodes," pp. 7540–7546, 2017.
- [53] Linjie Li, Bo Zhang, Binghua Zou, Ruijie Xie, Tao Zhang, Sheng Li, Bing Zheng, Jiansheng Wu, Jiena Weng, Weina Zhang, Wei Huang and Fengwei Huo, "Fabrication of Flexible Transparent Electrode with Enhanced Conductivity from Hierarchical Metal Grids," ACS Appl. Mater. Interfaces, vol. 9, no. 45, pp. 39110–39115, 2017.
- [54] Jung-Yong Lee, Stephen T. Connor, Yi Cui and Peter Peumans., "Solution-Processed Metal Nanowire Mesh Transparent Electrodes," *Nano Lett.*, vol. 8, no. 2, pp. 689– 692, 2008.
- [55] Y. Jia, C. Chen, D. Jia, S. Li, S. Ji, and C. Ye, "Silver Nanowire Transparent Conductive Films with High Uniformity Fabricated via a Dynamic Heating Method," *ACS Appl. Mater. Interfaces*, vol. 8, no. 15, pp. 9865–9871, 2016.
- [56] Haiwei Du, Tao Wan, Bo Qu, Fuyang Cao, Qianru Lin, Nan Chen, Xi Lin and Dewei Chu, "Engineering Silver Nanowire Networks: From Transparent Electrodes to Resistive Switching Devices," ACS Appl. Mater. Interfaces, vol. 9, no. 24, pp. 20762– 20770, 2017.
- [57] Colin Preston, Zhiqiang Fang, Joseph Murray, Hongli Zhu, Jiaqi Dai, Jeremy N. Munday and Liangbing Hu, "Silver nanowire transparent conducting paper-based electrode with high optical haze," *J. Mater. Chem. C*, vol. 2, pp. 1248–1254, 2014.
- [58] Luis Jose Andres, Maria Fe Menendez, David Gomez, Ana Luisa Martínez, Noel Bristow, Jeffrey Paul Kettle, Armando Menéndez and Bernardino Ruiz, "Rapid synthesis of ultra-long silver nanowires for tailor-made transparent conductive electrodes: proof of concept in organic solar cells.," *Nanotechnology*, vol. 26, no. 26, p. 265201, 2015.
- [59] A. Kim, Y. Won, K. Woo, C. H. Kim, and J. Moon, "Highly transparent low resistance ZnO/Ag nanowire/ZnO composite electrode for thin film solar cells," ACS Nano, vol. 7, no. 2, pp. 1081–1091, 2013.
- [60] S. H. Jo, Y. K. Lee, J. W. Yang, W. G. Jung, and J. Y. Kim, "Carbon nanotube-based flexible transparent electrode films hybridized with self-assembling PEDOT," *Synth. Met.*, vol. 162, no. 13–14, pp. 1279–1284, 2012.

- [61] Nara Kim, Seyoung Kee, Seoung Ho Lee, Byoung Hoon Lee, Yung Ho Kahng, Yong-Ryun, Jo Bong-Joong Kim and Kwanghee Lee, "Highly conductive PEDOT:PSS nanofibrils induced by solution-processed crystallization," *Adv. Mater.*, vol. 26, no. 14, pp. 2268–2272, 2014.
- [62] Sukang Bae, Hyeongkeun Kim, Youngbin Lee, Xiangfan Xu, Jae-Sung Park, Yi Zheng and Sumio Iijima, "Roll-to-roll production of 30-inch graphene films for transparent electrodes," *Nat. Nanotechnol.*, vol. 5, no. 8, pp. 574–578, 2010.
- [63] M. Lagrange, D. P. Langley, G. Giusti, C. Jiménez, Y. Bréchet, and D. Bellet, "Optimization of silver nanowire-based transparent electrodes: Effects of density, size and thermal annealing," *Nanoscale*, vol. 7, no. 41, pp. 17410–17423, 2015.
- [64] S. Ye, A. R. Rathmell, Z. Chen, I. E. Stewart, and B. J. Wiley, "Metal Nanowire Networks : The Next Generation of Transparent Conductors," pp. 1–18, 2014.
- [65] H. Wang, A. Smirnov, K. Wang, H. Huang, and X. W. Sun, "Optical haze of randomly arranged silver nanowire transparent conductive films with wide range of nanowire diameters," vol. 035201, no. January, 2018.
- [66] K. Zilberberg and T. Riedl, "thin- fi lm photovoltaics," pp. 14481–14508, 2016.
- [67] C. Mayousse, C. Celle, A. Carella, and J. Simonato, "Synthesis and purification of long copper nanowires . Application to high performance flexible transparent electrodes with and without PEDOT : PSS," vol. 7, no. 3, pp. 315–324, 2014.
- [68] V. B. Nam and D. Lee, "Copper Nanowires and Their Applications for Flexible , Transparent Conducting Films : A Review," 2016.
- [69] C. Preston, Y. Xu, X. Han, J. N. Munday, and L. Hu, "Optical haze of transparent and conductive silver nanowire films," vol. 6, no. 7, pp. 461–468, 2013.
- [70] Y. Liu, J. Feng, X. Ou, H. Cui, M. Xu, and H. Sun, "Ultrasmooth, highly conductive and transparent PEDOT : PSS / silver nanowire composite electrode for fl exible organic light-emitting devices," Org. Electron., vol. 31, pp. 247–252, 2016.
- [71] G. Kang, K. Bae, M. Nam, D. Ko, K. Kim, and W. J. Padilla, "Environmental Science Broadband and ultrahigh optical haze thin films with self-aggregated alumina nanowire bundles for photovoltaic applications +," pp. 28–31, 2015.
- [72] Q. D. Ou, Y. Q. Li, and J. X. Tang, "Light manipulation in organic photovoltaics," *Adv. Sci.*, vol. 3, no. 7, pp. 1–25, 2016.
- [73] I. E. Stewart, M. Jun Kim, and B. J. Wiley, "Effect of morphology on the electrical resistivity of silver nanostructure films," ACS Appl. Mater. Interfaces, vol. 9, no. 2, pp. 1870–1879, 2017.
- [74] S. M. Bergin, Y. Chen, A. R. Rathmell, P. Charbonneau, Z. Li, and B. J. Wiley, "Nanoscale The effect of nanowire length and diameter on the properties of transparent, conducting nanowire films †," pp. 1996–2004, 2012.
- [75] S. Kim, S. Y. Kim, J. Kim, and J. H. Kim, "Highly reliable AgNW/PEDOT:PSS hybrid films: Efficient methods for enhancing transparency and lowering resistance and haziness," *J. Mater. Chem. C*, vol. 2, no. 28, pp. 5636–5643, 2014.
- [76] S. Wang, X. Zhang, and W. Zhao, "Flexible, Transparent, and Conductive Film Based on Random Networks of Ag Nanowires," vol. 2013, pp. 1–7, 2013.
- [77] Y. Shen, R. Yao, A. Pham, X. Nguyen, J. Gu, and X. Wang, "High-Performance Silver Nanowire Film on Flexible Substrate Prepared by Meyer-rod Coating High-

Performance Silver Nanowire Film on Flexible Substrate Prepared by Meyer-rod Coating," pp. 0–6.

- [78] Yujing Liu, Simon Zeder, Sen Lin, Romain Carron, Günter Grossmann, Sami Bolat, Shiro Nishiwaki, Frank Clemens, Thomas Graule, Ayodhya N. Tiwari, Hui Wu and Yaroslav E. Romanyuk. "Draw-spun, photonically annealed Ag fibers as alternative electrodes for flexible CIGS solar cells," Sci. Technol. Adv. Mater., vol. 20, no. 1, pp. 26–34, 2019.
- [79] Shudi Lu,Yang Sun, Kuankuan Ren, Kong Liu, Zhijie Wang , and Shengchun Qu, "Recent Development in ITO-free Flexible Polymer Solar Cells" 2018.
- [80] G. P. Panta and D. P. Subedi, "Electrical characterization of aluminum thin films measured by using four- point probe method," vol. 8, pp. 31–36, 2012.
- [81] Madaria AR, Kumar , Zhou C., "Large scale , highly conductive and patterned transparent films of silver nanowires on arbitrary substrates and their application in touch screens," Nanotechnology. 2011 Jun 17;22(24):245201.
- [82] Y. Wang, X. Yang, D. Du, Y. Zhao, and X. Zhang, "New Insights into Flexible Transparent Conductive Silver Nanowires Films," 2019.
- [83] S. B. Sepulveda-mora and S. G. Cloutier, "Figures of Merit for High-Performance Transparent Electrodes Using Dip-Coated Silver Nanowire Networks," Journal of Nanomaterials, 2012.
- Y. Wang, D. Du, X. Yang, X. Zhang, and Y. Zhao, "Optoelectronic and Electrothermal Properties of Transparent Conductive Silver Nanowires Films," *Nanomaterials*, vol. 9, no. 6, p. 904, 2019.
- [85] P. Meredith and A. Armin, "Scaling of next generation solution processed organic and perovskite solar cells," *Nat. Commun.*, pp. 8–11, 2018.
- [86] P. M. Martinez *et al.*, "Silver Nanowires on Carbon Nanotube Aerogel Sheets for Flexible, Transparent Electrodes," ACS Appl. Mater. Interfaces, vol. 11, pp. 32235– 32243, 2019.
- [87] C. F. Guo and Z. Ren, "Flexible transparent conductors based on metal nanowire networks," *Biochem. Pharmacol.*, vol. 18, no. 3, pp. 143–154, 2015.
- [88] T. Tokuno, M. Nogi, M. Karakawa, J. Jiu, T. T. Nge, and Y. Aso, "Fabrication of Silver Nanowire Transparent Electrodes at Room Temperature," *Nano Res.*, vol. 4, no. 12, 2011.
- [89] D. P. Langley, M. Lagrange, G. Giusti, C. Jiménez, Y. Bréchet, and N. D. Nguyen, "Metallic nanowire networks: e ff ects of thermal annealing on electrical resistance," pp. 13535–13543, 2014.
- [90] H. H. Khaligh and I. A. Goldthorpe, "Hot-rolling nanowire transparent electrodes for surface roughness minimization," pp. 20–22, 2014.
- [91] B. X. Zeng, Q. Zhang, R. Yu, and C. Lu, "COMMUNICATION A New Transparent Conductor : Silver Nanowire Film Buried at the Surface of a Transparent Polymer," vol. 350002, pp. 4484–4488, 2010.
- [92] M. xiang Jing, C. Han, M. Li, and X. qian Shen, "High performance of carbon nanotubes/silver nanowires-PET hybrid flexible transparent conductive films via facile pressing-transfer technique," *Nanoscale Res. Lett.*, vol. 9, no. 1, pp. 1–7, 2014.

- [93] S. K. Jiu J, Sugahara T, Nogi M, Araki T, Suganuma K, Uchida H, "High-intensity pulse light sintering of silver nanowire transparent films on polymer substrates: the effect of the thermal properties of substrates on the performance of silver films.," *Nanoscale*, vol. 5, no. 23, pp. 11820–8, 2013.
- [94] H. Hosseinzadeh Khaligh and I. A. Goldthorpe, "Hot-rolling nanowire transparent electrodes for surface roughness minimization," *Nanoscale Res. Lett.*, vol. 9, no. 1, pp. 1–5, 2014.
- [95] R. P. Xu, Y. Q. Li, and J. X. Tang, "Recent advances in flexible organic light-emitting diodes," *J. Mater. Chem. C*, vol. 4, no. 39, pp. 9116–9142, 2016.
- [96] Jun Li, Ye Tao, Shufen Chen, Huiying Li, Ping Chen, Meng-zhu Wei, Hu Wang, Kun Li, Marco Mazzeo and Yu Duan, "A flexible plasma-treated silver- nanowire electrode for organic light-emitting devices," *Sci. Rep.*, no. November, pp. 1–9, 2017.
- [97] J. West, "Sintering of Advanced Materials," Woodhead Publishing, 2010, pp. 275–288.
- [98] Mo L, Guo Z, Yang L, Zhang Q, Fang Y, Xin Z, Chen Z, Hu K, Han L, Li L. "Silver nanoparticles based ink with moderate sintering in flexible and printed electronics," *Int. J. Mol. Sci.*, vol. 20, no. 9, 2019.
- [99] D. J. Lee, Y. Oh, J. M. Hong, Y. W. Park, and B. K. Ju, "Light sintering of ultra-smooth and robust silver nanowire networks embedded in poly(vinyl-butyral) for flexible OLED," *Sci. Rep.*, vol. 8, no. 1, pp. 1–9, 2018.
- [100] C. Lee, Y. Oh, I. S. Yoon, S. H. Kim, B.-K. Ju, and J.-M. Hong, "Flash-induced nanowelding of silver nanowire networks for transparent stretchable electrochromic devices," *Sci. Rep.*, vol. 8, no. 1, pp. 1–10, 2018.
- [101] A. Jilani, M. S. Abdel-wahab, and A. H. Hammad, "Advance Deposition Techniques for Thin Film and Coating," in *Modern Technologies for Creating the Thin-film Systems and Coatings*, 2017.
- [102] C. W. Shih and A. Chin, "Remarkably High Mobility Thin-Film Transistor on Flexible Substrate by Novel Passivation Material," *Sci. Rep.*, no. December, pp. 1–8, 2017.
- [103] L. Petti *et al.*, "Metal Oxide Semiconductor Thin-Film Transistors for Flexible Electronics," *Appl. Phys. Rev.*, vol. 3, pp. 1–53, 2016.
- [104] T. S. Kang, T. Y. Kim, G. M. Lee, H. C. Sohn, and J. P. Hong, "Highly stable solutionprocessed ZnO thin film transistors prepared via a simple Al evaporation process," *J. Mater. Chem. C*, vol. 2, no. 8, p. 1390, 2014.
- [105] Y. Y. Lin, C. C. Hsu, M. H. Tseng, J. J. Shyue, and F. Y. Tsai, "Stable and High-Performance Flexible ZnO Thin-Film Transistors by Atomic Layer Deposition," ACS Appl. Mater. Interfaces, vol. 7, no. 40, pp. 22610–22617, 2015.
- [106] B. Y. Su, S. Y. Chu, and Y. Der Juang, "Improved electrical and thermal stability of solution-processed Li-doped ZnO thin-film transistors," *IEEE Trans. Electron Devices*, vol. 59, no. 3, pp. 700–704, 2012.
- [107] D. Salgueiro *et al.*, "Solution based zinc tin oxide TFTs: The dual role of the organic solvent," *J. Phys. D. Appl. Phys.*, vol. 50, no. 6, 2017.
- [108] D. P. Heineck, B. R. Mcfarlane, and J. F. Wager, "Zinc Tin Oxide Thin Film -Transistor Enhancement / Depletion Inverter," *Ieee Electron Device Lett.*, vol. 30, no. 5, pp. 514–516, 2009.

- [109] X. Zhang, J. P. Ndabakuranye, D. W. Kim, J. S. Choi, and J. Park, "Stability study of solution-processed zinc tin oxide thin-film transistors," *Electron. Mater. Lett.*, vol. 11, no. 6, pp. 964–972, 2015.
- [110] Y. G. Zhang, G. M. Huang, L. Duan, G. F. Dong, D. Q. Zhang, and Y. Qiu, "Full-solutionprocessed high mobility zinc-tin-oxide thin-film-transistors," *Sci. China Technol. Sci.*, vol. 59, no. 9, pp. 1407–1412, 2016.
- [111] Jozeph Park, Keun-Tae Oh, Dong-Hyun Kim, Hyun-Jun Jeong, Yun Chang Park, Hyun-Suk Kim, and Jin-Seong Park "High-Performance Zinc Tin Oxide Semiconductor Grown by Atmospheric-Pressure Mist-CVD and the Associated Thin-Film Transistor Properties," ACS Appl. Mater. Interfaces, vol. 9, no. 24, pp. 20656–20663, 2017.
- [112] Se Yeob Park, Ji Hun Song, Chang-Kyu Lee, Byeong Geun Son, Chul-Kyu Lee, Hyo Jin Kim, Rino Choi, Yu Jin Choi, Un Ki Kim, Cheol Seong Hwang, Hyeong Joon Kim, and Jae Kyeong Jeong, "Improvement in photo-bias stability of high-mobility indium zinc oxide thin-film transistors by oxygen high-pressure annealing," *IEEE Electron Device Lett.*, vol. 34, no. 7, pp. 894–896, 2013.
- [113] G. M. Li *et al.*, "Design of high speed gate driver employing IZO TFTs," *Displays*, vol. 39, pp. 93–99, 2015.
- [114] W. J. Wu *et al.*, "A highly stable biside gate driver integrated by IZO TFTs," *IEEE Trans. Electron Devices*, vol. 61, no. 9, pp. 3335–3338, 2014.
- [115] J. Sheng, E. J. Park, B. Shong, and J.-S. Park, "Atomic Layer Deposition of an Indium Gallium Oxide Thin Film for Thin-Film Transistor Applications," ACS Appl. Mater. Interfaces, vol. 9, no. 28, pp. 23934–23940, 2017.
- [116] J. H. Kang *et al.*, "Mobility enhancement in amorphous InGaZnO thin-film transistors by Ar plasma treatment," *Appl. Phys. Lett.*, vol. 102, no. 22, pp. 130–133, 2013.
- [117] K.-S. Kim, Y.-H. Hwang, I. Hwang, and W.-J. Cho, "Improved performance of solutionprocessed a-InGaZnO thin-film transistors due to Ar/O2 mixed-plasma treatment," *J. Korean Phys. Soc.*, vol. 65, no. 3, pp. 399–403, 2014.
- [118] S. J. Kim, S. Yoon, and H. J. Kim, "Review of solution-processed oxide thin-film transistors," *Jpn. J. Appl. Phys.*, vol. 53, no. 2 PART 2, p. 02BA02, 2014.
- [119] T. Kim, Y. Nam, J.-H. Hur, S.-H. K. Park, and S. Jeon, "Effect of hydrogen on dynamic charge transport in amorphous oxide thin film transistors," *Nanotechnology*, vol. 27, no. 32, p. 325203, 2016.
- [120] Shiben Hu, Kuankuan Lu, Honglong Ning, Zeke Zheng, Hongke Zhang, Zhiqiang Fang, Rihui Yao, Miao Xu, Lei Wang, Linfeng Lan, Junbiao Peng, and Xubing Lu., "High mobility amorphous indium-gallium-zinc-oxide thin-film transistor by aluminum oxide passivation layer," *IEEE Electron Device Lett.*, vol. 38, no. 7, pp. 879–882, 2017.
- [121] J. K. Jeong, J. H. Jeong, H. W. Yang, J. S. Park, Y. G. Mo, and H. D. Kim, "High performance thin film transistors with cosputtered amorphous indium gallium zinc oxide channel," *Appl. Phys. Lett.*, vol. 91, no. 11, pp. 1–4, 2007.
- [122] C.-T. Lee, Y.-H. Lin, M.-M. Chang, and H.-Y. Lee, "Investigation of Amorphous Indium Gallium Zinc Oxide Thin Film Transistors Grown byTriple-Targets Magnetron Co-Sputtering," J. Disp. Technol., vol. 10, no. 4, pp. 293–298, 2014.
- [123] C. Hsu, W. Tzou, C. Yang, and Y. Liou, "Investigation of the High Mobility IGZO Thin

Films by Using Co-Sputtering Method," pp. 2769–2781, 2015.

- [124] C. W. Shih, A. Chin, C. F. Lu, and W. F. Su, "Remarkably High Hole Mobility Metal-Oxide Thin-Film Transistors," *Sci. Rep.*, no. October 2017, pp. 1–6, 2018.
- [125] S. G. Danjumma, "Nickel Oxide (NiO) Devices and Applications : A Review," no. April, 2019.
- [126] Fukai Shan, Ao Liu, Huihui Zhu, Weijin Kong, Jingquan Liu, Byoungchul Shin, Elvira Fortunato, Rodrigo Martinse and Guoxia Liu., "High-mobility p-type NiOx thin-film transistors processed at low temperatures with Al2O3 high-k dielectric," pp. 9438–9444, 2016.
- [127] Woo-Jin Nam, Jong-Sik Shim, Hong-Jae Shin, Jin-Mog Kim, Woo-Suk Ha, Kwan-Ho Park, Hong-Gyu Kim, Bum-Sik Kim, Chang-Ho Oh, Byung-Chul Ahn, Bong-Chul Kim and Soo-Youle Cha, "55-inch OLED TV using InGaZnO TFTs with WRGB Pixel Design," Journal of the Society for Information Display, 27, 5, 2013.
- [128] Jiawei Zhang, Jia Yang, Yunpeng Li, Joshua Wilson, Xiaochen Ma, Qian Xin, and Aimin Song, "High Performance Complementary Circuits Based on p-SnO and n-IGZO Thin-Film Transistors," Materials (Basel). 10(3), 2017.
- [129] S. H. Lim, Y. Zhou, H. Su, and S. Ong, "A stable solution-processed polymer semiconductor with record high-mobility for printed transistors," pp. 1–9, 2012.
- [130] S. Kumagai *et al.*, "Solution-processed organic–inorganic hybrid CMOS inverter exhibiting a high gain reaching 890," *Org. Electron. physics, Mater. Appl.*, vol. 48, pp. 127–131, 2017.
- [131] Jin-gang Liu, Hong-jiang Ni, Zhen-he Wang, Shi-yong Yang and Wei-feng Zhou, "Colorless and Transparent high – Temperature-Resistant Polymer Optical Films – Current Status and Potential Applications in Optoelectronic Fabrications." http://dx.doi.org/10.5772/60432
- [132] Yong-Hui Zhang, Zeng-Xia Mei, Hui-Li Liang, and Xiao-Long Du, "Review of flexible and transparent thin-film transistors based on zinc oxide and related materials," Chin. Phys. B Vol. 26, No. 4, 2017.
- [133] S. Yeom *et al.*, "Silver Nanowire / Colorless- Polyimide Composite Electrode: Application in Flexible and Transparent Resistive Switching Memory," no. August 2016, pp. 1–9, 2017.
- [134] M. G. Faraj, K. Ibrahim, M. K. M. Ali, "PET as a plastic substrate for the flexible optoelectronic applications," Vol. 5, No. 8, , p. 879 882, 2011.
- [135] W. A. MacDonald, M. K. Looney, D. MacKerron, R. Eveson, R. Adam, K. Hashimoto and K. Rakos, "Latest advances in substrates for flexible electronics," Journal of the SID15,12,2007.
- [136] N. Chou, S. Yoo, and S. Kim, "Fabrication of stretchable and flexible electrodes based on pdms substrate," *IEEE 25th Int. Conf. Micro Electro Mech. Syst.*, pp. 247–250, 2012.
- [137] A. Zahid, B. Dai, R. Hong, and D. Zhang, "Optical properties study of silicone polymer PDMS substrate surfaces modified by plasma treatment Optical properties study of silicone polymer PDMS substrate surfaces modified by plasma treatment," Mater. Res. Express 4 105301 2017.
- [138] J. Chen and C. T. Liu, "Technology Advances in Flexible Displays and Substrates," *IEEE Access*, vol. 1, pp. 150–158, 2013.

- [139] T. Murata, T. Yanase, S. Miwa, and H. Yamazaki, "Ultra thin glass roll for flexible AMOLED display," pp. 1735–1737, 2011.
- [140] https://www.corning.com/emea/en/innovation/corning-emerginginnovations/corning-willow-glass.html.
- [141] https://www.corning.com/emea/en/about-us/news-events/newsreleases/2012/06/cn_en_news_center_news_releases_2012_2012060401.html.
- [142] T. Maindron, T. Jullien, A. André, T. Maindron, T. Jullien, and A. Andr, "Defect analysis in low temperature atomic layer deposited Al2O3 and physical vapor deposited SiO barrier films and combination of both to achieve high quality moisture barriers, Journal of Vacuum Science & Technology A," vol. 34, no. 3 May, 2016.
- [143] G. Congedo, A. Lamperti, O. Salicio, and S. Spiga, "Multi-layered Al2O3 / HfO2 / SiO 2 / Si3N4 / SiO2 thin dielectrics for charge trap memory applications for Charge Trap Memory Applications," no. November 2014, pp. 3–8, 2012.
- [144] H. Kim, H. Lee, and W. Maeng, "Applications of atomic layer deposition to nanofabrication and emerging nanodevices," *Thin Solid Films*, vol. 517, no. 8, pp. 2563–2580, 2009.
- [145] Y. Leterrier, "Durability of nanosized oxygen-barrier coatings on polymers," Progress in Materials Science vol. 48, pp. 1–55, 2003.
- [146] M. Mativenga, M. H. Choi, J. W. Choi, and J. Jang, "Transparent Flexible Circuits Based on Amorphous-Indium – Gallium – Zinc – Oxide Thin-Film Transistors," *IEEE Electron Device Lett.*, vol. 32, no. 2, pp. 170–172, 2011.
- [147] B. Wang, W. Huang, L. Chi, M. Al-hashimi, T. J. Marks, and A. Facchetti, "High k Gate Dielectrics for Emerging Flexible and Stretchable Electronics," *Chem. Rev.*, vol. 118 no. 1, 2018.
- [148] V. Pecunia, K. Banger, and H. Sirringhaus, "High-Performance Solution-Processed Amorphous- Oxide-Semiconductor TFTs with Organic Polymeric Gate Dielectrics," Adv. Electron. Mater. pp. 1–6, 2015.
- [149] C. B. Park, H. Il Na, S. S. Yoo, and K. Park, "Electromechanical properties of amorphous indium-gallium-zinc-oxide transistors structured with an island configuration on plastic," Appl. Phys. Express vol. 9,2016.
- [150] E. Oishi, N. Araki, T. Goto, H. Awano, and T. Takahashi, "The Effect of the Terminal Functional Groups on Fluoropolymer on Electrowetting Device Performance," *Technologies*, vol. 7, no. 3, p. 52, 2019.
- [151] W. L. Kalb, T. Mathis, S. Haas, A. F. Stassen, and B. Batlogg, "Organic small molecule field-effect transistors with Cytop[™] gate dielectric: Eliminating gate bias stress effects," *Appl. Phys. Lett.*, vol. 90, no. 9, 2007.
- [152] Seungbeom Choi, Jeong-Wan Jo, Jaeyoung Kim, Seungho Song, Jaekyun Kim, Sung Kyu Park and Yong-Hoon Kim., "Static and Dynamic Water Motion-Induced Instability in Oxide Thin-Film Transistors and its Suppression by using Low-k Fluoropolymer Passivation," ACS Appl. Mater. Interfaces, vol. 9, no. 31, pp. 26161– 26168, 2017.
- [153] S. H. Choi, J. H. Jang, J. J. Kim, and M. K. Han, "Low-temperature organic (CYTOP) passivation for improvement of electric characteristics and reliability in IGZO TFTs," *IEEE Electron Device Lett.*, vol. 33, no. 3, pp. 381–383, 2012.

- [154] K. Y. Shin, Y. J. Tak, W. G. Kim, S. Hong, and H. J. Kim, "Improvement of Electrical Characteristics and Stability of Amorphous Indium Gallium Zinc Oxide Thin Film Transistors Using Nitrocellulose Passivation Layer," ACS Appl. Mater. Interfaces, vol. 9, no. 15, pp. 13278–13285, 2017.
- [155] Z. Yong-Hui, M. Zeng-Xia, L. Hui-Li, and D. Xiao-Long, "Review of flexible and transparent thin-film transistors based on zinc oxide and related materials," *Chinese Phys. B.*
- [156] B. Hwang *et al.*, "Role of ultrathin Al 2 O 3 layer in organic / inorganic hybrid gate dielectrics for flexibility improvement of InGaZnO thin film transistors," *Org. Electron.*, vol. 15, no. 7, pp. 1458–1464, 2014.
- [157] H. Lai, Z. Pei, J. Jian, and B. Tzeng, "Alumina nanoparticle / polymer nanocomposite dielectric for flexible amorphous indium-gallium-zinc oxide thin film transistors on plastic substrate with superior stability," vol. 112123, no. 2006, pp. 1–5, 2016.
- [158] S. Faraji, T. Hashimoto, M. L. Turner, and L. A. Majewski, "Solution-processed nanocomposite dielectrics for low voltage operated OFETs," *Org. Electron.*, vol. 17, pp. 178–183, 2015.
- [159] R. S. Dahiya, S. Member, and S. Gennaro, "Bendable Ultra-Thin Chips on Flexible Foils," IEEE sensors journal, vol. 13, no. 10, 2013.
- [160] W. T. Navaraj, S. Gupta, L. Lorenzelli, and R. Dahiya, "Wafer Scale Transfer of Ultrathin Silicon Chips on Flexible Substrates for High Performance Bendable Systems," vol. 1700277, pp. 1–14, 2018.
- [161] Luisa Petti, Niko Munzenrieder, Christian Vogt, Hendrik Faber, Lars Buthe, Giuseppe Cantarella, Francesca Bottacchi, Thomas D. Anthopoulos, and Gerhard Troster, "Metal oxide semiconductor thin - film transistors for flexible electronics," 2016.
- [162] J. Abu-khalaf, R. Saraireh, S. Eisa, and A. Al-halhouli, "Experimental Characterization of Inkjet-Printed Stretchable Circuits for Wearable Sensor Applications," Sensors (Basel),18, 10pp. 1–23, 2018.
- [163] T. Kamiya and H. Hosono, "Material characteristics and applications of transparent amorphous oxide semiconductors," *NPG Asia Mater.*, vol. 2, no. 1, pp. 15–22, 2010.
- [164] K. Nomura, H. Ohta, A. Takagi, T. Kamiya, M. Hirano, and H. Hosono, "Roomtemperature fabrication of transparent flexible thin-film transistors using amorphous oxide semiconductors," *Nature*, vol. 432, no. 7016, pp. 488–492, 2004.
- [165] A. Sharma, "Performance Evaluation of Thin Film Transistors : History , Technology Development and Comparison : A Review," vol. 89, no. 15, pp. 36–40, 2014.
- [166] Paul Heremans, Jan Genoe, Soeren Steudel, Kris Myny, Steve Smout, Peter Vicca, Christiane, Grillberger, Olaf R. Hild, Francois Furthner, Bas van der Putten, shutosh Tripathi, Gerwin H. Gelinck, "Thin-film transistors and circuits on plastic foil Thinfilm transistors and circuits on plastic foil," 97-4244-5640, 2010.
- [167] T. T. Applications, "ChemInform Abstract: Solution-Processable Metal Oxide Semiconductors for," no. June, 2013.
- [168] N. Wijeyasinghe and T. D. Anthopoulos, "Copper(I) thiocyanate (CuSCN) as a holetransport material for large-area opto/electronics," *Semicond. Sci. Technol.*, vol. 30, no. 10, p. 104002, 2015.
- [169] S. M. Sze, Physics of semiconductor devices, 2nd ed. New York: John Wiley & Sons,

1981.

- [170] A. R. Brown, C. P. Jarrett, D. M. De Leeuw, and M. Matters, "Field-effect transistors made from solution-processed organic semiconductors," *Synth. Met.*, vol. 88, no. 1, pp. 37–55, 1997.
- [171] S. M. Sze, "Semiconductor Devices: Physics and Technology," *Technology*, p. 568, 2006.
- [172] S. D. Brotherton, Introduction to Thin Film Transistors: Physics and Technology of TFTs. 2013.
- [173] S.-M. Kang and Y. Leblebici, *CMOS Digital Integrated Circuits Analysis & Design, 2nd* ed.: McGraw-Hill. 1999.
- [174] G. Adamopoulos, S. Thomas, P. H. Wöbkenberg, D. D. C. Bradley, M. A. McLachlan, and T. D. Anthopoulos, "High-mobility low-voltage ZnO and Li-doped ZnO transistors based on ZrO2 high-k dielectric grown by spray pyrolysis in ambient air," *Adv. Mater.*, vol. 23, no. 16, pp. 1894–1898, 2011.
- [175] G. Adamopoulos *et al.*, "Structural and electrical characterization of ZnO films grown by spray pyrolysis and their application in thin-film transistors," *Adv. Funct. Mater.*, vol. 21, no. 3, pp. 525–531, 2011.
- [176] T. W. Seo, H.-S. Kim, K.-H. Lee, K.-B. Chung, and J.-S. Park, "High Mobility and Stability of Thin-Film Transistors Using Silicon-Doped Amorphous Indium Tin Oxide Semiconductors," J. Electron. Mater., vol. 43, no. 9, pp. 3177–3183, 2014.
- [177] H. Kim and J. Park, "High-Performance Zinc Tin Oxide Semiconductor Grown by Atmospheric-Pressure Mist-CVD and the Associated Thin-Film Transistor Properties," 2017.
- [178] Y. Nam, H.-O. Kim, S. H. Cho, and S.-H. Ko Park, "Effect of hydrogen diffusion in an In-Ga-Zn-O thin film transistor with an aluminum oxide gate insulator on its electrical properties," *RSC Adv.*, vol. 8, no. 10, pp. 5622–5628, 2018.
- [179] C. Liu, Y. Sun, H. Qin, Y. Liu, S. Wei, and Y. Zhao, "Low-Temperature, High-Performance InGaZnO Thin-Film Transistors Fabricated by Capacitive Coupled Plasma-Assistant Magnetron Sputtering," *IEEE Electron Device Lett.*, vol. 40, no. 3, pp. 415–418, 2019.
- [180] I. B. Society, "Sir Ronald Fisher and the Design of Experiments Author (s): F. Yates Reviewed work (s): Source: Biometrics, Vol. 20, No. 2, In Memoriam: Ronald Aylmer Fisher, 1890-1962 (Jun., Published by: International Biometric Society Stable URL: http," Society, vol. 20, no. 2, pp. 307–321, 2011.
- [181] H. Atil, "A Different Approach of Experimental Design: Taguchi Method," Experimental Design pp 449-480.
- [182] Paul D. BergerRobert E. MaurerGiovana B. Celli "Introduction to Taguchi method," .
- [183] X. Bie, J. Lu, Y. Wang, L. Gong, Q. Ma, and Z. Ye, "Applied Surface Science Optimization of parameters for deposition of Ga-doped ZnO films by DC reactive magnetron sputtering using Taguchi method," *Appl. Surf. Sci.*, vol. 257, no. 14, pp. 6125–6128, 2011.
- [184] W. Yang, C. Tsao, C. Hsu, H. Chang, C. Chou, and J. Kao, "Fabrication and Characterization of Transparent Conductive ZnO: Al Thin," vol. 2147, no. 30321, pp. 2140–2147, 2012.

- [185] B. Trust, "Biometrika Trust The Design of Optimum Multifactorial Experiments Author (s): R. L. Plackett and J. P. Burman Published by : Oxford University Press on behalf of Biometrika Trust Stable URL : http://www.jstor.org/stable/2332195 Accessed : 08-07-2016," vol. 33, no. 4, pp. 305–325, 2016.
- [186] T. Song, Y. Chen, C. Chung, Y. M. Yang, and B. Bob, "Nanoscale Joule Heating and Electromigration Enhanced Ripening of Silver Nanowire Contacts," no. 3, pp. 2804– 2811, 2014.
- [187] J. Lee, S. T. Connor, Y. Cui, P. Peumans, E. Engineering, and S. U. V, "Solution-Processed Metal Nanowire Mesh Transparent Electrodes," Nano Lett., 8, 2, pp. 7– 10, 2008.
- [188] D. M. Tanenbaum *et al.*, "The ISOS-3 inter-laboratory collaboration focused on the stability of a variety of organic photovoltaic devices," *RSC Adv.*, vol. 2, no. 3, pp. 882– 893, 2012.
- [189] J. Kettle *et al.*, "Printable luminescent down shifter for enhancing efficiency and stability of organic photovoltaics," *Sol. Energy Mater. Sol. Cells*, vol. 144, pp. 481– 487, 2016.
- [190] H. H. Khaligh and I. A. Goldthorpe, "Failure of silver nanowire transparent electrodes under current flow," *Nanoscale Res. Lett.*, vol. 8, no. 1, p. 1, 2013.
- [191] D. Chen, F. Zhao, K. Tong, G. Saldanha, C. Liu, and Q. Pei, "Mitigation of Electrical Failure of Silver Nanowires under Current Flow and the Application for Long Lifetime Organic Light-Emitting Diodes," *Adv. Electron. Mater.*, vol. 2, no. 8, pp. 1– 10, 2016.
- [192] D. Fantanas, A. Brunton, S. J. Henley, and R. A. Dorey, "Investigation of the mechanism for current induced network failure for spray deposited silver nanowires," *Nanotechnology*, vol. 29, no. 46, 2018.
- [193] J. Jiu *et al.*, "Reliability of silver nanowire transparent electrode under atmospheric environment," *IEEE-NANO 2015 15th Int. Conf. Nanotechnol.*, pp. 1390–1393, 2015.
- [194] F. Galeotti, W. Mróz, M. Catellani, B. Kutrzeba-Kotowska, and E. Kozma, "Tailorable perylene-loaded fluorescent nanostructures: A multifaceted approach enabling their application in white hybrid LEDs," J. Mater. Chem. C, vol. 4, no. 23, pp. 5407– 5415, 2016.
- [195] G. Horowitz, R. Hajlaoui, and P. Delannoy, "Temperature Dependence of the Field-Effect Mobility of Sexithiophene. Determination of the Density of Traps," J. Phys. III, vol. 5, no. 4, pp. 355–371, 1995.
- [196] X. Meng *et al.*, "Roll-To-Roll Printing of Meter-Scale Composite Transparent Electrodes with Optimized Mechanical and Optical Properties for Photoelectronics," ACS Appl. Mater. Interfaces, vol. 10, no. 10, pp. 8917–8925, 2018.
- [197] H. H. Khaligh and I. A. Goldthorpe, "Failure of silver nanowire transparent electrodes under current flow," *Nanoscale Res. Lett.*, vol. 8, pp. 2–7, 2013.
- [198] Y. Ahn, Y. Jeong, and Y. Lee, "Improved Thermal Oxidation Stability of Solution-Processable Silver Nanowire Transparent Electrode by Reduced Graphene Oxide," *ACS Appl. Mater. Interfaces*, vol. 4, p. 6410–6414, 2012.
- [199] B. Kuys, "AN IMPROVED TEST METHOD FOR ASSESSING THE RESISTANCE OF PAINT COATINGS TO SEPARATION FROM SURFACE MODIFIED TIMBER AN IMPROVED TEST METHOD FOR ASSESSING THE RESISTANCE OF PAINT COATINGS

TO SEPARATION FROM SURFACE MODIFIED," no. February 2015, 2013.

- [200] M. P. Seah, I. S. Gilmore, and G. Beamson, "XPS: Binding energy calibration of electron spectrometers 5 - Re-evaluation of the reference energies," *Surf. Interface Anal.*, vol. 26, no. 9, pp. 642–649, 1998.
- [201] X. Lin and S. Wang, "Electrochemical generation of silver nanowires," *Indian J. Chem.*, vol. 44A, no. May, pp. 1016–1018, 2005.
- [202] T. C. Gomes, R. F. De Oliveira, C. J. L. Constantino, M. M. Da Silva, É. M. Lopes, and N. Alves, "Thermal inkjet printing of copper tetrasulfonated phthalocyanine (CuTsPc) as a semiconducting layer on flexible MIS capacitors," *Mater. Res.*, vol. 17, no. 6, pp. 1466–1473, 2014.
- [203] M. Fakhri, H. Johann, P. Görrn, and T. Riedl, "Water as Origin of Hysteresis in Zinc Tin Oxide Thin-Film Transistors," ACS Appl. Mater. Interfaces, vol. 4, no. 9, pp. 4453– 4456, Sep. 2012.
- [204] T. W. Hickmott, "Interface States at the Anodized Al2O3-Metal 735 Interface," *J. Appl. Phys.*, vol. 10, no. 89, p. 5502–5508, 2001.
- [205] S. Sanjeev and D. Kekuda, "Effect of annealing temperature on the structural and optical properties of zinc oxide (ZnO) thin films prepared by spin coating process," *IOP Conf. Ser. Mater. Sci. Eng.*, vol. 73, no. 1, 2015.
- [206] T. Ohshima, T. Ikegami, K. Ebihara, and R. K. Thareja, "Photo-excited photonic characteristics of ZnO thin films deposited by laser ablation method," *Electr. Eng. Japan (English Transl. Denki Gakkai Ronbunshi)*, vol. 144, no. 3, pp. 1–7, 2003.
- [207] S. Marouf, A. Beniaiche, H. Guessas, and A. Azizi, "Morphological, Structural and Optical Properties of ZnO Thin Films Deposited by Dip Coating Method," *Mater. Res.*, vol. 20, no. 1, pp. 88–95, 2016.
- [208] D.-W. Jeon *et al.*, "Effect of annealing temperature on optical band-gap of amorphous indium zinc oxide film," *J. Alloys Compd.*, vol. 509, no. 41, pp. 10062– 10065, 2011.
- [209] D. Tahir and K. H. Jae, "Effect of growth temperature on structural and electronic properties of ZnO thin films," *AIP Conf. Proc.*, vol. 1801, no. January, 2017.
- [210] Y. Tu *et al.*, "Control of oxygen vacancies in ZnO nanorods by annealing and their influence on ZnO/PEDOT:PSS diode behaviour," *J. Mater. Chem. C*, vol. 6, no. 7, pp. 1815–1821, 2018.
- [211] L. Liu *et al.*, "Oxygen vacancies : The origin of n -type conductivity in ZnO," vol. 235305, pp. 1–6, 2016.
- [212] J. Li, C. X. Huang, J. H. Zhang, W. Q. Zhu, X. Y. Jiang, and Z. L. Zhang, "Temperature stress on a thin film transistor with a novel BaZnSnO semiconductor using a solution process," *RSC Adv.*, vol. 5, no. 13, pp. 9621–9626, 2015.
- [213] B. J. Park, K. Kim, Y. Park, Y. Mo, H. D. Kim, and J. K. Jeong, "Novel ZrInZnO Thin-film Transistor with Excellent Stability," vol. 577, pp. 329–333, 2009.
- [214] Y. Soo, S. Chang, and S. Yeol, "Microelectronic Engineering Effects of gate insulators on the performance of a-IGZO TFT fabricated at room-temperature," *Microelectron. Eng.*, vol. 88, no. 7, pp. 1590–1593, 2011.
- [215] J. Jeong *et al.*, "Effective mobility enhancement of amorphous In-Ga-Zn-O thin-film transistors by holographically generated periodic conductor Effective mobility

enhancement of amorphous In-Ga-Zn-O thin-film transistors by holographically generated periodic conductor," AIP Advances vol. 6, no. 8, 2018.

- [216] B. R. A. Street, "Thin-Film Transistors," pp. 2007–2022, 2009.
- [217] A. You, M. A. Y. Be, and I. In, "Role of silicon in silicon-indium-zinc-oxide thin-film transistor," Appl. Phys. Lett. vol. 97, pp. 95–98, 2010.
- [218] A. You, M. A. Y. Be, and I. In, "Effect of channel thickness on density of states in amorphous InGaZnO thin film transistor," vol. 122105, no. February, 2011.
- [219] I. L. S. Lee and J. G. K. Kim, "Comparative Study of Device Performance and Reliability in Amorphous InGaZnO Thin-Film Transistors with Various High-k Gate Dielectrics," Jpn. J. Appl. Phys. vol. 52, 2013.
- [220] D. Son, D. Kim, S. Sung, E. Jung, and J. Kang, "High performance and the low voltage operating InGaZnO thin fi lm transistor," *Curr. Appl. Phys.*, vol. 10, no. 4, pp. e157– e160, 2010.
- [221] L. F. Tang *et al.*, "Influence of white light illumination on the performance of a-IGZO thin film transistor under positive gate-bias stress," *Chinese Phys. B*, vol. 24, no. 8, 2015.
- [222] W. Cai *et al.*, "One-Volt IGZO Thin-Film Transistors with Ultra-Thin, Solution-Processed AlxOy Gate Dielectric," *IEEE Electron Device Lett.*, vol. 39, no. 3, pp. 1–1, 2018.
- [223] J. B. Kim, C. Fuentes-Hernandez, W. J. Potscavage, X. H. Zhang, and B. Kippelen, "Low-voltage InGaZnO thin-film transistors with Al2 O3 gate insulator grown by atomic layer deposition," *Appl. Phys. Lett.*, vol. 94, no. 14, pp. 10–13, 2009.
- [224] K. S. Shamala, L. C. S. Murthy, and K. N. Rao, "Studies on optical and dielectric properties of Al 2 O 3 thin films prepared by electron beam evaporation and spray pyrolysis method," Materials Science and Engineering B, vol. 106, pp. 269–274, 2004.
- [225] K. S. Kim *et al.*, "An all oxide-based imperceptible thin-film transistor with humidity sensing properties," *Materials (Basel).*, vol. 10, no. 5, 2017.
- [226] R. R. L. De Oliveira, D. A. C. Albuquerque, and T. G. S. Cruz, "Measurement of the Nanoscale Roughness by Atomic Force Microscopy: Basic Principles and Applications."

Appendix A Surface Roughness Parameters

This appendix (A) details the theory and mathematical equations of surface roughness parameters used to define physical properties of AgNW TCEs, gate dielectric and active layer of TFTs during this PhD research work.

Roughness

Among height parameters, the roughness average (R_a) is the most widely used because it is a simple parameter to obtain when compared to other surface roughness parameters. The roughness average is defined by equation (A.1) given below [226]:

Where Z(x) is the function that defines the surface profile analysed in terms of height (Z) and position (x) of the sample over the evaluation length "L" Thus, the Ra is the arithmetic mean of the absolute values of the height of the surface profile Z(x). Roughness average is also called as the Arithmetic Average (AA), Centre Line Average (CLA) or Arithmetical Mean Deviation of the Profile. The average roughness is just the mean absolute profile and its disadvantage is that it does not make any distinction between peaks and valleys on surface of sample. The average roughness has its own advantages and disadvantages. Advantage is that less sophisticated instrument, for example, a profilometer can provide (R_a).

$$R_a = \int_0^L \frac{Z(x)}{L} dx \tag{A.1}$$



Figure A.1: Profile of a surface (Z). It represents the average roughness Ra and Rq is the RMS roughness based on the mean line (B.C. MacDonald & Co, 2011) [50].

Another surface roughness parameter is root mean square of roughness (R_q). The root mean square (RMS) is a statistical measure and is widely used in different fields of science and technology. The root mean square of roughness (R_q) is a function that takes the square of the measured values. The RMS roughness of a surface is similar to the roughness average, however the only difference being the mean squared absolute values of surface roughness profile. The advantage of R_q is that it is more sensitive to peaks and valleys than the average roughness R_a due to the squaring of the amplitude in its calculation. The software does not need to be sophisticated in order to measure R_q and for this reason most of the surface analysis equipment provides R_q . The function R_q is defined (Gadelmawla et al., 2002) by equation (A.2) as follows:

$$R_q = \left[\int_0^L Z^2(x)/L \, dx\right]^{1/2} \tag{A.2}$$

Ten-point average roughness (R_z) is the arithmetic mean of the five highest peaks added to the five deepest valleys over the evaluation length, L measured shown in Figure A.2.



Figure A.2: a) Profile of a surface (Z). It represents the average roughness Ra and Rq is the RMS roughness based on the mean line (B.C. MacDonald & Co, 2011).

Appendix B Fabrication recipe of AgNWs TCEs

This appendix (B) explains the recipe and details of developed spray technique for fabrication of transparent conducting electrodes (TCEs).

Three types of silver nanowires were used for fabrication of AgNWs TCEs namely *thin-short* nanowire; AgNWs-60S ($d_{mean} = 60 \text{ nm}$, $l_{mean} = 20-30 \mu\text{m}$), *thin-long* nanowire; AgNWs-30L ($d_{mean} = 30 \text{ nm}$, $l_{mean} = 100-200 \mu\text{m}$) and *thick-long* nanowires; AgNWs-100L ($d_{mean} = 100 \text{ nm}$, $l_{mean} = 100-200 \mu\text{m}$). The shorter nanowires are advantageous for achieving high conducting networks at low concentration and the longer nanowires provide higher conductivity performance. The combined advantages of short and long nanowire were exploited by coating them together. Two mixed silver nanowires composition were prepared, one consisted of the type: *thin-short* nanowires (AgNws-60S) with *thin-long* nanowires (AgNWs-30L), denoted as 'AgNWs-M1', and another one consist of the type: *thin-short* nanowires (AgNWs-60S) with *thick-long* nanowires (AgNWs-100L), denoted as AgNWs-M2. The results show a moderate improvement in the electrical/optical performance and in the surface roughness (Shown in chapter 4).

Silver nanowires of the type AgNWs-60S, AgNWs-30L and AgNWs-100L (silver purity 99.5%, concentration 20 mg/mL in ethanol) were purchased from ACS materials. All materials were diluted with ethanol to the concentration of 0.5 mg/mL prior to spray coating. Flexible transparent substrates of planarised polyethylene-napthalate (PEN) with thickness 125 µm were used for this work. PEN was spin coated with poly methyl methacrylate (PMMA) (molecular weight 495K, 8% solution in anisole) purchased from Microchem. For this work, a PMMA buffer layer was employed to reduce the surface roughness of the substrates. HPLC grade isopropyl alcohol (IPA), ethanol (EtOH) and acetone were purchased form Sigma-Aldrich and used as such.

Mixed composite silver nanowires AgNWs-M1 dispersion was prepared by mixing AgNW-60S and AgNWs-30L as supplied in 1:1 volume by volume (V/V) ratios and then diluted with ethanol to 0.5 mg/mL final concentration. "After some initial tests, the ratio of 1:1 was found to be the optimum, so all benchmarking of performances was done at this ratio." Similarly, AgNWs-M2 was prepared with AgNWs-60S and AgNWs-100L. Silver nanowires dispersions in ethanol of each type (0.5 mg/mL) were sonicated for 5 minutes to minimized aggregates before air spray coating. "Ultrasonication and was used prior to spray coating to ensure that the AgNWs were dispersed in the suspension prior

to spray coating. A short duration was used (<5 minutes) at low power in order to ensure no breakage of AgNWs " For air spray coating, a back pressure of air kept at 1 bar using a piston type oil-less airbrush compressor was used for uniform gas/liquid flow rate (0.2 ml/min) at the nozzle. The spray nozzle diameter was 0.5 mm and the fluid cup capacity was 7 cm⁻³ with the air compressor. Substrates PEN of area 15 mm × 18 mm were thoroughly cleaned with DI water, acetone and IPA followed by drying with a jet flow of nitrogen. PMMA (average molecular weight~15,000) was spin coated on cleaned PEN substrates at 2500 rpm for 20 seconds followed by annealed at 100 °C for one minute. The substrates were placed on the substrate heater at 60 °C and the silver nanowires dispersion was air sprayed onto the heated substrates to evaporate the EtOH. The deposition time for preparing electrodes with R_{Sh} \approx 25 Ω sq⁻¹ was found to be approximately 5 mins ±30 seconds. However, using mixed geometry electrodes, the deposition time was approximately 50%. ZnO nanoparticles (diameter < 100) were purchased from sigma Aldrich. Composite ZnO/ AgNWs transparent electrodes were manufactured by spray coating nanoparticles onto post-processed AgNWs.

Three sequential post processing steps namely thermal embossing, N₂ plasma treatment and photonic sintering were applied to fresh sprayed TCEs. The AgNW-coated PEN substrates were inserted into a thermal nanoimprint lithography (NIL) system to compress the AgNW on the surface of PEN at different temperatures, times and pressure. The thermal-embossing process was conducted using a 2" nickel shim, formed from a polished silicon wafer to ensure flatness. The anti-sticking layer BGL-GZ-83 was spin coated onto the nickel shims prior to use in the thermal embossing to avoid sticking of the AgNWs and thermoplastic to the nickel shim during the embossing procedure. Pressure, time and temperature of the embossing step were all varied and the process was optimised using the Taguchi method. The optimal conditions found in this way were 5 Bar, 1 minute and 120°C, respectively. All embossing operations were conducted under ambient pressure rather than under vacuum.

The photonic curing was performed using a Novacentrix system to rapidly anneal the surface for enhanced electrical conductivity. Samples were mounted to the system and exposure varied from 450V, 400µs to 750µs with single energy pulse 1488mJ/cm² to 2712mJ/cm². The process ensured sintering occurred within the top 1µm of the film surface, rather than the entire thickness of substrate and therefore impacted mostly upon the AgNW film. By concentrating the sintering so close to the surface, flexible and polymeric substrates can be used and surface damage can be minimised. The exposure
time or energy pulse duration was varied to optimise the lowest sheet resistance. The optimal energy was set at 700μ s with an energy of 2549mJ/cm².

A plasma treatment step was also applied to AgNW-coated PEN substrates. The AgNW electrodes were Nitrogen (N₂) plasma treated for 2 minutes, which result in reduction of sheet resistance, although no change in transmittance was noted. The treatment process was conducted using a Diener second 'Nano' plasma treatment system at ambient pressure after a 30 seconds nitrogen purge. In this system, a 13.56 MHz RF Generator was used at 100 watts power to generate N₂ plasma for 120 seconds.

Appendix C Aluminum Anodization Process and Transfer Curves

This appendix (C) details the process of the aluminium anodization and transfer curves of ZnO TFTs manufactured using Plackett–Burman designs.

The aluminium anodisation process was carried out by using the Al-coated glass substrate as anode and a stainless-steel sheet (40 mm x 40 mm in area, 1.0 mm thick) as the cathode, both submerged into an electrolytic solution comprising ethylene glycol, deionized water and tartaric acid (0.1 M). The electrodes were kept at a fix distance of 50 mm during the anodisation process, which was carried out in two steps: i) a constant current step and ii) a constant voltage step (using a Keithley SMU model 237). The initial constant current step promotes the growth of an Al₂O₃ layer onto metallic Al film. At constant current, the film resistance increases linearly with the thickness, resulting in a linear increase of the applied voltage. The determination of the oxide thickness (t), therefore, depends only on the final voltage (VF) of the constant current step. The oxide thickness follows the relationship t = cVF, where c was empirically determined to be ~ 1.2 nm/V. After the desired final voltage is achieved, the source unit is switched from constant current to constant voltage and an exponential decay of the current is observed. The constant voltage step does not increase the thickness of the Al₂O₃ layer but is responsible for eliminating pinholes in the oxide film.

Shown in Figure C.1 and C.2 (please see next page) the transfer curves for ZnO TFTs manufactured using process conditions according to runs # 1 to # 12 used in Plackett–Burman Design (Chapter 5).



Figure C.1: Transfer curves for transistors built using conditions according to runs # 1 to # 6.



Figure C.2: Transfer curves for transistors built using conditions according to runs # 7 to # 12.